

U.S. - JAPAN COORDINATED PROGRAM

FOR

MASONRY BUILDING RESEARCH

PB93-214641

REPORT NO. 5.2-1

ASSEMBLY OF EXISTING DIAPHRAGM DATA

by

MAX L. PORTER FRANCISCO S. YEOMANS ALBIN W. JOHNSON

JULY 1990

supported by:

NATIONAL SCIENCE FOUNDATION GRANT NO. CES-8517028 FORMERLY GRANT NO. ECE-8517028



Disclaimer

This report presents the results of a research project which was part of the U.S. Coordinated Program for Masonry Building Research. The program constitutes the United States part of the United States - Japan Coordinated Masonry Research Program conducted under the auspices of the Panel on Wind and Seismic Effects of the U.S.-Japan Natural Resources Development Program (UJNR).

This material is based on work supported by the National Science Foundation under the direction of Program Director, Dr. S.C. Liu.

Engineering data, designs, details, and suggested specifications presented in this publication have been delineated in accordance with recognized professional principles and practices and are for general information only. The date, designs, details, and suggested conclusions should not, therefore, be used without first securing competent advice with respect to their suitability for any given application. The responsibility for the use of the information in this report remains with the user. Since additional analysis and subsequent journal publications are ongoing, this report is for information purposes and is made available with the understanding that it will not be cited or reproduced without the permission of the authors.

The contents of this report do not represent a warranty of the products used on behalf of the State of Iowa, Iowa State University, or the authors. Any opinions, findings, and conclusions or recommendations expressed in this publication are those of the authors and do not necessarily reflect the views of the National Science Foundation and/or the United States Government. The authors do not guarantee the accuracy or completeness of any of the information published herein and will not be responsible for any errors, omissions, or damages arising out of the use of this information.

1a

PB93-214641

ASSEMBLY OF EXISTING DIAPHRAGM DATA

A report submitted to the National Science Foundation Research Contract No. ECE - 8517028

by

Max L. Porter, P.E., Ph.D., Professor; Department of Civil and Construction Engineering Town Engineering Building Iowa State University Ames, IA 50011

Francisco S. Yeomans, Research Assistant; Department of Civil and Construction Engineering Town Engineering Building Iowa State University Ames, IA 50011

and

Albin W. Johnson S. B. Barnes and Associates 2236 Beverly Blvd. Los Angeles, CA 90057

JULY 1990

ABSTRACT

Masonry buildings are subjected to earthquake and other lateral forces. These loads must be distributed through the different structural components. In masonry structures the floor diaphragm transmits these loads from one wall to another. Adequate diaphragm action is necessary for the floor system to absorb or transmit this dissipated energy. Proper connections between the diaphragm and the masonry walls, as well as between individual diaphragm elements, are essential to the development of this diaphragm action. Other parameters affecting the diaphragm behavior are its thickness, orientation, and (in the case of precast concrete elements) the addition of topping, and the number of boundary sides connected. In short, the structural integrity of the masonry buildings is linked directly to the resistivity of the diaphragm to lateral in-plane loads.

Category 5.0 of the U.S. - Japan coordinated program for masonry research is a detailed investigation of horizontal diaphragms. The objective of this research task (5.2) is to assemble extensive experimental data on various types of floor diaphragms and to reduce the data to a form required for static and dynamic analysis models. An extensive bibliographic search was done on cold-formed steel decks, composite steel decks, timber diaphragms, reinforced concrete diaphragms, and precast prestressed concrete diaphragms. Equations required to predict the initial stiffness and peak strength (parameters related to the EKEH model) for steel deck reinforced concrete, cold-formed steel, precast prestressed, and plywood diaphragms are included. Experimental data related to the Lumped Parameter Model (EKEH Model) is also presented.

TABLE OF CONTENTS

P	age
TITLE PAGE	i
ABSTRACT	ii
LIST OF FIGURES	vi
LIST OF TABLES V	iii
1. INTRODUCTION	1
1.1 General Remarks	1
1.2 Objective of the Overall Research Program	1
1.3 Scope of the Project	3
2. REVIEW OF PREVIOUS WORK	5
2.1 General	5
2.2 Cold-Formed Steel Diaphragms	5
2.3 Composite Steel Deck	7
2.4 Wood Diaphragms	10
2.5 Precast Concrete Diaphragms	11
2.6 Reinforced Concrete Diaphragms	13
2.7 Hysteretic Models	14
3. DATA FORMULATION OF PREVIOUS WORK	23
3.1 General	23
3.2 Research perfomed by Luttrell, et. al.	23
3.3 Research performed by Lukens	25
3.4 Research performed by Young and Medearis	32
3.5 Research performed by Porter and Easterling	36
3.6 Research performed by ABK, a Joint Venture	40
3.7 Research performed by Porter, et. al.	48
4. INITIAL STIFFNESS AND MAXIMUM LOAD PREDICTION	55
4.1 General	55
4.2 Cold-Formed Steel Diaphragms	55
4.2.1 Diaphragm Strength	55
4.2.1.1 Strength of the Connections	55
4.2.1.2 Strength Controlled by Panel Buckling	60
4.2.2 Diaphragm Stiffness	60
4.3 Steel Deck Reinforced Concrete Diaphragms	60

					Page
		4.3.1	Diaphrag	m Strength	61
			4.3.1.1	Diagonal Tension	61
			4.3.1.2	Edge Fasteners	61
			4.3.1.3	Shear Transfer Mechanism	66
		4.3.2	Initial	Stiffness	66
			4.3.2.1	Bending Stiffness	67
			4.3.2.2	Shear Stiffness	67
			4.3.2.3	Edge Zone Stiffness	67
			4.3.2.4	Frame Connection Stiffness	68
	4.4	Precas	t Prestre	ssed Diaphragms	68
		4.4.1	Initial	Stiffness	. 68
			4.4.1.1	Bending Stiffness	68
			4.4.1.2	Shear Stiffness	69
			4.4.1.3	Edge Zone Stiffness	69
			4.4.1.4	Frame Connection Stiffness	70
		4.4.2	Peak Str	rength	70
			4.4.2.1	FME and Limit State Prediction :	for
				Shear-Bond Failure	71
			4.4.2.2	FME and Limit State Prediction	for
				Tensile-Bond Failure	72
			4.4.2.3	FME and Limit State Prediction	for
				Diagonal Tension Failure	73
	4.5	Plywoo	d Diaphra	gms	73
		4.5.1	Diaphrag	m Strength	73
		4.5.2	Diaphrag	m Stiffness	74
5.	SUMM	ARY			76
6.	ACKN	OWLEDGE	MENTS		77
7.	REFE	RENCES			78
8.	APPE	NDIX A:	DIAPHRAG	M DATA	124
	8.1	Cold-F	ormed Ste	el Deck	124
		8.1.1	Data obt	ained from Reference 25, 305,	,
			and 306		124
		8.1.2	Data obt	ained from Reference 299	128
		8.1.3	Data obt	ained from Reference 3	129

								Page
		8.1.4	Data	obtained	from	Reference	320	130
	8.2	Wood D	iaphra	agms				131
		8.2.1	Data	obtained	from	Reference	518	131
	_	8.2.2	Data	obtained	from	Reference	3	132
	8.3	Steel-	Deck-H	Reinforced	1 Cond	crete Diaph	iragms	133
		8.3.1	Data	obtained	from	Reference	378	133
	8.4	Plank 1	Diaphı	ragms				135
		8.4.1	Data	obtained	from	Reference	376	135
9.	APPE	NDIX B:	DIAPH	HRAGM SYST	rems			136

v

•

LIST OF FIGURES

		r -	aye
Fig.	1	Distribution of lateral forces on a structure	2
Fig.	2	Example of several cycles of hysteretic curve for ISU	
		Test #5 employing modified Takeda model [376]	17
Fig.	3	Rules for modified Takeda model [431]	18
Fig.	4	Example of several cycles of hysteretic curve for ISU	
		Test #5 emplying EKEH/LPM model [376]	19
Fig.	5	EKEH model skeleton curve [376]	21
Fig.	6	Definition of shear stiffness G' [148]	24
Fig.	7	Schematic of diaphragm test showing location of	
		deflection gauges [148]	26
Fig.	8	Definition of shear stiffness C [299]	28
Fig.	9	Typical timber framing [299]	29
Fig.	10	Cantilever test apparatus [299]	31
Fig.	11	Typical plywood framing [518]	33
Fig.	12	Panel test showing location of deflection gauges [518]	34
Fig.	13	Test interpretation [518]	35
Fig.	14	Diaphragm test frame schematic [378]	38
Fig.	15	Schematic of servo-hydraulic testing system [378]	39
Fig.	16	Framing for wood sheathed diaphragms [2]	41
Fig.	17	Steel diaphragm testing [2]	44
Fig.	18	Schematic of the test setup for quasi-static testing	
	of	diaphragms [2]	46
Fig.	19	Diaphragm instrumentation [2]	47
Fig.	20	Typical dimensions for all diaphragms [481]	49
Fig.	21	Example of SPD load program [481]	50
Fig.	22	Example of hysteretic curves [481]	51
Fig.	23	Example of envelope curves [481]	52
Fig.	24	Schematic layout for diaphragm [307]	56
Fig.	25	Outer edge panel [307]	57
Fig.	26	Interior panel force distribution [307]	59
Fig.	27	Assumed forces acting on edges [378]	62
Fig.	28	Assumed forces acting on edge fasteners [378]	65

			Page
Fig.	B1	Typical steel-deck profiles [378]	137
Fig.	B2	Plywood and diagonally sheathed panels [22,149]	138
Fig.	B3	Typical span-deck hollow-core precast prestressed	
		cross-section with topping [481]	139
Fig.	B4	Typical reinforced concrete floor systems [344]	141
Fig.	B5	Typical cross-sections for Cold-Formed-Steel-	
		Decks [113,299]	142

LIST OF TABLES

			Page
TABLE	1	Summary of parameters for the diaphragm tests	53
TABLE	A1 _	Welded diaphragm test data [306]	124
TABLE	A2	Screwed diaphragm test data [306]	127
TABLE	A3	Steel-Deck on timber framing test data [299]	128
TABLE	A4	Steel-Deck test data [3]	129
TABLE	Α5	Steel-Deck on timber framing test data [320]	130
TABLE	A 6	Wood diaphragm test data [518]	131
TABLE	A7	Wood diaphragm test data [3]	132
TABLE	A8 -	SDRC test data [378]	133
TABLE	A9	Plank diaphragm test data [376]	135

1. INTRODUCTION

1.1 General Remarks

Buildings subjected to earthquake and wind forces must have floors and roofs capable of transferring in-plane shear forces from one wall to another. These floors are termed as diaphragms and are typically made of hollow core planks, reinforced concrete, composite steel deck reinforced slabs, or timber (See Appendix B a particular description on each type of diaphragm). for Diaphragms are relatively thin, usually rectangular, structural elements capable of resisting shear parallel to their edges. They must be used as walls in a vertical position (shear walls), or as roofs or floors in a horizontal position. Diaphragms may be considered to be a horizontal beam with the roof or the floor system acting as the web of the beam, the joists or beams function web stiffeners, and the peripheral beams or integral as reinforcement function as flanges.

The function of the diaphragm is to brace a structure against lateral forces, such as wind or earthquake loads, and to transmit these forces to the other resisting elements of the structure. Figure 1 illustrates the distribution of lateral forces on a typical structure. The distribution of lateral forces to the shear wall or space frame systems depends on the properties of the diaphragm and the resisting system.

In order to improve the performance of the floor and roof systems, it is desirable for the in-plane stiffness of the diaphragm to exceed that of their respective vertical subsystems [90]. Diaphragms of this type have been categorized as rigid. In this instance, the diaphragms act as a flat plate that transmits lateral loads to the bracing elements in proportion to their relative rigidities. Conversely, with flexible diaphragms, loads are distributed to vertical subsystems as a continuous beam using tributary areas. Regardless, both rigid and flexible systems should be able to retain a sufficient amount of in-plane stiffness or strength in order to prevent collapse, well beyond the elastic range [376].

1.2 Objective of the Overall Research Program

The research undertaken for this project is part of the U.S.-Japan Coordinated Program for Masonry Building Research. Each category of this program is conducted under the supervision of the Technical Coordinating Committee for Masonry Research (TCCMAR). The TCCMAR committee was organized to function under the auspices of the Panel of Wind and Seismic Effects of the U.S.-Japan Cooperative Program in Natural Resources (UJNR). Study of floor diaphragms, which is the objective of this project, is the fifth research task. Additional information of the organization of the Masonry Building Research Program is available in Reference [348].



Figure 1.

The study of diaphragm characteristics was divided into two separate tasks. Task 5.1 involved both experimental and analytical investigation of precast horizontal diaphragms subjected to inplane loading. Task 5.2 focused on the collection of existing literature and data generated from the discussion and testing of horizontal diaphragms. The subject matter of this report focuses on Task 5.2 only.

The objectives of the overall Task 5.0 research project were to determine the basic failure modes, ascertain behavioral characteristics, and investigate analytical properties for the analysis of full-scale precast, prestressed hollow-core plank diaphragms subjected to in-plane shear. Basic analytical features of interest include determination of the force-deflection relations, first major event load, limit state strength, ductility, stiffness, failure modes, and hysteretic behavior.

In order to fulfill these objectives, the effects of various parameters were investigated (Task 5.1). These parameters include:

- boundary condition (number of sides connected to the loading frame)
- orientation (placement of the planks with respect to the direction of the applied lateral load)
- slab thickness (plank depth of six, eight and twelve inches)
- aspect ratio (geometric configuration of the diaphragm)
- topping (addition of a two-inch cast-in-place concrete slab)
 seam connectors (variation in the number of seam connectors to verify the implications of attaining an alternate failure mode for the untopped tests)

The second phase of the project (Task 5.2) entailed gathering and reviewing existing literature and data on horizontal diaphragms. The information collected from both Task phases 5.1 and 5.2 of the project is to be combined and used in other task areas of the Masonry Building Research Program which culminates with the eventual construction and testing of a full-scale masonry building.

1.3 Scope of the Project

In order to assemble and reduce existing experimental data on various types of floor diaphragms to a form required for static and dynamic analysis models, the following diaphragm items were searched:

- a) Limit state strength
- b) Initial stiffness
- c) Failure mode type
- d) Test identification
- e) Type of testing

This document includes a summary of the basic data obtained from each of the items listed above. The purpose of this tabulation (Task 5.2) is to provide information to allow the formulation of a model for analysis of masonry buildings containing diaphragms. Section two contains review of previous work done on cold-formed steel, composite steel deck, wood, precast concrete, and reinforced concrete diaphragms, and also includes a review of hysteretic models. Section three provides a detailed description of the models used in the diaphragm research including the test rationale, brief description of the specimens, test setup, instrumentation, and description of the test data acquired which is reported in Appendix A. Section four contains the equations required to predict the initial stiffness and maximum load of composite steel deck, cold-formed steel, precast concrete, and plywood diaphragms. A summary of this project is contained in Section seven presents an exhaustive list of Section five. references.

2. REVIEW OF PREVIOUS WORK

2.1 General

A well-designed diaphragm is essential for the structural integrity of a building subjected to earthquake induced motions or wind forces. The distribution of in-plane forces from the diaphragm to the various elements of the vertical load resisting system depends on the properties of the diaphragm and the vertical load resisting system. Thus, knowledge of the behavior characteristics of a diaphragm is necessary to perform a lateral load (seismic) analysis of a multi-story building [404].

Diaphragms may be categorized according to their composition into the following common types: cold-formed steel, composite steel deck, timber, reinforced concrete, and precast concrete. The seismic performance of each of these systems is different and depends on the characteristics of the diaphragm and the event.

This chapter will review past research that relates to the behavioral and design of each one of the systems listed above. The following sections address studies of cold-formed steel, composite steel deck, timber, reinforced concrete, and precast concrete.

2.2 Cold-Formed Steel Diaphragms

The following review of steel-deck diaphragm research was partially taken from Section 2.3, pages 10-12, of Reference [378]. It was written by Max L. Porter, P.E. and W. Samuel Easterling.

Early testing of steel deck diaphragms established behavioral characteristics which are keys to understanding and attempting to predict strength and stiffness. Nilson [342] provided the first generally published article on steel deck diaphragm research which was performed at Cornell University. In this research Nilson established techniques for testing steel deck diaphragms which were published by the American Iron and Steel Institute (AISI) [14] and have since been accepted by the American Society for Testing and Materials (ASTM) [33]. He also developed and refined several welding techniques particularly suited to steel deck sections.

Diaphragm research at Cornell University and West Virginia University was done by Luttrell [301,302]. In addition to deriving a semi-empirical expression for prediction the shear stiffness of a steel-deck diaphragm, he concluded that the panel length influenced shear stiffness but not strength. Luttrell noted that warping of the profile occurs near the ends of the panel and extends for a finite distance towards the panel center. This warping can be decreased by adding additional fasteners or by stiffening the ends as is done with closure angles [303]. He also stated that the length of the warped region is independent of the panel length.

A series of about 160 full-scale shear diaphragm tests were done by Luttrell and Ellifritt at West Virginia University [308]. Numerous test parameters were varied. these parameters included welded fasteners, thickness of sheets, panel width, panel length, purlin spacing, and fastener spacings. They derived empirical equations to predict the stiffness and strength of the shear diaphragms.

Luttrell and Huang [309] worked on a summary of deck diaphragm studies done at Major Units Laboratory of West Virginia University (WVU) over the past fifteen years. The work done at WVU covered tests on a wide range of deck panel types, metal building wall systems, panels with insulating concrete fill, and plywood assemblies, all used as diaphragms, and under static, cyclic, and dynamic loading. Connection performance was evaluated through an extensive test program covering arc-spot welds, different types of screws, and power driven pins. This study was the support document for the 1981 and 1987 <u>Steel Deck Institute Diaphragm Design Manual</u> [306,307].

Numerical modeling of steel deck diaphragms using finite analysis was performed at Cornell University [343,38]. A linear elastic analysis was used by Nilson and Ammar [343] in which steel deck panels were modeled as one-dimensional linear beam elements and connections were modeled with two one-dimensional elastic link elements. The mechanical properties for the plane stress elements and the linkage elements were determined by either experimental or analytical means.

Analytical models and comparisons to past experimental work were made using tests by Nilson [342] and Luttrell [301] according to Atrek and Nilson [38]. Due to nonlinearities in connectors, which could not be adequately modeled with the software used by Ammar, the analytical results were compared with the experimental results up to 40 percent of the experimental ultimate load. According to Atrek and Nilson [38] stiffness comparisons from Ammar's work were satisfactory for cellular type diaphragms, but for noncellular diaphragms, the analysis showed a more flexible result than the experimental specimens. This result was attributed to poor evaluation of the effective shear modulus of the deck.

Atrek and Nilson [38] expanded the finite element analysis that was begun by Ammar to include the nonlinear behavior of the connections. In addition, they considered different approaches for determining equivalent orthotropic properties for the steel deck such that plane stress elements could be used in the analysis. These different approaches were developed by other researchers and are referenced extensively by Atrek and Nilson [38]. Work by Hussain and Libove [235] of Syracuse University was utilized by Atrek and Nilson for determining the shear modulus. In their work Atrek and Nilson reiterated the fact that the majority of the shear deformation occurs near the panel ends in steel-deck diaphragms. As part of their conclusions, they stated that increased accuracy can be obtained when comparing the analytical results to the experimental results by assigning the elements near the ends of the diaphragm a reduced effective shear modulus compared to the centrally located elements. However, they also conclude that the increase in accuracy does not, in most cases, overcome the added expense that comes with such mesh refinement.

A series of 14 full-scale diaphragm tests were performed and reported in 1981 by ABK, a joint venture [2,3]. Of the 14 diaphragms, two specimens were cold-formed steel deck diaphragms. The specimens were 20-gage steel decking, unfilled, the first one was unchorded with button-punched seams 6 in. o.c. Each specimen was attached to a 20-ft by 60-ft test frame. Loading consisted of a series of reversed cyclic, quasi-static and dynamic inputs with each series generally increasing in magnitude. The document is a data report on the experimental testing.

Twenty-two diaphragm tests were performed at Iowa State University by Lukens [299]. Each diaphragm consisted of 29-gage steel cladding on a timber frame. Numerous test parameters were These parameters included diaphragm length, purlin wood varied. type, nail pattern, and loading type. Of the 22 diaphragms, four specimens were subjected to reverse loading for a defined number of cycles and then were subjected to monotonic load to the ultimate As part of his conclusions Lukens stated the following: load. first, the panel strength had a tendency to be independent of length when failure was caused by the purlin to chord connection splitting the chord. Second, panels with Douglas fir-larch purlins were about 20% stronger at failure load than panels with sprucepine-fir purlins. Third, panels with nails through the ribs were about 40% less stiff than panels with screws in the valleys. Finally, reverse loading panels were about 12% weaker at failure load than nonreversed loading panels.

2.3 Composite Steel Deck

The following review of steel-deck-reinforced concrete diaphragms (SDRC) was partially taken from Section 2.2, pages 7-9, of Reference [378]. It was written by Max L. Porter, P.E. and W. Samuel Easterling.

Prior to the studies by Porter, et. al. [378,390] at Iowa State University (ISU), the quantity of general research done on SDRC diaphragm systems had been minimal. Most of the work that had been done was proprietary testing that was typically sponsored by various steel deck manufacturers. As a result of the proprietary nature of the work, information regarding the testing has not been

widely published, but has been incorporated into various design methods.

In reviewing the previous SDRC diaphragm research, both from the limited publishing and proprietary data reports, Porter and Easterling [378] concluded that a substantial amount of the SDRC diaphragm testing was performed by S. B. Barnes and Associates of Los Angeles, California [44-46]. Since the late 1940's numerous proprietary tests have been performed for several different steeldeck manufacturers. The results of some of these early tests were used to formulate the design approach in the <u>Seismic Design for</u> <u>Buildings</u> [127,128].

Discussions of the test results [44-46] indicate the predominant modes of failure were deck tearing around welds at the edge framing member, shearing of welds, shear failure of the concrete above the flutes of deck, localized cracking and concrete separation from the deck. In most citations of the separation of the deck and concrete, a simultaneous failure of the welds was noted. Diaphragms with both structural concrete fill and vermiculite fill were tested with a minimum of 1-1/2 inches of concrete above the top flange of the deck. Steel deck sections were typically welded to the framing members.

Based on some of the early tests by Barnes, a design method was developed using a guided cantilever concept and basic statics [29]. Resulting expressions were empirical and calibrated to fit the limited number of tests that were available at the time. The equations were typically adjusted for each different deck manufacturer's tests [127,128]. Methods for both strength and stiffness were formulated based on all available tests prior to the 1973 publishing of <u>Seismic Design for Buildings</u> [127,128].

A series of nine diaphragm tests with lightweight insulating fill were performed at WVU and reported by Luttrell [304]. Corrugated galvanized steel deck was used with nominal depths between 9/16 in. and 1-3/8 in. Welded connections were used with welding washers that had 3/8 in. diameter holes. Luttrell reported that the most noticeable effect of the fill was increasing the stiffness of the diaphragm system. The warping restraint provided to the deck by the insulating fill was adequate to force the failure to be at the welds as opposed to sheet instability, which had occurred in similar unfilled diaphragm tests. Expressions are presented that correlate diaphragm strength to the number of welds along the edges of the diaphragm.

Four composite diaphragm tests were performed at the University of Salford by Davies and Fisher [114]. Trapezoidal and re-entrant steel-deck profiles were used and attached to the perimeter framing members with self-drilling, self-tapping screws. Concrete cover was either approximately 2 in. or 2-3/4 in. In each case the controlling failure mode was reported as a fastener

failure, with one specimen failing by a combination of fastener failure and profile collapse. The fastener spacing ranged from approximately 12 in. to 28 in. Equations were presented to determine the strength of the diaphragm based only on a fastener failure. The expressions were developed based on assumptions regarding the force distribution on the fastener patterns.

A series of nine diaphragm tests (Phase I) were performed at ISU by Porter and Greimann [390]. Numerous SDRC diaphragm parameters were varied and tested. These parameters included steel deck type, fastener type and number, and concrete thickness. Equations were developed to predict stiffness and strength of SDRC diaphragms. These equations were based on an edge zone concept, which considered the force to be transferred from the load frame into the diaphragm within a relatively narrow region adjacent to the framing members. A key component of this edge zone was considered to be the interface between the steel deck and concrete. Edge force distributions that were used to derive predictive equations were based on a linear elastic finite element analysis. Additionally, force distribution at ultimate load levels were assumed.

A series of 14 diaphragm tests were performed and reported in 1981 by ABK, a joint venture [2,3]. The ABK group is made up of members from three firms, Agbabian Associates, S. B. Barnes and Associates and Kariotis and Associates, all in the Los Angeles area. Of the 14 diaphragms only one specimen was an SDRC diaphragm. The specimen was a 20-gage deck with 2-1/2 in. of concrete cover and attached to a 20-ft. by 60-ft. test frame with intermediate framing members. Loading consisted of a series of 11 reversed cyclic, quasi-static and dynamic inputs with each series generally increasing in magnitude; however, the specimen was not loaded to complete failure. The document [2] is a data report on the experimental testing.

An additional twenty-three diaphragm tests (Phase II) were performed at ISU by Porter and Easterling [378]. Numerous SDRC diaphragm parameters were varied. These parameters included steel deck type, fasteners type and number, concrete thickness, depth-tospan ratio, loading and framing member size. As a continuation of previous research at ISU by Porter and Greimann [390], this study modified predictive equations and the means for determining the components of the equations, and inelastic force distributions were determined. A major component of the analytical portion of the study was verifying the previously assumed force distributions at ultimate. The principal focus of the analyses was to verify and define the components of the predictive equations, such that the equations might be incorporated into a design methodology. Recommendations were made regarding design parameters.

2.4 Wood Diaphragms

The quantity of general research done on wood diaphragm systems has been extensive. However, most of the diaphragm test programs to date have used static monotonic loads. In 1980, a workshop on the design of horizontal wood diaphragms [22] sponsored by the National Science Foundation and conducted by Applied Technology Council recommended: First, perform additional dynamic diaphragm tests using either cyclic loads or input from realistic earthquake motions. Second, develop mathematical models and analysis methods to predict the inelastic response of diaphragms.

Early testings of diagonally sheathed diaphragms were performed by Doyle [133-135], Atherton and Johnson [36], Stillinger et. al. [462], Stillinger [459,460], Johnson [263,265], Currier [106] and Burrows and Johnson [73,268]. Several series of 20x60 foot diaphragm parameters were varied and tested. These parameters included combinations of lumber sheathing, framing, and fasteners. Testing performed by Stillinger [460] has shown that diagonally sheathed diaphragms can sustain ultimate loads up to about 1600 lb. per foot. An extensive review of wood diaphragm testing was done by Elliott [149].

Full-scale plywood-sheathed diaphragm tests were performed at Oregon Forest Products Laboratory by Stillinger and Countryman [461], and Johnson [264,266]. The same type of tests were performed by Countryman [101], Countryman and Colbenson [102], and Tissell [478] at American Plywood Association Laboratory. A review of these tests shows that the behavior of wood diaphragms is highly nonlinear. Most of these tests were, however, noncyclic.

An investigation of the structural damping characteristics of composite wood structures subjected to cyclic loading was conducted by Young and Medearis [518] at Stanford University. A series of eight 8x16 ft. plywood shear wall tests were performed with varied parameters. These parameters included plywood thickness, perimeter nailing, re-nailing, and pre-cycled diaphragms. As part of their conclusions, Young and Medearis stated the following: First, considerable frictional damping was generated at the common surface of the plywood sheets, as well as between the plywood and the framing members. Second, none of the panels experienced any sudden failures; and, although most of them were loaded until a number of different failure locations were observed, most of which were quite Third, the energy absorption properties of the local in nature. panels were very great, with about 60% of the input energy being absorbed during any given half-cycle. From their theoretical work using the test results, values of an equivalent viscous damping ratio were found to average 0.10 for shear walls with plywood on both sides.

Ewing et. al. [158], and Luttrell et. al. [186], have been working in development of mathematical models for static and dynamic analysis of wood diaphragms to predict their inelastic response when subjected to time variant forces. A lumped parameter analytical model was used by Ewing for analyzing wood diaphragms supported on masonry walls. The analysis provided quantitative estimates of the out-of-plane excitations induced in the masonry The analytical model included a nonlinear hysteretic walls. element that simulates the behavior observed in cyclic tests of The results of the nonlinear dynamic analysis wood diaphragms. were compared with corresponding results obtained with linear elastic and nonlinear elastic assumptions. The nonlinear elastic case showed a general reduction (relative to the linear elastic case) in peak responses, and the nonlinear hysteretic case showed further significant reductions in all response quantities. As part of their work they stated that elastic analyses overestimate the actual response and that consideration of the hysteretic characteristics of the wood diaphragms is necessary for more realistic response predictions.

Research made by Luttrell et. al. [186] has been toward the establishing of a method for predicting joint slip and its relationship to overall shear displacements. The aim of their work was toward developing a dynamic performance model and related exploratory physical testing. Full scale models were tested to determine stiffness and slip with large groups of nails, and associated small scale tests for nail slip tendencies; and preliminary stiffness models were developed. As part of their comments, they stated that for extension to the dynamic model, the key appears to be in the damping characteristics associated with joint slip. To that end, many separate tests have been done to quantify joint slip characteristics.

A series of 14 diaphragm tests were performed and reported by ABK, a joint venture [2,3]. Of the 14 diaphragms, eleven specimens were wood diaphragms. The wood diaphragms were 20x60 foot, three with straight sheathing, three with diagonal sheathing, and five with plywood, and included double board systems. As part of their results, they showed the values of the basic characteristics for a nonlinear hysteretic spring model, associated with wood diaphragms tested. (The document [2] is a data report on the experimental testing).

2.5 Precast Concrete Diaphragms

During previous seismic events, the performance of precast concrete units without topping has been poor, while the precast concrete units with topping have exhibited variable to good performance [150]. Martin and Korkosz stated that the absence of continuity and redundancy (between the precast slabs) has caused some designers to question the stability (of precast structures) under high lateral loads [314].

An experimental investigation of the shear diaphragm capacity was undertaken by Concrete Technology Corporation in February 1972 [93]. Three 15 ft. long, 4 ft. wide, 8 in. thick Spiroll Corefloor hollow-core slabs were assembled side by side. The two longitudinal joints were filled with nonshrink grout. At the ends of the plank system two end beams 4 in. x 8 in. x 12 ft. each were cast in place. Two #3 grade 60 deformed bars were placed in each beam. The objectives of this test were to measure and evaluate the ability of 8-inch Spiroll Corefloor slabs to transfer horizontal shear through the grouted longitudinal joints without shear keys, as well as to determine the coefficient of friction, which served as a direct measure of the effectiveness of shear friction reinforcement in the end beams. The longitudinal joints were subjected to pure shear as the load was applied to the center slab while the exterior slabs were held in place. The shear strength was not tested to ultimate capacity, since a measure of the shear friction effectiveness was one of the desired objectives. After the joints were artificially cracked, the coefficient of friction was measured and was found to vary between 1.3 and 2.0. These values indicated that the reinforcement had performed satisfactorily and that the 1.0 value was conservative for planks with extruded edges.

A publication of the Concrete Technology Associates by Cosper, et. al. [98] reviewed hollow-core diaphragm test results for the shear strength of the grouted keyway between adjacent 12inch Dy-Core panels. The deck specimen consisted of four 22 ft. long by 12 in. thick DY-CORE planks. The three longitudinal joints were filled with shrinkage compensating grout. The edge beam at the end of the planks measured 9 in. x 12 in. x 16 ft. long and were cast in place. Reinforcement in the edge beams included six #5 longitudinal bars, four of which were anchored into the end core a distance of 24 in. Number four grade 60 U-bars were anchored around these four rods and into the core a distance of 24 in. An L-shaped #4 bar held the two bottom longitudinal #5 bars in position. Longitudinal shear loading was accomplished by applying a load against sixteen 1/2-inch prestressing strands, which were in an "x" arrangement across the seam. Parameters researched included the following: 1) the shear capacity of an uncracked grouted joint, 2) the effectiveness of shear friction reinforcement in transferring shear across a joint, 3) the ductility of the system after the bond between panels had fractured, and 4) the effect of cyclic loading on the system. The uncracked grouted seam demonstrated a high capacity in resisting lateral shear loads. Shear friction steel placed in the edge beam supplied adequate clamping forces once the seam had fractured. Ductility demands were satisfied as well, since the shear strength continued to increase after joint displacement. Finally, the diaphragm exhibited sufficient resistance to cyclic loading by maintaining a stabilized strength after repeated cycles above design

requirements.

Another experimental study, by Reinhardt and Hartjes [411], tested the joint between hollow-core planks under shear loading while subjected simultaneously to a normal force. Variable strengths of mortar and lengths of the grouted connection (0.3 to 2.1 meters) were accommodated for the single seam. Three different mortars were tested. The Lime-Cement mortar reached a maximum shear stress of 0.4 N/mm2 in a joint of 0.3 m length and also of the 2.0 m length. Clearly the strength of this very low-quality mortar did not depend upon the absolute length of the joint. The other two mortars showed higher shear stresses and an influence exercised by the joint length. The shorter the joint the higher was the maximum average shear stress. With better mortar quality the length influence became stronger. Joint length was found to have a significant influence upon shear stress at fracture for their particular testing configuration. Failures were characterized by brittle fractures of the bond at the mortar and grout interface.

Each of these tests used a slightly different testing frame. With such testing arrangements, however, the actual maximum shear is not simply the load divided by the contact area. A correction factor which accounts for the non-uniformity of the shear stresses must be used. Chow, Conway, and Winter state that the distribution of shear stresses in deep beams depart radically from that given by the ordinary, simple formulas [88]. Using finite difference, strain-gage measurements, and photoelastic measurements, Roarke and Young have tabulated the correction factor for various testing arrangements [416].

2.6 Reinforced Concrete Diaphragms

Lateral forces, typically produced by earthquakes or winds, are resisted by the use of a space frame system and/or shear walls. In either case, the lateral loads are transmitted from one wall to another through the floor system. The distribution of horizontal forces to the shear walls or space frame system depends on the properties of the diaphragm slab and the resisting system. The required diaphragm **st**iffness varies from one structural configuration to another. In-plane flexibility can be beneficial in a long roof supported on low, uncoupled concrete perimeter walls, due to dynamic attenuation of lateral ground accelerations. On the other hand, to improve the lateral force resistance of a multistory structure, it is desirable for the in-plane stiffness of floor and roof diaphragms to exceed the stiffness of the vertical subsystems to which they are connected [90].

Nakashima, Huang, and Lu [332], reported in 1982 an experimental work done at Lehigh University. This research considered reinforced concrete floor slab systems with edge beams

subjected to in-plane loads. The main objectives of their experimental study were to examine the strength, stiffness, and deformation characteristics of sample units of the floor slab system and to provide a basis for comparison with an analytical The testing program included three types of tests: the model. stiffness test, the strength test, and the repaired strength test. In the stiffness test, each specimen was tested as a whole unit to determine the elastic in-plane stiffness characteristics of the slab panels. In the strength test, each slab panel of the specimen was tested individually as a cantilever to examine its stiffness, strength and deformation characteristics in both the elastic and postelastic ranges. After the strength test the slab panel was repaired using an epoxy injection technique; it was then retested under the same loading conditions. A nonlinear finite element program was developed to analytically investigate the in-plane characteristics of floor slabs. As part of their conclusions, they stated the following: First, the elastic finite element analysis reasonably predicted the elastic in-plane stiffness of the test Second, in all strength tests, the development of the panels. major crack controlled the in-plane ultimate load. Cyclic loading or the application of the design service vertical load reduced the ultimate load by 15 to 25 percent. Third, stiffness of the test panels continuously degrade as load increased. Fourth, repair by the epoxy injection technique restored the test panels to their earlier strength, but the repaired panels were consistently less stiff than the original panels. Finally, the nonlinear finite element analysis successfully predicted the in-plane strength of the test panels and reasonably duplicated the experimental loaddeflection curves and crack patterns.

2.7 Hysteretic Models

A complete description of the behavioral characteristics of a structure throughout the plastic and elastic ranges can be obtained with a hysteretic model. This type of model predicts the force-displacement relation for a system utilizing stiffness and strength information. Riddell and Newmark have established a set of desirable characteristics for a hysteretic model. These features can be summarized as follows [413]:

- 1. Reality The selection of model parameters that are directly associated with known physical characteristics.
- 2. Accuracy The model must portray the measured results as closely as possible.
- 3. Simplicity The prediction should be completed with the simplest method possible.
- 4. Consistency The relationship between the response variable and any specific parameter should be consistent.

Many hysteretic models have been developed in the past. Each successive model has improved upon the first effort in some way. The characteristics of any reinforced or prestressed concrete model have become more refined and can be briefly stated as follows [354]:

- 1. The stiffness must change with the cracking of the concrete.
- 2. The loading stiffness in the second cycle is lower than the that of the first, and the load at peak displacement is nearly the same strength.
- 3. The average peak-to-peak stiffness decreases with the increase of the maximum displacement amplitude.
- 4. There is a tendency for very low incremental stiffness near the origin followed by a stiffening region (pinching).
- 5. Prestressed concrete is represented by a softening model.
- 6. The hysteretic curve for a prestressed structure has relatively small residual curvature.

The simplest type of hysteretic model is the bilinear model. A special case of this type of model is the elasto-plastic system. Many versions of the bilinear model have been developed; however, this type of model provides only a rough estimate. It contains no degradation properties and has an unrealistic transition point in the skeleton curve (a curve which describes the force-displacement relation obtained by increasing the magnitude of the load acting on the structure).

Jennings developed a general nonlinear hysteretic force displacement relation in the early 1960's which utilized closed form mathematical formulas [256]. The skeleton curve used a formula similar to that first proposed by Ramberg and Osgood [408] to describe relations between stress and strain. This model employed smooth rounded curves which were general enough to describe the behavior of systems ranging from linear to elastoplastic. The skeleton curve utilized the following relation:

$$x / x_{y} = p / p_{y} + \alpha (p / p_{y})^{r}$$
 (2-1)

where:

The ascending and descending branches of the hysteresis loop were defined respectively by:

$$(x+x_{o}) / 2x_{v} = (p+p_{o}) / 2p_{v} + \alpha[(p+p_{o}) / 2p_{v}]^{r}$$
(2-2)

$$(x-x_{o}) / 2x_{v} = (p+p_{o}) / 2p_{v} + \alpha[(p+p_{o}) / 2p_{v}]^{r}$$
(2-3)

where all the variables are the same as above and

- x_o = the displacement at the point where the loading is reversed, in.
- p_o = the load at the point where the loading is reversed, kips.

Degrading-type models introduced the effect of stiffness degradation caused by load reversals in inelastic ranges. Clough and Johnston [92] in 1966 proposed a degrading bilinear model, improving the elasto-plastic model by accounting for the stiffness degradation observed during the cyclic loading of reinforced concrete components.

Iwan [250] developed a similar model. This model was slightly more general than the previous one. It employed four types of behavior: elastic, elasto-plastic behavior, simple coulomb slip behavior, and irreversible or directional behavior. A complex set of definitions and integrations compose the major elements in this model.

The next type of model was developed by Takeda in the early 1970's and stands as one of the founding models. The Takeda model is based on experimental results and operates on a trilinear skeleton curve [471]. This primary curve is composed of segments connecting the origin, cracking point, yielding point, and ultimate point. The model contains many rules which govern the slopes of all the components. These guidelines mandate degradation, but not pinching action.

In the late 1970's, another model was developed to simplify the rules associated with the Takeda model and to account for the effects of pinching. This model, which is known as the Sina Hysteresis Model [431], contains a skeleton curve made up of three parts and is defined by nine rules. An example of this type of model can be seen in Figure 2 and the rules for this model are summarized in Figure 3.

Another modified version of the Takeda Model is the Otani Hysteresis Model [355]. This model was originally used to represent the stiffness variation of a joint spring in conjuntion with a flexural spring. The primary curve in this system is bilinear because the breakpoint is at yielding. Eleven rules are associated with this model, making it somewhat more complicated.

The final type of model employs many of the characteristics associated with the techniques previously described. This model is part of the Lumped Parameter Model Program, LPM, and is named the EKEH model (element #11) [158,159]. The Technical Coordinating Committee on Masonry Research (TCCMAR) has requested that this model be considered as part of the overall project. It has nonlinear, inelastic, degrading and pinching capabilities as is shown in Figure 4. The skeleton curve consists of a second order







Figure 3. Rules for modified Takeda model [431]





function and two linear segments with the following format:

$$F(e) = F_{p} * e / (F_{p} / k_{i} + \alpha | e |) -e_{p} < e < e_{p}$$

$$F(e) = F_{p}[(e_{m} - |e|)/(e_{m} - e_{p})]sign(e) -e_{p} < |e| < e_{t} (2-4)$$

$$F(e) = \beta F_{p} sign(e) -e_{t} > e > e_{t}$$

where

F(e)	= spring deformation force, kips.	
е	= spring deformation, in.	
ep	= deformation at peak strength, in.	
e _π	= intersection of the post peak strength	envelope line
	with the deformation axis (e Axis)	-
ß	= post peak strength lower limit factor.	
Fp	= peak strength, kips.	
k,	= initial stiffness, kips/in.	•
et	$= e_m - \beta (e_m - e_p)$	(2-5)
α	$= 1 - F_{p}/k_{i}e_{p}$	(2-6)

The post peak strength factor, ß, can be defined as the load after the usable strength has been achieved. The ß may be determined for each limit state mode. Unloading occurs along a degrading stiffness slope defined by:

$$K_{\mu} = K_{i} (F_{p} / K_{i} * e_{max})^{\gamma}$$
 (2-7)

where

 K_u = the degraded stiffness slope, kips/in.

 γ = a degradation stiffness constant.

 e_{max} = maximum displacement reached in all previous cycles.

The latter constant, γ , is defined from experimental test results and acceptable values range between 0.5 and 0.8. Values above 0.8 lead to nonconservative hysteresis loops. Skeleton curve for the EKEH/LPM model is shown in Figure 5.

Reloading occurs along a path controlled by the maximum force and the current pinch force. (This force is defined as in the Sina model). It starts from the point of zero force and continues through the pinch force until it intersects the straight line connecting the origin and point"a" (see description below). At this intersection, it follows the latter line segment until it intersects point "a" on the skeleton curve. The location of point "a" is defined by one of the following conditions:

1. If the peak strength has not been exceeded in all previous cycles, point "a" is defined as the point on the envelope that corresponds to the maximum displacement.



Figure 5. EKEH model skeleton curve [376]

2. If the peak strength has been exceeded, point "a" is defined by the deformation equal to the maximum deformation and a force equal to 0.8 times the force on the envelope curve that corresponds to the maximum deformation.

In other areas, loading and unloading follow the same basic rules developed in the models described earlier.

Many hysteretic models have been developed over time. Improvements in concrete models have been made to allow for stiffness degradation and pinching action. The Takeda model, and it subsequent improved versions, as well as the EKEH/LPM model represent the most comprehensive models currently in use. These models are in concordance with the four desirable characteristics established by Ridell and Newmark [413]. Experimental data associated with the EKEH/LPM model are given in the Appendix A, Section 8.3, Table A8, and Section 8.4, Table A9.

Additional work on the development of a hysteretic model for hollow-core plank diaphragms is currently being done at Iowa State University as part of TCCMAR Research program (Task 2.4a) [348].

3. DATA FORMULATION OF PREVIOUS WORK

3.1 General

This chapter addresses a detailed description of the experimental program done in diaphragms including the test rationale, brief description of the specimens, test setup, instrumentation, hydraulic actuation system, and test data acquisition for the following systems:

DIAPHRAGM SYSTEM

REFERENCE

1.	Cold-formed Steela)	Luttrell, et. al. [148,308]
	b)	Lukens [299]
2.	Wooda)	Young and
		Medearis [518]
3.	Steel-Deck-Reinforced Concretea)	Porter and
		Easterling [378]
4.	Wood, Cold-Formed Steel and	
	Steel-Deck-Reinforced Concretea)	ABK, A Joint
5.	Precast Prestressed Concretea)	Porter, et. al. [376]

3.2 Research performed by Luttrell, et. al. [148,308]

The following discussion of data formulation on cold-formed steel diaphragms was partially taken from Reference [148]. Experimental data result of this research is included in the Appendix A, Section 8.1.1, Tables A1 and A2 of this report. A series of 160 tests were performed and reported by Luttrell and Ellifritt at West Virginia University as a result of research on three general types of steel deck under various conditions of fastener arrangement, purlin spacing, gage, and material yield strength. This study was part of the support document for the Steel Deck Institute Diaphragm Design Manual [306,307]. Two major behavioral parameters were obtained, ultimate strength and shear stiffness. Ultimate strength, S_u, designates the total jacking force required to produce failure in a diaphragm divided by the length of the diaphragm in the direction of the applied load. Shear stiffness, G', is a measure of the relationship between inplane load and the deflection in the direction of that load. The shear stiffness is defined in Figure 6.

Three types of deck were considered: narrow rib ("A" deck), wide rib ("B" deck), and intermediate rib deck. Among the wide rib decks tested, there were two variations in the side lap arrangement: the standing seam side lap ("WB"), and the flat side lap ("W"). Tests were made on 16, 18, 20, and 22 gage decks with





lengths of 12, 16, and 20 feet. Panel widths tested were 18, 24, 30, and 36 inches. All tests were made on a horizontal cantilever test frame. The connections between the perimeter members of the frame were made with light clip angles and considered as pinned. The entire frame was supported on rollers to eliminate the possibility of developing frictional resistance during deformation. Purlins were fastened to the frame with pinned connections and spacing was variable. Welds were made with E6013 1/8" diameter electrodes with sufficient heat for fusion. Various weld arrangements were used.

The loading apparatus for all tests consisted of one hydraulic jack and load cell arrangement in line with the center line of the south edge member at the southwest corner. A tensile load was applied by means of a high strength rod threaded through the reaction frame and connected to the edge beam at a level where the diaphragm attaches to the frame. Load was applied in increments from zero to failure with deflection measurements made at each stage of loading. Deflections were measured with Ames dial gages accurate to 0.001" at all corners in the plane of the diaphragm as shown in Figure 7. From these measurements, the true diaphragm

$$\Delta = \Delta_4 - [\Delta_2 + (\Delta_1 + \Delta_5) a/b]$$
(3-1)

where Δ_1 , Δ_2 , Δ_4 , and Δ_5 are measured movements at the corners in inches. In each test, the diaphragm was loaded to failure, which was initiated in a variety of ways. Weld failure was generally because of the sheet tearing away from the weld. Although unusual, welds sometimes separated from the perimeter beam while still attached to the sheet. If the lip on the male rib is small, as in the narrow rib decks, sudden local buckling of the lip leads to overall buckling of the flute. If the return on the male rib is large, as in the wide rib decks, the flute will generally fail as a slender compression strut. In many of the diaphragms tested, weld failure and buckling were very closed allied and an assignment of precedence to one failure over the other was difficult. In general, all the modes of failure were present to some degree. The failure of a weld or the buckling of a flute did not necessarily mean that the ultimate diaphragm load had been reached. However, the additional increase in load after an initial failure of this type usually did not exceed ten percent.

3.3 Research performed by Lukens [299]

The following discussion of data formulation on cold-formed steel on wood frame diaphragms was taken from Reference [299]. Experimental data result of this research is included in the Appendix A, Section 8.1.2, Table A3 of this report. Twenty-two full-scale cantilever tests on steel clad, timber-framed diaphragms were performed at Iowa State University by Lukens. Ultimate




strength, S_u , and shear stiffness, c, were evaluated. Variables included reverse loading, types of wood for purlins, purlin to chord connections, stitch screws and length. Ultimate strength, S_u , is defined as the ultimate load at which a panel will not carry any additional load for a given period of time (approximately two to three minutes). The shear stiffness, c, is defined in Figure 8.

Each panel was constructed with three sheets of Grandrib 3, manufactured by Fabral, Lancaster, Pennsylvania. These sheets were 29 gage galvanized steel with a coated thickness of 0.0172 in. and base metal thickness of 0.0135 in. The minimum yield strength of the base metal was 80 ksi. The sheets spanned the length of the panel. In all panels except Panel 13, sheets were fastened in valleys with 1.0 in. long, #10 wood-grip, self-tapping screws. The screws at the seams of the sheets were fastened only through one sheet. In addition to these screws, panels 20-22 were fastened at the seams of the sheets with 0.5 in. long, #8 self tapping stitch screws. These stitch screws were spaced approximately 12 in. on center through the ribs of the two sheets at the seams. Panel 13 was fastened with 16d threaded galvanized steel nails. Nails were hammered through the ribs.

The timber-frames were constructed with 2x8 in. top chords and 2x4 in. purlins as shown in the Figure 9. The chords for Panels 1-4 were #2 and better spruce-pine-fir, and the chords for panels 5-22 were #2 and better Douglas fir-larch. The purlins for Panels 1-9 were standard and better spruce-pin-fir, and the purlins for Panels 10-22 were standard and better Douglas fir-larch. In all the panels, except Panels 2 and 3, the purlins were nailed on edge to the chords with one 60d spike and 10d toenails. The purlins for Panels 2 and 3 were nailed on edge to the chords with one 60d spike at all location plus one Simpson H-1 anchor nailed to the purlin and the chord at left corners C and D. The length of each panel was measured from the outside faces between the first and last purlin. The different lengths for the panels were 6, 8, 12, 16 and 20 ft. with each length having the purlins spaced equally at 23.5, 23.625, 23.75, 23.813 and 23.85 in. on center, respectively. The width, a, of the panels was the distance between the centerline of the two chords. All panels had a width of 9 ft.

A cantilever testing procedure was used to calculate shear stiffness and strengths. All panels, except Panels 2, 3, and 13 were tested twice (Test A and Test B). During Test A, panels were loaded from zero to failure, which was a split in the chord at corner C and D caused by edge purlins pulling or pushing across the chord. After Test A was completed, four 3x3x3/2 in. angles 1.5 in. long were screwed into the chord and the purlin at corners C and D. These angles repaired and strengthened the purlin to chord connection so that it would not fail during Test B. In Test B, this strengthened panel was loaded beyond the failure of Test A to a new failure.









Load, P, was applied to the chord at Corner A with a turnbuckle and recorded from a 5000 lb. capacity dynometer, except for Panel 20 Test B. The load for Panel 20 Test B was recorded from a 10,000 lb. capacity load cell. For Test A with lengths 12 ft. or greater the load was generally applied in 100 lb. increments to failure with the load released every 500 lb. to record permanent deflection. For lengths less than 12 ft. the load was generally applied in 50 lb. increments to failure with the load released every 250 lb. to record permanent deflection. For Test B the load was applied in about 500 lb. increments to the Test A failure load. Then the load was applied in 100 lb. increments to failure for lengths less than 12 ft. and in 200-300 lb. increments to failure for lengths 12 ft. and greater. Deflections were recorded every load increment.

Panels 6-9 were subjected to reversing loads. Reverse loading was started by pulling the panels at Corner B in 100 or 200 lb. increments to 1100 lb. for Panels 6, 7 and 8 and to 700 lb. for Panel 9. The 1100 lb. and 700 lb. loads were approximately equal to 40% of the strength of a 12 ft. panel with spruce-pine-fir purlins for Test B and A, respectively. Then the load at Corner B was released to zero and applied to Corner A in the same manner. After a given number of cycles, load was applied at Corner A to failure. Panel 6 had 10 cycles, Panel 7 had 9 cycles, Panel 8 had 20 cycles, and Panel 9 had 20 cycles. This load history completed Test A. Test B loaded the same as the non-reverse loading panels.

The cantilever test apparatus is shown in Figure 10. The chord at Corners C and D was fixed to the floor and the chord between Corners A and B was placed on rollers. Deflections were recorded from dial deflection gages.

The shear deflection, δ_s , was calculated with the following equations:

$0_{\pi} = 0_1 - 0_2 - (0_3 + 0_4) d/D $ (3-	(3-2)
--	------	---

 $\delta_{\rm B} = {\rm Pa}^3/3{\rm E_p}{\rm I} \tag{3-3}$

$$\delta_{\rm s} = \delta_{\rm T} - \delta_{\rm B} \tag{3-4}$$

where

δ_	= net total deflection at Corner A, in.
δ	= bending deflection at Corner A, in.
δ,	$\delta_2, \delta_3, \delta_4 = deflection$ gauge reading at corners, in.
₽ [¯]	= load applied to panel, lb.
E,	= modulus of elasticity of edge purlins, psi
I.	= second moment of inertia of edge purlins about
	centerline of panel, in ⁴ .





3.4 Research performed by Young and Medearis [518]

The following discussion of data formulation on wood shear walls was taken from Reference [518]. Experimental data result of this research is included in the Appendix A, Section 8.2.1, Table A6 of this report. A series of eight plywood shear wall tests were performed by Young and Medearis as part of their work on the structural damping characteristics of composite wood structures subjected to cyclic loading. Variables included plywood thickness, precycling, nailing, and re-nailing.

Each 8x16 ft. panel tested consisted of two 8x8 ft. wood walls joined side by side by 3-5/8 in. machine bolts at 3 ft. c-c. All tests were made on a vertical test frame. The connections between the panel and the steel frame was made with one 3x3x1/2 in seat angle with 3/4 in. stiffener plates on each corner as shown in Figure 11.

Each 8x8 ft. plywood wall was constructed with plywood on both sides except for Test 7 which used plywood on one side. Typical 2x6 in. studs and 2x6 in. blocking was used. For the exterior side connecting to steel frame 3x6 in. redwood plate was used and for interior side double 2x6 in. plate was used. See Figure 12 for typical plywood framing. All plywood was exterior grade and the framing material was No. 1 construction grade. The studs were nailed to redwood plate with 2-10d toenails. The studs were nailed to double plate with 2-20d toenails. The double plate was nailed with 16d toenails at 9 in. c-c. The plywood was nailed in the perimeter with 8d at 4 in. c-c for Test 1, 2, 3, and 7, 8d at 6 in. c-c for Test 4, 10d at 6 in. c-c for Tests 5 and 6, 6d at 4 in. c-c for Panel 8. The plywood was nailed to redwood with 8d at 4 in. cc for Panels 1, 2, 3, 4, and 7, 10d at 4 in. c-c for Panel 6, and 6d at 3 in. c-c for Panel 8. The plywood was nailed to double 2x6 in. plate with 8d at 4 in. c-c for Panels 1, 2, 3, and 7, 8d at 6 in. c-c for Panel 4, 10d at 6 in. c-c for Panels 5 and 6, and 6d at 4 in. c-c for Panel 8. The plywood thickness was 3/4 in. both sides for Test 1, 3/8 in. both sides for Tests 2, 3, 4 and 8, 1/2 in. both sides for Tests 5 and 6, and 3/8 in. one side only for Test 7.

Young and Medearis used a conversion, by means of symmetry, from single-span action to that of two cantilevers whose origins are at the single span center line; and rearranging of the cantilevers to give the final test interpretation, that of two cantilever shear walls, side by side, and subjected to a total shear load, P_c . (Figure 13).

The testing machine frame, was equipped with two special channel end supports designed to take reactions from reversed loading. Cyclic loading was applied to the double shear wall, through a vertical channel on the axis of symmetry, by hydraulic



Figure 11. Typical plywood framing [518]













jacks. Fifty-kip capacity SR-4 load cells were used in conjunction with the jacks. The experiments were conducted by using peak load increases, from one cycle to another, of 4 kips. The load was applied in increments of 1/4 the peak, or maximum, load to be attained per given cycle and unloaded using 1/2 the maximum load.

The true test panel center deflection was determined by using three Ames dial gages, one at each end support, and one at the center. The support gages were utilized to account for movements of the deflection reference line resulting from end-support crushing of the wood. The true center deflection, Δ_c , was obtained from the following equation:

 $\Delta_{\rm c} = \Delta'_{\rm c} - (\Delta_{\rm L} + \Delta_{\rm R}) / 2 \tag{3-5}$

where

The larger deflections of the panels, resulting from aboveyield loading, always exceeded the capacity of the center dial gage. Thus, to measure these high-load deflections, a transit was used in conjunction with a 50 divisions/inch scale which was mounted on the center loading channel.

Test 3 (identical to Panel 2) was planned to investigate panel deterioration effect, as a result of being subjected to cyclic loading at the design load prior to subsequent cyclic loading to ultimate. Test 6 was planned to investigate the possibility of renailing a panel that had already been subjected to a near ultimate load. This was accomplished by taking the previously tested #5 panel, jacking it back to its neutral configuration, and then renailing it.

3.5 Research performed by Porter and Easterling [378]

The following discussion of data formulation on SDRC diaphragms was taken from Reference [378]. Experimental data result of this research is included in the Appendix A, Section 8.3.1, Table A8 of this report.

A series of 32 full-scale SDRC diaphragms were tested at ISU as part of a two phase research. The first phase included Tests 1-9 and was performed by Porter and Greimann [390]. The second phase included Tests 10-32 and was performed by Porter and Easterling [378]. Variables included deck type, deck thickness, fastener type, number of fasteners, concrete thickness, depth to span ratio, load combinations of in-plane and gravity loads and edge member size. Each of these parameters were investigated with regard to their influence on behavior and strength of SDRC diaphragms.

Numerous SDRC diaphragm parameters were varied and tested on the experimental sections of the study. Diaphragms 1-21 were 15 ft. x 15 ft. in plan while Diaphragms 22-32 were 15 ft. x 12 ft. in plan. Diaphragms 1-31 were constructed with W24x76 steel sections and Diaphragm 32 was constructed with W14x22 steel sections. All diaphragms except diaphragm 26 were constructed with normal weight concrete, with 26 being constructed with structural lightweight concrete. Deck sections were classified as different if the profile, deck thickness or embossment configuration is unlike any other.

A horizontal cantilever test frame was designed in the first phase of the research [30] and is shown schematically in Figure 14. The north to south dimension of the span of the diaphragm was the 12 ft. dimension. The south edge of the test frame was constructed using three reinforced concrete blocks. These blocks served as the reaction edge and were anchored to the structural test floor by post-tensioning 2 in. diameter high strength steel rods. Connection of the framing members to each other and to the south edge was made using flexible tee sections.

Load was applied via two reversible hydraulic actuators as shown in Figure 14. Each actuator has a capacity of approximately 200 kips and both are driven with a closed loop servo-valve controlled system. Six diaphragms were loaded with both in-plane and vertical load [335]. Vertical load was applied to 20 distributed neoprene pads on the surface of the specimen. The amount of vertical load applied was chosen to model an equivalent distributed load based on equivalent shear area in the one-way direction (parallel to the corrugations).

Data were collected for each of the diaphragms utilizing mechanical dial gages, electrical resistance strain gages, electronic displacement transducers (DCDT), and load cells. The electrically recorded data were collected using a 150 channel data acquisition system (DAS). Components of the system are microcomputer, digital plotter and printer and a 150-channel digital voltmeter with five independent power supplies. Figure 15 shows a schematic of the experimental testing arrangement.

Reversed cyclic loading was used for all test specimens in the second phase of the research. The tests were displacement controlled using the DCDT in the northeast corner in line with the push beam as the feedback to the closed loop system. At each level of displacement in the displacement history a minimum of three complete cycles was made. The criteria for increasing to the next level was that the load had to stabilize within a certain margin. This margin was defined as being less than a five percent change in load from the previous cycle at the same displacement. An initial displacement level of 0.025 in. was selected for the experimental



Figure 14. Diaphragm test frame schematic [378]

STRAIN GAGES

DISPLACEMENT TRANSDUCERS





program. This displacement was thought to be within the elastic range of behavior for the SDRC diaphragms.

Initial or elastic stiffness values for each diaphragm were determined. These values were determined based on the first nominal displacement to 0.025 in. The total load for both actuators was divided by the displacement of the controlling DCDT in order to obtain the initial stiffness. No adjustments to the experimental stiffness values for test frame stiffness were made, since pre-test frame calibrations without an attached diaphragm, showed the frame to be relatively flexible.

Load and displacement were continuously monitored at intervals of approximately one second during the displacement histories. This enabled the determination of peak load and corresponding displacement between load points. No adjustment to strength values based on test frame strength, without an attached diaphragm, were made, since frame calibration indicated that the load carried by the base frame was approximately 1 kip. This contribution was deemed negligible.

3.6 Research performed by ABK, a Joint Venture [2,3]

The following discussion of data formulation on wood, steel deck, and SDRC diaphragms was taken from Reference [2]. Experimental data result of this research is included in the Appendix A, Section 8.1.3, Table A4, and Section 8.2.2 Table A7 of this report.

A series of 14 configurations of 20 ft. by 60 ft. diaphragm specimens were tested by ABK A Joint Venture. The testing was conducted at the El Segundo Structures Laboratory of the North American Aircraft Division of Rockwell International Corporation. From 14 tests performed, 11 were on wood diaphragms, two were steel deck diaphragm, and one was steel deck concrete filled diaphragm.

Four 20 ft. x 60 ft. frames were constructed to accommodate the seven prime wood sheathed diaphragm configurations. These had 4" x 12" edge and end members with 2" x 12" joists at 24 inches on center as shown in Figure 16.

All framing lumber was 2" x 12" and 4" x 12" Douglas Fir No. 1, and the blocking was 2" x 4" Douglas Fir No. 2. The lumber sheathing was 1" x 6" Douglas Fir, Construction Grade. The straight or diagonal sheathing boards were laid so that there were not less than two boards between joints on the same bearing and not less than two bearings between joints in adjoining boards. These were joined with three 8d nails at ends of each board and two 8d nails at all other bearings. The plywood sheathing conformed to U.S. Product Standard PS-1-74, Structural I, Exterior Glue, Grade C-D, Douglas Fir. Common wire nails were used for the most part,





and duplex head nails were used where they did not interfere with overlaying material. Power driven "short" plywood nails had the same gage as for specified nail.

Diaphragm C was constructed of 1/2 inch thick plywood, unblocked, with built-up roofing applied. The plywood sheets were laid across the joists and staggered. The plywood was nailed with 8d nails at 6 inches on center at the edges of the sheets and at 12 inches on center for intermediate bearings. 8d nails at 6 inches on center were used to connect the edges of the plywood sheets to the 2" x 4" blocking along the 4" x 12" chord members to provide an unchorded condition.

Diaphragm D was a rework of Diaphragm C. After Diaphragm C was tested, about 12 ft. of the roofing at each end was rolled back to observe the plywood sheathing. Any nail torn out of the edge of the plywood was replaced by another nail into adjacent solid plywood. The roofing was laid back in place and renailed with roofing nails through the roofing 6 inches o.c. along the end 20 feet of the two edges and across the 20 ft. ends, and field nailed at 18 inches o.c. each way for 20 ft. at each end. In addition, the edges of the plywood were nailed to the 4" x 12" chords with 6d nails at 6 inches o.c. through the roofing.

Diaphragm B was a rework of Diaphragm D after it was tested. Five full sheets and five half sheets at the two ends of the diaphragm were replaced and nailed. In addition, the other plywood sheets within 20 feet of the ends of the diaphragm were renailed. The chord nailing was also replaced with 8d nails at 6 inches o.c.

Diaphragm E was constructed of 1" x 6" straight lumber sheathing with built-up roofing. The sheathing was laid across the joists in the long dimension of the diaphragm. The edge board was nailed to the 2" x 4" blocking rather than the 4" x 12" chord to simulate an unchorded diaphragm. Prior to testing, the roofing was nailed with roofing nails at 6 inches o.c. to the edge and end 4" x 12".

Diaphragm E1 was a retest of Diaphragm E by renailing the roofing for 20 ft. at each end with roofing nails through 2-3/4 inch diameter by 30-gage metal washers. Renailing was at 6 inches o.c. along the edges and ends and at 18 inches o.c. each way in the interior field.

Diaphragm H was a rework of Diaphragm E1 after it was tested. The entire diaphragm was overlain with 5/16 inch thick plywood. The plywood was nailed through the sheathing with 8d nails at 6 inches o.c. at the ends of the sheets and at 12 inches o.c. at the intermediate bearings. The exterior edges of the plywood were nailed to the 4" x 12" chord with 8d nails at 6 inches o.c. to provide a chorded condition. Diaphragm I was constructed of 1" x 6" lumber sheathing laid at 45 degrees diagonally to the diaphragm and covered with built-up roofing. The boards were nailed to the 2" x 4" blocking rather than the 4" x 12" chords to simulate an unchorded diaphragm. Prior to testing, the roofing was nailed with roofing nails at 6 inches o.c. to the edge blocking and the end 4" x 12".

Diaphragm I1 was a retest of Diaphragm I by renailing the roof for 20 ft. at each end with roofing nails through 2-3/4 inch diameter by 30-gage metal washers. Renailing was at 6 inches o.c. along the edges and ends, and at 18 inches o.c. each way in the interior field.

Diaphragm K was a rework of Diaphragm I1 after it was tested. The broken boards were replaced and all broken and bent nails were renailed. 1" x 6" straight lumber sheathing was laid across the joists in the long dimension of the diaphragm. The straight sheathing was nailed through the diagonal sheathing with the specified nailing.

Diaphragm N was constructed of 1/2 inch thick plywood. Flat 2" x 4" blocking was provided between the joists along the edges of the plywood sheets. The plywood was nailed at all edges with 8d nails at 4 inches o.c. and at 12 inches o.c. to the intermediate bearings. The nailings to the 4" x 12" edge members provided a chorded diaphragm.

Diaphragm P was constructed on the framework of Diaphragm N after it was tested. The 1/2 inch plywood was removed and two layers of 3/4 inch plywood were nailed to the frame. The lower sheets were laid across the joists as in the previous case of the 1/2 inch plywood diaphragm. These sheets were nailed at all edges with 8d nails at 4 inches o.c. and at 12 inches o.c. to the intermediate bearings. The upper sheets were laid at right angles to the lower sheets and nailed at all edges with 8d nails at 4 inches o.c. and at 12 inches o.c. in rows 2 feet apart. The layout of the sheets was slightly offset from that in Diaphragm N to allow end nailing into areas of the 2" x 12" joists that had no previous nailing.

A separate 20 ft. x 60 ft. structural steel frame was provided for each of the two prime steel deck diaphragm configurations. These had W12x27 edge, end, and interior beam members and W12x19 joist members as shown in Figure 17. This provided a three-span condition for the steel decks.

The structural steel conformed to ASTM A36, and the bolts conformed to ASTM A307. ASC Pacific, Inc. - Type B (interlock), 20 gage, galvanized, was used for the unfilled deck diaphragm and ASC Pacific, Inc.-type B(Hiform), 20 gage, galvanized, was used for the diaphragm with concrete fill. The connections at the support members were 1/2 inch diameter effective plug welds at each flute. For the marginal and parallel members, 1/2 inch diameter effective



Figure 17. Steel diaphragm testing [2]

44

plug welds at 12 inches on center were used. The side laps were button-punched at 18 inches on center.

The concrete fill was standard weight concrete with an ultimate compressive strength $f'_{\rm c}$ of 3000 psi at 28 days. The concrete was reinforced with 6" x 6"/W1.4xW1.4 electric welded mesh. The concrete was covered with an impervious membrane immediately after placing.

Diaphragm Q was constructed of twenty-four 30-inch wide by 1-1/2 inch deep, 20-gage steel deck panels laid over the structural steel framework. The deck was attached to the steel frame members with the specified plug welds and the seams were button punched 18 in. on center. The splice bars were not attached to simulate an unchorded condition.

Diaphragm R was a rework of Diaphragm Q after it was tested. Broken plug welds were replaced or equivalent fillet welds were placed on the end of the deck panels at the edge (chord) members of the framework. The seams were button punched at 6 inches on center for the entire diaphragm. The splice bars were bolted in place to provide a chorded condition.

Diaphragm S was constructed of twenty-four 30-inch wide by 1-1/2 inch deep, 20-gage steel deck panels laid over the structural framework. These panels had embossments along the webs of the flutes to provide composite action with the concrete fill. The deck was attached to the steel frame members with the specified plug welding and the seams were button punched as specified. The splice bars were bolted in place to provide a chorded condition. The specified electric welded wire mesh was laid on the deck and 2-1/2 inches of concrete was placed over the top of the flutes. The concrete was screeded and bull floated to grade and covered with a black polyethylene film to retain the moisture for curing.

The test setup for the quasi-static, in-plane displacement of the diaphragms is shown in Figure 18. The diaphragm specimens were supported on eight low friction roller assemblies and were displaced, in-plane, at the two ends by programmable hydraulic actuators. For the quasi-static tests, removable reaction pillars were moved into place at the diaphragm's third points and the diaphragm was cyclically displaced in its plane by the hydraulic actuators.

The basic instrumentation for the measurement of the diaphragm responses and forcing functions is shown in Figure 19 and consisted of displacement sensors, accelerometers and load cells; and all of the instrumentation measured in-plane responses and forcing functions. The displacements were measured using string potentiometers. The displacement sensors, including feedback deflection sensors, were mounted to a stable reference frame that was independent of the frame for the forcing system. This type of



Figure 18. Schematic of the test setup for quasi-static testing of diaphragms [2]



Figure 19. Diaphragm instrumentation [2]

instrument mounting was used to eliminate the need to account for the flexibility of the actuator reaction stands. The data from each instrument was recorded on magnetic tape in digital form.

The quasi-static loading cycles were normalized in terms of multiplicative factors of the basic static deflection delta of each diaphragm. Diaphragms E and E1 had deltas of 0.16 inches. Diaphragms H, C, D, B, N, Q, and R, had deltas of 0.4 inches. Diaphragms I and I1, had deltas of 0.72 inches. Diaphragm K had delta of 0.3 inches. Diaphragm P had delta of 0.25 inches, and Diaphragm S had delta of 0.1 inches. The delta for each diaphragm corresponds to an estimate of the approximate elastic limit of the diaphragm.

The test results reported by ABK [2] includes measured data from instrumentation. Other reports presented by ABK [3] includes the interpretation of diaphragm tests and give us initial stiffness, strength and pinch force for 12 of 14 diaphragms tested.

3.7 Research performed by Porter, et. al. [376]

Experimental data result of this research is included in the Appendix A, Section 8.4.1, Table A9 of this report.

Sixteen hollow-core diaphragm tests have been conducted at the Structures Laboratory at Iowa State University. The configuration of the test slabs is shown in Figure 20. Each of these tests employed connecting the planks to a cantilever steel I-beam loading frame with Nelson studs and high strength grout. A list of the parameters involved with each test is found in Table 1.

All slabs were subjected to the sequential phased displacement program as presented in Figure 21. Complete hysteretic curves were recorded for perimeter displacements and seam slips at several locations. One of these hysteretic curves is shown as an example in Figure 22. An envelope of the entire hysteretic behavior was established for both the initial (virgin) and the stabilized curves, as shown in Figure 23.

Comparisons were made between the significant limit state strengths associated with the first major event (FME) and other key events (or modes of failure), stiffness, displacements, and peak strengths.

The plank orientation was determined to be one of the most important parameters affecting the strength of the untopped diaphragms. The perpendicular orientation provide for higher FME load, and peak strength. For the topped slabs the orientation effects were dependent on the number of sides connected to the loading frame. The tests with two sides connected produced greater FME and peak load, while the tests with four sides connected did



Figure 20. Typical dimensions for all diaphragms (except Test #3) [481]



DISPLACEMENT, In.

Figure 21. Example of SPD load program [481]



Figure 22. Example of hysteretic curves [481]



רסעם' אוש

Figure 23. Example of envelope curves [481]

Test No	Sides Connected	Orientation	Topping	Plank Depth (in.)
1	2	EW	NO	8
2	2	EW	NO	8
3	2	NS	NO	8
4	2	NS	NO	8
5	4	NS	NO	8
6	4	EW.	NO	8 .
7	3	EW	NO	. 8
8	2	EW	NO	8
8B	2	EW	YES	8
9	2	NS	NO	6
10	4	EW	NO	6
11	2	NS	NO	12
12	2	NS	YES	8
13	4	NS	YES	8
14	4	EW	YES	8
15	4	NS	NO	8

Table 1. SUMMARY OF PARAMETERS FOR THE DIAPHRAGM TESTS [376]

Note

1)

- Test #15 had 15 weld ties per seam. Topped tests did not have any weld ties. 2)
- 3) Untopped tests (with the exception of #15) had 3 weld ties per seam.

ین میں میں میں میں کی کہ کہ کہ این میں میں میں میں جو این اور میں این میں میں میں میں میں این این اور اور اور

All two sided tests, (except Test #2), were connected to the loading beam and restrained end. 4)

not show any significant change. This orientation behavior is attributed to the fact that for untopped slabs, oriented in the weak direction, the seam fails prematurely before the planks have attained their full capacity; thereafter, the planks act separately leading to a reduction in the peak load. For the topped slabs, the topping prevents the seam failure; therefore forcing the planks to act as one diaphragm, as in the case more like an ordinary reinforced concrete diaphragm.

The addition of the two-inch topping to the planks increased the diaphragm action and produced a stiffer diaphragm. The largest change in the FME load was attained for slabs connected on all four sides. The weak orientation (EW - planks oriented parallel to the applied load) exhibited a bigger increase in the FME and peak loads compared with the increase of the strong orientation (NS). Also, the planks with four sides connected to the loading frame experienced a larger increase than those with only two sides connected.

Increasing the thickness of the planks produced larger effects for the weak orientation, while a less noticeable change was recorded for the strong orientation. The increase of the number weld ties produced a stronger diaphragm, especially for the peak load level.

In order to establish the shear and tension strength of the seams, a series of elemental tests were performed. The results of these tests were reported to the Third Meeting of the U.S.-Japan Joint Technical Coordinating Committee on Masonry Research.

An analytical research on the development of a hysteretic model for hollow-core plank diaphragms is being conducted at Iowa State University as part of TCCMAR research program (Task 2.4a) [348].

4. INITIAL STIFFNESS AND MAXIMUM LOAD PREDICTION

4.1 General

A complete description of the behavioral characteristics of a structure throughout the elastic and plastic ranges can be obtained with a hysteretic model. This type of model predicts the forcedisplacement relation for a system utilizing stiffness and strength information. This chapter addresses the equations required to predict the initial stiffness and peak strength for steel deck reinforced concrete (SDRC), cold-formed steel, precast prestressed, and plywood diaphragms.

4.2 Cold-Formed Steel Diaphragms

The following discussion of initial stiffness and peak load prediction was taken from Reference [307]. The design recommendations in the following paragraphs are limited to properly interconnected diaphragm panels having thicknesses to between 0.014 inches and 0.064 inches with panel depths D between the nominal limits of 9/16 inches and 3 inches.

4.2.1 Diaphragm Strength

The shear strength of a diaphragm system can be limited by the strength of the connections, local buckling in the panels, or by general plate-like buckling of the whole diaphragm area.

4.2.1.1 Strength of the Connections

The shear strength of cold-formed steel diaphragm, when controlled by the strength of connections is limited to the smaller value from Equations 4-1, 4-2, or 4-3. Consider the diaphragm in Figure 24 where three panels are shown. The panels may be connected to the support frame by structural connections, having a strength Q_f , along any structural member. Stitch or sidelap connections, having a strength Q_s , may be installed along the dashed lines to form sheet-to-sheet connections away from cross supports.

Edge Fasteners

Figure 25 represents the edgemost half panel and the forces transferred to it from P as the ultimate value P_u is approached. For edge fasteners the strength of the diaphragm is given by

$$S_u = P_u/L = (2\alpha_1 + n_p\alpha_2 + n_e)Q_f/L$$
 (4-1)



Schematic layout for diaphragm [307] Figure 24.





where

W

- P_u = diaphragm strength, kips
- S_u = ultimate shear, kips/ft.
- n_e = number of edge connectors between cross supports (n_e = 3 in Figure 25)
- $\alpha_1 = (1/w)\Sigma x_e$, end distribution factor with summation across full width w
- x_e = distance from panel centerline to any fastener in a panel at the end support, in.
 - = panel width, in.
- α_2 = purlin dist. factor similar to α_1
- L = panel length, ft.
- $n_{p_{1}}$ = number of purlins excluding those at ends or end laps where connection patterns may differ.
- Q_r = structural fastener strength, kips (see Section 4, Reference [307])

Interior Panels

Figure 26 shows a free body of an interior panel where Q_s represents a sidelap (stitch) connector and P_uw/L is the transferred end-member axial force. For interior panels the strength of the diaphragm is given by

$$S_{u} = P_{u}/L = \left[2(\lambda-1) + n_{s}\alpha_{s} + (1/w^{2})(2n_{p}\Sigma x_{p}^{2} + 4\Sigma x_{e}^{2})\right]Q_{f}/L \quad (4-2)$$

where

Q_s = stitch connector strength, kips (see Section 4, Reference [307])

 n_s = number of stitch connectors within the length L.

 $\alpha_{\rm s} = Q_{\rm s}/Q_{\rm f}$

 $\lambda = 1 - DL_v / [240(t)^{0.5}]$

D = panel depth, in.

 L_v = purlin spacing, ft.

A = 2 for double edge fasteners.

- A = 1 for single fasteners at panel edges.
- L = panel length, ft.

End Members

The fastener at panel corners limit S_u to:

 $S_{ii} = [N^2 B^2 / (L^2 N^2 + B^2)]^{0.5} Q_f$

(4-3)

where

N = number of fasteners per foot along ends. $P = P = \frac{1}{2} \left(\frac{1}{2} \left(2P \sum_{i=1}^{N} \frac{1}{2} + \frac{1}{2} \sum_{i=1}^{N} \frac{1}{2} \right) \right)$

 $B = n_{s}\alpha_{s} + (1/w^{2})(2n_{p}\Sigma x_{p}^{2} + 4\Sigma x_{e}^{2})$



Figure 26. Interior panel force distribution [307]

4.2.1.2 Strength Controlled by Panel Buckling

The probability of plate-like shear buckling, is small for most common installations. For relatively ideal corrugated diaphragms, Easley [139] has presented an approach to the critical shear load. That approach is modified at Reference [307], conservatively treating the limiting case as being controlled by two end spans, resulting in a critical load of:

$$S_{c} = (12.95 \times 10^{3} / L^{2}) (I^{3} t^{3} d/s)^{0.25}$$
(4-4)

where

 $S_c = critical load, kips/ft$ I = panel moment of inertia, in.4/ft. of width. d = corrugation pitch, in. s = developed flute width = 2(e + w) + f, in. L = design length, ft.

4.2.2 Diaphragm Stiffness

The stiffness of cold-formed steel deck diaphragms can be obtained with the following equation

$$G' = E t / [2(1 + v)s/d + \varphi D_n + C]$$
(4-5)

where

E = modulus of elasticity, kips/in²v = Poisson's ratio, 0.3 D_n = warping constant (see Appendix IV, Reference [307]) s = girth of corrugation per rib, in. d = corrugation pitch, in. t = base metal thickness, in. = purlin effect factor on warping (see Table 3.3-2, φ Reference [307]). = connector slip parameter. С (4-6) $C = (E t/w)S_{f}[24L/(2\alpha_{1} + n_{p}\alpha_{2} + 2n_{s}S_{f}/S_{s})]$

where

= panel width, in. Ŵ

- S_f = structural connection flexibility, in/kip
- S_s = sidelap connection flexibility, in/kip
- L = panel length, feet.
- E = modulus of elasticity, 29500 kips/in²

4.3 Steel Deck Reinforced Concrete Diaphragms

The following discussion of initial stiffness and peak strength prediction was taken from Reference [378]. In this Reference equations were developed to predict the initial stiffness and ultimate strength of SDRC diaphragms. These equations were based on an edge concept, which considered the force to be transferred from the load frame into the diaphragm within a relatively narrow region adjacent to the framing members. Elastic edge force distributions were used to determine the initial stiffness expressions and inelastic edge force distributions were used to determine the ultimate strength expressions.

4.3.1 Diaphragm Strength

The strength of a SDRC diaphragm was limited by one of the three primary failure modes which are a diagonal tension failure of the concrete, a failure of the edge fasteners or a failure of the shear transfer mechanism. The limiting predictive capacity of the diaphragm will be taken as the minimum value from Equations 4-7, 4-21, or 4-22.

4.3.1.1 Diagonal Tension

The strength of SDRC diaphragm, when controlled by diagonal tension failure of the concrete, was determined in Reference [390] by using the ACI [524] shear wall equation. For application to SDRC diaphragms the tensile strength was taken as $4/f'_{\circ}$ and the effective depth was taken as the total depth minus two times the edge zone distance which could be taken conservatively as 80 percentage of the depth. Thus the equation is

$$V = 3.2t_{b}/f'_{c}$$

where

V = shear strength of composite diaphragm, lbs.

 $t_e = t_c + n_s t_s (d/s) = effective composite web thickness, in.$

t_c = average concrete thickness considering ribbed geometry, in.

(4 - 7)

- n_s = shear modular ratio of steel deck to concrete.
- $t_s = steel deck thickness, in.$
- d = corrugation spacing, in.
- s = developed width around one corrugation, in.

f'_c = concrete compressive strength, psi.

b = depth of diaphragm, in.

4.3.1.2 Edge Fasteners

The capacity of a SDRC diaphragm based on a fastener failure utilized the edge force distributions obtained from a finite element analysis as shown in Figure 27 (the reader is referred to Reference [378] for a more complete discussion regarding the derivations.) Since different forces typically act along each edge, which may have different number of fasteners, the strength




based on the fasteners along either of the two sides must be checked.

Considering fastener "A" in Figure 28, the strength is given by

$$V = 1 / \{ [b/(b + \alpha l_t')/(n_b Q_{ut})]^2 + [b/(\beta b + l_p')/(n_b Q_{up}')]^2 \}^{0.5} (4-8)$$

where

b = depth of diaphragm, in. $\alpha = q_t'/q_t = (0.87k_t/k_t' + 0.73)^{-1}$ (4-9) $\beta = q_p/q_p' = 0.81k_p/k_p' + 0.23$ (4-10) $q_t,q_t',q_p,q_p' = edge forces$ $l_t' = 2a'-2a'^2/a$ $l_p' = (b^2 + 4bb' - 4b'^2)/4a$ $n_b =$ number of fasteners along b. $Q_{ut} =$ strength of fastener in direction of q_t . $Q_{up}' =$ strength of fastener in direction of q_p' .

For diaphragms welded according to the guidelines of References [525, 526, 527], the values of α and β are recommended as follows:

$$\alpha = 0.64$$
 $\beta = 1.32$

when headed shear studs are used as fasteners, Equations (4-9) and (4-10) should be used to determine α and β . In Equations (4-9) and (4-10) the edge zone stiffness can be evaluated using the following Equation [130]:

 $k = 145.3 Q_{sl}/S_{s}$

(4 - 11)

where

k = edge zone stiffness, kips/in. Q_{su} = stud fastener capacity in load direction, kips S_c = stud spacing, in.

The values recommended for a' and b' are as follows:

welded diaphragms: $a' = b' = 5.83I_y^{0.35} - 2$ studded diaphragms: a' = b' = 15 inches.

where

I_y = moment of inertia about the weak axis of the framing member.

Considering fastener "B" in Figure 28, the strength is given by

$$V = 1/\{[a/(b + \beta l_p')/(n_a Q_{up})]^2 + [a'/(\alpha^{-1}b + l_t')/(n_a' Q_{ut}')]^2\}^{0.5}$$
(4-12)

where

If there are a sufficient number of fasteners designated "C" in Figure 27, to carry load along side a, after "B" connectors have failed the strength is given by:

$$V = (b + \beta^{-1}l_{p})(n_{a} - 2n_{a})Q_{up}/a$$
(4-13)

When headed shear studs are used as fasteners the strength were determined with the following expressions [353]:

$$Q_{sol} = 0.00666A_{st} f'_{c}^{0.3} E_{c}^{0.44}$$
(4-14)

where

 A_{st} = cross sectional area of stud, in² E_c = concrete modulus of elasticity, psi Q_{sol} = stud strength in solid slab, kips

Equation (4-14) is used for stud strength in a direction parallel with the ribs of the deck. For the strength in a direction perpendicular to the ribs of the deck the following equation is used [525,526]:

$$Q_{rib} = (0.85/\sqrt{N})[(H - h)/h](w/h)Q_{sol} \le Q_{sol}$$
(4-15)

where

 Q_{rib} = strength of stud in a transverse rib. N = number of studs in a rib. H = height of stud after welding. h = heigth of stud after welding. w = average rib width.

Where a stud is located near an edge in the direction of an applied shear force a reduction in calculated strength [321] is applied as given by:

$$Q'_{sol} = Q_{sol}[(d_{es} - 1)/(8d_{st})] \le Q_{sol}$$
(4-16)

where

 d_{es} = distance from the center of the stud to the free edge. d_{st} = diameter of stud.

When arc spot welds are used as fasteners, the strength, Q_u , is determined using the AISI [528] weld strength equations given by the smaller of:

$$Q_{u} = 0.625d_{e}^{2}F_{xx}$$
(4-17)

or one of the following

for
$$d_a/t \le 140/\sqrt{F_u}$$
 $Q_u = 2.2td_aF_u$ (4-18)
for $140/\sqrt{F_u} < d_a/t \le 240/\sqrt{F_u}$

$$Q_{u} = 0.28[1 + 960t/(d_{a}/F_{u})]td_{a}F_{u} \qquad (4-19)$$

for d_/t > 240//F_u = 1.4td_F_u (4-20)





where

d	= weld diameter, in.
da	= average diameter of arc spot weld at mid-thickness (d-t
	for single sheet, d-2t for double sheet), in.
d.	= effective diameter of fused area = $0.7d - 1.5t \le 0.55d$,
	in.
t	= thickness of sheet, in.
F_{u}	= ultimate strength of sheet, ksi
Fxx	= AWS weld designation strength, ksi

The predicted strength of a SDRC diaphragm controlled by a fastener failure is taken as

 $V = Minimum \{Eqn. 4-8, Maximum [Eqn. 4-12, Eqn. 4-13]\}$ (4-21)

4.3.1.3 Shear Transfer Mechanism

The strength of a SDRC diaphragm, when controlled by a shear transfer mechanism failure is determined using results from the elemental pushoff tests [404]. The strength is given by

V	=	Minimum{Eqn. 4-23, Eqn. 4-24}	(4-22)
v	=	$C_{po}Q_{tpo}(b + \alpha l_t')$	(4-23)
v	=	$C_{po}Q_{ppo}(b + \beta^{-1}l_{p}')$	(4-24)

where

 C_{po} = pushoff test correction factor = $(t_sd/s)n_s + t_c)/t_c$ Q_{tpo} = transverse pushoff test strength. Q_{ppo} = parallel pushoff test strength.

4.3.2 Initial Stiffness

For determining the initial stiffness, the diaphragm was modeled as a deep beam. The total stiffness was considered to consist of several components as is shown in the following equation

$$K_{tot} = 1/(1/K_{b} + 1/K_{s} + 1/K_{z} + 1/K_{f})$$
(4-25)

where

4.3.2.1 Bending Stiffness

The bending deflection term was based on the assumption that the composite slab act as the web in the deep beam model and the edge beams act as the flanges.

The bending stiffness is given by

$$K_{b} = 3(E_{c}I_{c} + E_{b}I_{b})/a^{3}$$

where

a = span at the diaphragm.

 E_{c} = concrete modulus of elasticity.

- I_c = moment of inertia of composite web.
- E_b = moment of inertia of edge beams about "deep beam" neutral axis.

4.3.2.2 Shear Stiffness

Shear deflection for steel deck diaphragms was given by Lutrell [306]. In the shear stiffness evaluation only the web of the deep beam model is assumed effective in resisting shear deformation. The shear stiffness is given by

$$K_{s} = b[G_{s}t_{s}(d/s) + G_{c}t_{c}]/a$$
 (4-27)

where

b = depth of diaphragm. G_s = shear modulus of elasticity. G_c = shear modulus of concrete.

4.3.2.3 Edge Zone Stiffness

The edge zone stiffness is given by the following equation:

$$K_{z} = \{2/[k_{t}(b + \alpha l_{t})] + 2a/[k_{p}b(b + \beta^{-1}l_{p})]\}^{-1}$$
(4-28)

where

 $l_t = a' - 2a'^2/3a$ $l_p = (b^2 + 3bb' - 2b'^2)/6a$

for welded connections using the results from elemental pushoff test $k_{\rm t}$ and $k_{\rm p}$ become

$$k_{t} = C_{po}Q_{tpo}$$
$$k_{p} = C_{po}Q_{ppo}$$

for studded fasteners use Equation (4-11).

(4 - 26)

.

4.3.2.4 Frame Connection Stiffness

The last component of the initial stiffness of the diaphragm be considered is the stiffness of the test frame support connection at the abutment. This rigid body rotation was originally deemed negligible but subsequent monitoring of test specimens revealed that indeed there was a small, yet measurable deflection. Measurements indicated that for the test frame constructed with the W24X76 sections the stiffness k_r was approximately 10,000 kips/in. and for the test frame constructed with the W14X22 sections the stiffness was approximately 4250 kips/in.

4.4 Precast Prestressed Diaphragms

The following discussion of initial stiffness and peak strength prediction was taken from Reference [376]. The strength and stiffness predictions for the hollow-core plank systems were based in the assumption that the connections between the planks and the testing frame would not fail before the diaphragm reached its limit state. Thus, by determination of the distribution of forces in the edge zone, the forces throughout the diaphragm were studied.

4.4.1 Initial Stiffness

The initial stiffness calculations were based on a predictive method initially developed by Porter and Greimann [390], later modified by Dodd [130], and Porter and Easterling [378]. The total deflection was assumed to consist of four components: Bending deflection, shear deflection, edge zone deflection, and deflection due to axial flexibility of edge beam framing connection. The initial stiffness is given by

$$K_{tot} = 1/(1/K_{b} + 1/K_{a} + 1/K_{z} + 1/K_{f})$$
(4-29)

where

K_{tot}	=	total diaphragm stiffness, kips/in
K	=	bending stiffness component, kips/in
Ks	=	shear stiffness component, kips/in
K _z	=	edge zone stiffness component, kips/in
Kf	=	frame connection component, kips/in

4.4.1.1 Bending Stiffness

For bending, the diaphragm was considered to be a cantilever girder with the hollow-core planks acting as the web and the edge beam acting as the flanges. The bending stiffness is given by

$$K_{\rm b} = 3(E_{\rm c}I_{\rm c} + E_{\rm b}I_{\rm b})/a^{3}$$

where

K_b = bending stiffness of system, kips/in

- a = length of cantilever girder, in.
- E_{c} = concrete modulus of elasticity of the plank, ksi
- I_c = moment of inertia of web, in⁴
- E_{b} = modulus of elasticity of edge beam, ksi
- I_b = moment of inertia of edge beams about girder neutral axis, in⁴

4.4.1.2 Shear Stiffness

For shear stiffness evaluation, only the web is assumed effective against shear and is given by

$$K_s = bG_c t_c / a$$

(4-31)

where

- K_s = shear stiffness of system, kips/in
- a = length of cantilever girder, in.
- b = depth of cantilever girder, in.
- G_c = shear modulus of concrete, ksi
- t_c = average thickness of concrete, in.

4.4.1.3 Edge Zone Stiffness

The deflection of the system due to edge zone deformation was based on a idealized edge zone force distribution. The edge zone stiffness is given by

$$K_{z} = 1/\{(r_{3} + r_{4})/[k_{t}(b + l_{t}^{"})] + 2ar_{1}/[k_{b}b(b + l_{b}^{"})]\}$$
(4-32)

where

K₂ = edge zone stiffness, kips/in

a = length of cantilever girder, in.

b = depth of cantilever girder, in.

 k_t , k_p = equivalent edge zone spring stiffnesses determined by using

$$k_{eq} = 145.3Q_{su}/S_s$$
 (4-33)

 k_{eq} = equivalent stiffness k_t or k_p , kips/in

- S_s = stud spacing, in.
- Q_{su} = stud connector capacity in the load direction, kips

$$Q_{su} = 6.66 \times 10^{-3} A_s f'_c^{0.3} E_c^{0.44} \le 0.9 A_s f_s$$
 (4-34)

(4-30)

A _s E _c f _s f' _c		area of stud, in ² modulus of elasticity of concrete, psi yield strength of the stud, ksi compressive strength of the concrete, psi	
r ₁	=	$g_{p}acoth(g_{p}a)$	(4-35)
r ₂	=	$g_{p}acsch(g_{p}a)$	(4-36)
r ₃	=	$g_t(b/2) \operatorname{coth}(g_tb/2)$	(4-37)
r ₄	=	$[b + r_3 a''(2-a''/3a)]/(b + a'')$	(4-38)
l _t "	=	a"[r ₃ (3a - a") - a"r ₄]/3a	(4-39)
l _p "	=	$(r_1b^2 + 3r_2bb'' - 2r_2b''^2)/6a$	(4-40)
a ^b	=	$\sqrt{(k_{p}/E_{b}A_{b})}$	(4-41)
g,	=	$\sqrt{(k_t/E_bA_b)}$	(4-42)
A	=	area of edge beam, in ²	

 $E_b = modulus of elasticity of edge beam, ksi$ a" = b" = initial effective edge zone width = 6 inches

4.4.1.4 Frame Connection Stiffness

The final component of the initial diaphragm deflection was the axial flexibility of the edge beam abutment connections. This frame connection stiffness component served as a correction due to framing and connection movements. The frame connection stiffness has been experimentally determined to be approximately 10,000 kips/in.

4.4.2 Peak Strength

The FME load capacity of the plank diaphragm systems was limited by one of three major categorical failure modes: shear-bond seam failure, tensile-bond seam failure, or diagonal tension failure. Limit states were identified as one of three potential events: initial weld tie shear failure, initial weld tie tension failure, or initial diagonal tension crack. The limiting predictive capacity of the diaphragm will be taken as follows: a) Peak strength in planks transverse to load direction:

 $V = Minimum \{ Maximum [Eqn.(4-52), Eqn.(4-54)], Eqn.(4-70) \}$

b) Peak strength in planks parallel to load direction:

 $V = Minimum \{ Maximum [Eqn.(4-61), Eqn.(4-66)], Eqn.(4-70) \}$

In order to predict the FME and limit state strength the following equations are needed

$$Q_p = (q_{p2}/g_p) \{ \sinh[g_p(42-a)] - \sinh(-g_pa) \}$$
 (4-51)

4.4.2.1 FME and Limit State prediction for Shear-Bond Failure (planks transverse to load direction)

FME prediction for shear-bond failure (planks transverse to load direction) is given by

$$V_{fme}^{p} = [T_{av}'d_{p}l_{s} + \gamma d_{t}l_{s}](b + l_{p}'')/(ar_{5})$$
(4-52)

where

 $T_{av}' = 69 \text{ psi} = \text{reduced shear stress}$ $d_p = \text{Grout depth of penetration, in}$ $l_s = \text{bond length, usually equal to a, in}$ $d_t = \text{depth of topping, in}$ $f'_{ct} = \text{topping concrete compressive strength, psi}$ $\gamma = 0.2f'_c \le 800 \text{ psi}$

$$r_{5} = 1 + g_{p} \operatorname{sech}(g_{p} a) b'' / [2 \tanh(g_{p} a)] + g_{p} (b^{2} / 4 - 48^{2}) / [b \tanh(g_{p} a)].$$
(4-53)

Limit state prediction for shear-bond failure (planks transverse to load direction) is given by

$$V_{ls}^{p} = [5.5n + 0.9(N_{c} + N_{t}) + \gamma d_{t}l_{s}](b + l_{p}')/(a + b/2 - 42)$$
(4-54)

where

$$l_{p}' = (b^{2} + 4bb' - 4b'^{2})/4a$$
(4-55)
n = number of weld ties.

$$N_{t} = [r_{7}/(a - l_{c}/3) - r_{6}][1 - l_{t}/(3a - l_{c})]$$
(4-56)

$$N_{c} = (N_{t}l_{t}/3 + r_{7})/(a - l_{c}/3)$$
(4-57)

$$r_6 = q_{tb}(b/2 - 48) + (q_{tb} - q_{tf1})a''/2 - Q_{tf}$$
 (4-58)

$$rr_{7} = a(b/2 - 48)(q_{tb} - q_{pav}) - (q_{tb} + q_{tf1})a''^{2}/6 + q_{tb}aa''/2$$
(4-59)

$$r_{7} = rr_{7} - (3b/2 - b'' - 144)q_{p2}b''/6 - q_{p1}(b/2 - 48)^{2}(16/b + 1/3)$$

$$l_{c} = l_{t} = a/2$$
(4-60)

4.4.2.2 FME and Limit State Prediction for Tensile-Bond Failure (planks parallel to load direction)

FME prediction for tensile-bond failure (planks parallel to load direction) is given by

$$V_{fme}^{p} = Minimum\{ V_{1} Eqn.(4-62), V_{2} Eqn.(4-63) \}$$
 (4-61)

$$V_{1} = v'_{tav}d_{p}l_{t}(3b - l_{t} - l_{c})/(-3r_{g}) + 5\sqrt{(f'_{ct})}d_{t}l_{s} \qquad (4-62)$$

$$V_{2} = (2/3) [T'_{av}d_{p}l_{s} + \gamma d_{t}l_{s}] (b + l_{t}") / [r_{4}(b + a")]$$
(4-63)

where

 $v'_{tav} = 40 \text{ psi} = \text{reduced tensile stress}$

$$rr_8 = bg_p/6 + sech(g_pa) \{sinh[g_p(42 - a)] + sinh(g_pa)\} (4-64)$$

$$r_{8} = r_{4}(42a'' - a''^{2}/3 + 42b)/(b + l_{t}'') - ab/[(b + l_{p}'')tanh(g_{p}a)])rr_{8}$$
(4-65)

Limit state prediction for tensile-bond failure (planks parallel to load direction) is given by

$$V_{ls}^{p} = Minimum\{ V_{3} Eqn.(4-67), V_{4} Eqn.(4-68) \}$$
 (4-66)

$$V_{3} = [5.5n + 0.9(N_{c} + N_{t}) + \gamma d_{t}l_{s}](b + l_{t}')/(b + 2a') \quad (4-67)$$

$$V_4 = (b - \frac{1}{3} - \frac{1}{4}) [N_c + 5\sqrt{(f'_{ct})} d_t l_s] [(b^2/4 + 42b)/(b + l_p') - (84a' - a'^2 + 42b)/(b + l_t')] (4-68)$$

where

$$l_{+}' = 2a^{*} - 2a^{*}/2 \qquad (4-69)$$

4.4.2.3 FME and Limit State Prediction for Diagonal Tension Failure

The diagonal tension failure represented an upper limit for a concrete diaphragm. Diagonal tension failure calculations are based on Equation (11-32) from the American Concrete Institute 318-83 code [524],

$$V_{c} = 3.3\sqrt{(f'_{c})}bd + N_{cp}d/(4l_{u})$$

(4 - 70)

where

 V_c = diagonal shear capacity of the concrete. f_c^{\dagger} = plank concrete compressive strength. b = diaphragm width. d = effective plank depth. N_{cp} = normal compressive force (prestressing) l_w = 0.8b

4.5 Plywood Diaphragms

In wood structures, the roof and floor sheathing are designed for gravity loads, but they can also carry lateral load. The use of roof and floors as diaphragms in structures basically require no modification to the basic construction. Common types of wood diaphragms are plywood diaphragms, heavy timber decking, diagonal sheathing, double diagonal sheathing, and transverse sheathing. In new construction the most common of the various sheathing materials is plywood because of a combination of factors including low installation cost, excellent rigidity and strength.

Resistance of the sheathing to shear caused by lateral loads depends on four things: sheathing thickness and layout, nailing type and spacing, provisions for blocking, and width of framing members. Deflection of plywood diaphragms can be evaluated considering the usual bending and shear components as well as other factors, such as joint slip in the chords, and nail deformation.

4.5.1 Diaphragm Strength

A plywood diaphragm may fail by nail heads pulling through the panel face, by nails pulling out through panel edges, by nails causing framing members to split, or by buckling of the plywood. Generally, shear-through-the-thickness failure is not a factor.

Table No. 25-J-1 [531], shows allowable shears in horizontal plywood diaphragms as a function of the sheathing thickness and layout, nailing type and spacing, provisions for blocking, and width of framing members. The allowable shears from Table No. 25-J-1 are based on extensive tests results - see Appendix A from Reference 477 - and using a factor of safety against ultimate strength of about 3 to 4. See also Table No. 25-K-1 [531] for allowable shears in plywood shearwalls.

For the cases not covered by Table No. 25-J-1, the allowable shear in blocked diaphragms is calculated as the smaller of (1) the allowable shear, $V_{\rm cp}$, based in plywood shear stress and (2) the allowable shear, $V_{\rm np}$, based on lateral fastener load at the boundary. $V_{\rm cp}$ and $V_{\rm np}$ are evaluated as follows [477]:

$$V_{\rm m} = \alpha \beta \gamma V \qquad (4-71)$$

where

- α = a factor of 1.33 for load duration (assuming wind or seismic load)
- β = a factor of 12 (in/ft)
- γ = the effective thickness for shear in the plywood (Table 1 of Reference 529)
- V = allowable stress for shear through the thickness of the plywood (Table 3 of Reference 529)

(4 - 72)

where

- δ = a factor of 1.30 for diaphragm construction by NDS Section 8.8.5.5 [530]
- ϵ = a factor of 1.33 for load duration
- ζ = number of nails per foot, except use 2.25 nails per foot (experimentally derived) when there are actually 2 nails per foot.
- η = a factor of 0.89 if 2-in. nominal lumber is used, or if two rows of fasteners are used in 3-in. nominal lumber, or if three rows are used in 4-in. nominal lumber.
- θ = a factor of 0.90 if the diaphragm consists of Douglas fir framing members and non-structural-I plywood.
- 1 = a factor of 0.85 if nails are spaced 2 in. c/c at boundary.
- κ = for 10d nails only, an additional factor of 0.85 if nails are spaced 2 in. or 2.5 in. c/c at boundary when boundary members are single 2-in. nominal members.
- λ = for non-Douglas fir lumber, a factor of 0.82 for lumber group III and a factor of 0.65 for lumber group IV.
- V_n = lateral load design value for the particular nail [530]

4.5.2 Diaphragm Stiffness

The deflection of a uniformly loaded rectangular blocked plywood diaphragm can be evaluated with the following formula [477]:

$$\Delta = 5vL^{3}/(8EAd) + vL/(4Gt) + 0.188Le_{n} + (\Sigma\Delta_{c}x_{s})/2d \qquad (4-73)$$

where

- Δ = the calculated deflection at the centerline, in.
- L = diaphragm length, ft.
- d = diaphragm depth, ft.
- E = elastic modulus of the flange material, psi
- A = area of flange cross section, in^2 .
- G = modulus of rigidity of plywood, psi
- t = effective thickness of plywood for shear, in.
- e_n = nail deformation for a given load on a nail (see Table B-4 in Reference 477), in.
- x_s = distance to the splice from the support, ft.
- Δ_{c}^{*} = individual chord splice slip, in.

The deflection of a blocked plywood shearwall may be calculated by use of the following formula:

$$\Delta = 8vh^{3}/(EAb) + vh/(Gt) + 0.376he_{n} + d_{a}$$
(4-74)

where

- Δ = the calculated deflection at the top of the wall, in.
- v = maximum shear due to design loads at the top of the wall, lb/ft.
- A = area of boundary elements cross section (vertical member at shearwall boundary), in².
- h = wall heigth, ft.
- b = wall width, ft.
- d_a = deflection due to anchorage details, in.
- E = elastic modulus of boundary element (vertical member at shearwall boundary), psi

The initial stiffness, K, can be derived from Equation (4-73) or Equation (4-74) by differentiating Δ as follows:

$$K = [(d\Delta/dV)^{-1}]_{V=0}$$

(4 - 75)

where

K = initial stiffness, lb/in.

 Δ = diaphragm or shearwall deflection, in.

V = maximum shear due to design loads, lb.

The research undertaken for this project is part of the U.S.-Japan Coordinated Program for Masonry Building Research. Each category of this program is conducted under the supervision of the Technical Coordinating Committee for Masonry Research (TCCMAR). The Study of floor diaphragms, which is the fifth research task was divided into separate tasks. The objective of this project (Task 5.2) was the collection of existing literature and data generated from the discussion and testing of horizontal diaphragms.

As part of this study previous diaphragm research was reviewed. The review included Cold-Formed Steel, Composite Steel Deck, Wood, Reinforced Concrete, and Precast Concrete diaphragms. Data generated from diaphragm testing is also included for the following diaphragms types: Planks, Steel-Deck-Reinforced diaphragms, Wood Diaphragms, and Steel Deck on wood framed diaphragms.

TCCMAR has requested that the hysteretic model named EKEH model which is part of the Lumped Parameter Model, be considered as part of the overall project. In order to assemble existing experimental data of floor diaphragms and reducing to a form required for static and dynamic analysis (e.g. EKEH model), the following data is needed: maximum deformation, deformation at peak strength, peak strength, initial stiffness, and post peak strength lower limit factor. Only some of this parameters were found on the experimental data generated from previous work. Additional work is needed to provide the complete set of data required for EKEH model, principally on Wood Diaphragms.

Additional work on the development of a hysteretic model for hollow-core planks is being conducted at Iowa State University as part of TCCMAR research program (Task 2.4a) [348].

6. ACKNOWLEDGEMENTS

This research investigation was sponsored by the National Science Foundation (Grant No. CES - 8517028, Formerly Grant No. CEE - 8517028). The work was conducted at Iowa State University through the auspices of the Engineering Research Institute.

The authors wish to thank the contributions made by the following graduate students: Ron J. Meyer, Paul M. Tremel, and Aziz A. Sabri.

The authors are greatly appreciative of the work done by Terri Smith and Karen Gilbertson who aided in the data collection for the research.

7. REFERENCES

- Abdel-Sayed, George, "Critical Shear Loading of Curved Panels of Corrugated Sheets." <u>Journal of the Engineering</u> <u>Mechanics Division</u>, ASCE, Vol. 96, No. EM6, December 1970, pp. 895-912.
- ABK, A Joint Venture, "Methodology for Mitigation of Seismic Hazards in Unreinforced Masonry Buildings: Diaphragm Testing." ABK-TR-03. ABK, A Joint Venture, El Segundo, California, December, 1981.
- 3. ABK, A Joint Venture, "Methodology for Mitigation of Seismic Hazards in Existing Reinforced Masonry Buildings: Interpretration of Diaphragm Tests." Topical Report 05, ABK-TR-05. ABK, A Joint Venture, South Pasadena, California.
- Abrams, D. P., "Measured Hysteresis in Masonry Building." Third U.S. National Conference on Earthquake Engineering, Vol. II, 1986, pp. 1371-1382.
- 5. Aktan, Ahmet E. and Bertero, Vitelmo V., "Seismic Response of R/C Frame-Wall Structures." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol 110:8, August 1984, pp. 1803-1821.
- 6. Aktan, A. E. and Bertero, V. V., "The Seismic Resistant Design of R/C Coupled Structural Walls." Report No. UCB/EERC-81/07, Earthquake Engineering Research Center, University of California, Berkeley, CA, June 1981.
- 7. Aktan, A. E., Bertero, V. V. and Piazza, M., "Prediction of the Seismic Response of R/C Frame-Coupled Wall Structures." Report No. UCB/EERC-82/12, Earthquake Engineering Research Center, University of California, Berkeley, CA, Aug., 1982.
- 8. Aktan, Haluk M. and Hanson, Robert D., "Nonlinear Cyclic Analysis of Reinforced Concrete Plane Members." Reinforced Concrete Structures Subjected to Wind and Earthquake Forces, <u>ACI Publication SP-63</u>, 1980, pp. 135-152.
- 9. Al-Yousef, Abdulrahman Fahd, "Composite Shear Connections Subjected to Reverse Cyclic Loading with and Without Ribbed Metal Deck." M. S. Thesis, University of Washington, Seattle, Washington, 1975.
- 10. Ambrose, J. and Vergun, D., <u>Simplified Building Design for</u> <u>Wind and Earthquake Forces</u>. 1980.
- 11. Ambrose, J. and Vergun, D., <u>Seismic Design of Buildings</u>. 1985.

- 12. American Concrete Institute, <u>Reinforced Concrete Structures</u> <u>Subjected to Wind and Earthquake Forces</u>. ACI SP-63, 1980.
- 13. American Concrete Institute, <u>Dynamic Modeling of Concrete</u> <u>Structures</u>. ACI Publication SP-73, 1982.
- 14. American Iron and Steel Institute, <u>Design of Light Gage</u> <u>Steel Diaphragms</u>, New York, American Iron Steel Institute, 1967.
- 15. Amrhein, J. E., "Design, Construction and Performance of Reinforced Masonry Buildings Subjected to Seismic Forces." <u>Proceedings of the Seventh World Conference on Earthquake</u> <u>Engineering.</u> Istanbul, Turkey, Vol. 4, Sept. 1980, pp. 161-168.
- 16. Anand, S. C. and M. A. Rahman, "Stresses in Composite Masonry Shear Walls Due to Earthquake Loads." Clemson University, sponsored by National Science Foundation, pp. 1359-1370.
- 17. Anderson, G. A., "Evaluation of Light-Gage Metal Diaphragm Behavior and the Diaphragm Interaction with Post Frames." Ph.D. Dissertation, Iowa State University, Ames, Iowa, 1987.
- 18. Anderson, J. C. and Townsend, W. H., "Models for RC Frames with Degrading Stiffness." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 103, No. ST12, December 1977, pp. 2361-2376.
- 19. Anderson, L. O., "Guides to Improved Framed Walls for Houses." U. S. Forest Service Research Paper FPL 31, Forest Products Laboratory, Madison, Wisconsin, 1965.
- 20. Ang, A., "The Development of a Distribution Procedure for the Analysis of Continuous Rectangular Plates." University of Illinois, Civil Engineering Studies, Structural Research Series No. 176, May 1959.
- 21. Applied Technology Council, "Tentative Provisions for the Development of Seismic Regulation for Buildings." National Bureau of Standards, June 1978.
- 22. Applied Technology Council, <u>Proceedings of a Workshop on</u> <u>Design of Horizontal Wood Diaphragm</u>, Nov. 19-21, 1980.
- 23. Applied Technology Council, <u>Proceedings of a Workshop on</u> <u>Design of Prefabricated Concrete Buildings for Earthquake</u> <u>Loads</u>, April 27-29, 1981.

- 24. Apparao, T. V. S. R., Errera, Samuel J. and Fisher, Gordon P., "Columns Braced by Girts and a Diaphragm." <u>Journal of</u> <u>the Structural Division</u>, ASCE, Vol. 95, ST5, May 1969, pp. 965-990.
- 25. Apparao, T. V. S. R., and Winter G., "Tests on Light Gage Steel Diaphragms". Department of Structutral Engineering, Cornell University, Report No. 328. Dec.. 1966.
- 26. Aristizabal-Ochoa, J. D., Oesterle, R. G., Fiorato, A. E. and Corley, W. G., "Cyclic Inelastic Behavior of Structural Walls." <u>Portland Cement Association, submitted for</u> <u>publication in Proceedings of Sixth European Conference on</u> <u>Earthquake Engineering</u>, 1978.
- 27. Aristizabal-Ochoa, J. and Sozen, M. A., "Behavior of Ten-Story Reinforced Concrete Walls Subjected to Earthquake Motions." University of Illinois Civil Engineering Studies, Structural Research Series No. 431, October 1976.
- 28. Arnold, C., "Building Configuration: Characteristics for Seismic Design." <u>Proceedings of the Seventh world Conference</u> <u>on Earthquake Engineering.</u> Istanbul, Turkey, Vol. 4, Sept. 1980, pp. 589-582.
- 29. Arnold, V. E., Greimann, L. F., and Porter, M. L., "Pilot Tests of Composite Floor Diaphragms." ISU-ERI-Ames-80133, Iowa State University, Ames, Iowa, Final Report submitted to National Science Foundation, May 1980, 174pp.
- 30. Arnold, V. E., Greimann, L. F., and Porter, M. L., "Pilot Tests of Composite Floor Diaphragms." ISU-ERI-Ames-79011, Iowa State University, Ames, Iowa, Progress Report Submitted to National Science Foundation, Sept. 1978, 179 pp.
- 31. Arnold, V. E., "Pilot Tests of Composite Floor Diaphragms." Master's Thesis, Iowa State University, Ames, Iowa, 1979, 179pp.
- 32. ASTM Standard E455-76, "Static Load Testing of Framed Floor or Roof Diaphragm Constructions for Buildings." ASTM Standards.
- 33. ASTM, <u>Standar Method for Static Load Testing of Framed Floor</u> <u>or Roof Diaphragm Constructions for Buildings</u>. ANSI/ASTM E 455. Philadelphia: ASTM, 1976.
- 34. Aswad, Alex and F. J. Jacques, "Detailed Design of Walls and Connections, Post-Tensioning Alternative." PCI Convention, October, 1983.

- 35. Atalay, B. and Penzien, J., "Inelastic Cyclic Behavior of Reinforced Concrete Flexural Members." Sixth World Conference on Earthquake Engineering, 1977, pp. 3062-3068.
- 36. Atherton, G. H. and Johnson, J. W., "Diagonally Sheathed Wood Diaphragms." Oregon Forest Products, Laboratory Report T-2, October 1951.
- 37. Atlas, A., Siess, Chester P. and Kesler, C. B., "Behavior of One-Way Concrete Floor Slabs Reinforced with Welded Wire Fabric." <u>Journal of the American Concrete Institute</u>, May 1965, pp. 539-557.
- 38. Atrek, E. and Nilson, A. H., "Non-Linear Finite Element Analysis of Light Gage Steel Shear Diaphragms." Report No. 363, Department of Structural Engineering, Cornell University, Ithaca, N. Y., September, 1976.
- 39. Atrek, E. and Nilson, A. H., (Istanbul Tech. Univ., Turkey), "Nonlinear Analysis of Cold-Formed Steel Shear Diaphragms." <u>Journal of the Structural Division</u>, ASCE, Vol. 106:ST3, March 1980, pp. 693-710.
- 40. Attewell/Taylor, <u>Ground Movements and Their Effects on</u> <u>Structures</u>. 1984.
- 41. Barda, F., Hanson, J. M. and Corley, W. G., "An Investigation of the Design and Repair of Low-Rise Shear Walls." (RD035.01D), Protland Cement Association, 1975. <u>Preprinted from Proceedings of the Fifth World Conference on EArthquake Engineering</u>, Rome, Italy, June 1973, Vol. 1, pp. 872-881.
- 42. Barda, F., Hanson, J. M. and Corley, W. G., "Shear Strength of Low-Rise Walls with Boundary Elements." (RD043.01D), Preprinted with permission from ACI Symposium, "Reinforced Concrete Structures in Seismic Zones." American Concrete Institute, by Portland Cement Association, 1976.
- 43. Barnes, C. R., "Design of One-Story Industrial Building for Lateral Loading."
- 44. Barnes, S. B. and Associates, "Report of Test Program on Mahon Steel Decks to Demonstrate Adequacy when used as Horizontal Diaphragms." S. B. Barnes and Associates, Los Angeles, California, April, 1957.
- 45. Barnes, S. B. and Associates, "Report on Use of H. H. Robertson Steel Roof and Floor Decks as Horizontal Diaphragms." S. B. Barnes and Associates, Los Angeles, California, July, 1963.

- 46. Barnes, S. B. and Associates, "Report on Concrete Filled Hi-Bond Steel Deck Diaphragms for the Inland Steel Products Company." S. B. Barnes and Associates, Los Angeles, California, June, 1966.
- 47. Bazant, Z. P. and Bhat, P. D., "Prediction of Hysteresis of Reinforced Concrete Members." <u>Journal of the Structural</u> <u>Division</u>, No. ST1, January 1977, pp. 153-167.
- 48. Beck, J. and Jayakumar, P., "Pseudo-Dynamic Testing and Model Identification." Third U.S. National Conference on Earthquake Engineering, 1986, pp. 1311-1322.
- 49. Becker, James M., Llorente, Carlos, and Mueller, Peter, "Seismic Response of Precast Concrete Walls." <u>Earthquake</u> <u>Engineering and Structural Dynamics</u>, Vol. 8, Revised April, 1980, pp. 545-564.
- 50. Berg, G. V. and Thomaides, S. S., "Energy Consumption by Structures in Strong-Motion Earthquakes." <u>Proceedings of the</u> <u>Second World Conference on Earthquake Engineering</u>, Tokyo and Kyoto, Japan, July 1960, Vol. II, pp. 681-696.
- 51. Bernander, K. G. and Liefendahl, T., "Shear Force Transmission Through Joints in Precast Hollow Floors Subjected to Bending in the Plane of the Floor." Nordisk, Betong, No. 6, 1976, pp. 23-26.
- 52. Bertero, Vitelmo V., "Identification of Research Needs for Improving a Seismic Design of Building Structures." Earthquake Engineering Research Center Report No. EERC 75-27, University of California Berkeley, 1975.
- 53. Bertero, Vitelmo V., "Seismic Behavior of R/C Wall Systems." <u>Proceedings of the 7th World Conference on Earthquake</u> <u>Engineering</u>, Istanbul, Turkey, Vol. 6, Sept., 1980.
- 54. Bertero, Vitelmo V., et al, "Pseudo-Dynamic Testing of Wall Structural Systems." California University, Berkeley, CA, Department of Civil Engineering, March 1976.
- 55. Bertero, V. V., Popov, E. P., Wang, T. Y. and Vallenas, J., "Seismic Design Implications of Hysteretic Behavior of Reinforced Concrete Structural Walls."
- 56. Bethlehem Steel Company, "Properties of Composite Sections for Bridge and Buildings." Bethlehem, PA.
- 57. Bethlehem Steel Company, "Properties of Composite Sections for Buildings." Bethlehem, PA.

- 58. Bethlehem Steel Company, "Properties of Composite Sections with Steel Deck." Bethlehem, PA, 1978.
- 59. Bjorhovde, Reidar and Zimmerman, T. J., "Some Aspect of Stub-Girder Design." <u>AISC Engineering Journal</u>, Third Quarter, 1980, pp. 54-69.
- 60. Blodgett, O. W., "Report on Proposed Standards for Sheet Steel Structural Welding." <u>Welding Journal</u>, April 1978.
- 61. Blume, J. A., "Dynamic Characteristics of Multistory Buildings." <u>Journal of the Structural Division</u>, ASCE, Vol. 94, No. ST2, February, 1968, pp 377-402.
- 62. Bockemohle, L. W., "Design of Topped Concrete Diaphragms on Precast Concrete Structures." <u>Proceedings for a workshop on</u> <u>the Seismic Design of Prefabricated Buildings</u>, Applied Technology Council, Berkeley, CA, 1981, pp. 503-516.
- 63. Boone, G. R. and Manbeck, H. B., "Using Existing Models to Predict Diaphragm Behavior." ASAE Winter Meeting, Paper No. 85-4501, 1985 Winter Meeting, Dec. 17-20, 1985.
- 64. Boppana, R. R. and F. Naeim, "Modeling of floor Diaphragms in Concrete Shearwall Buildings." <u>Concrete international</u>, July 1985, pp. 44-46.
- 65. Bower, Warren H., "Lateral Analysis of Plywood Diaphragms." Journal of the Structural Division, April 1974, pp.759-773.
- 66. Bolluyt, J. E., "Seismic Resistance of Composite Floor Diaphragms." Master's Thesis, Iowa State University, Ames, Iowa, May 1980, 174 pp.
- 67. Brangwin, D. J., "Interfacial Shear of Composite Floor Diaphragms." Master's Thesis, Iowa State University, Ames, Iowa, 1979, 113pp.
- 68. Brown, Colin B., "Factors Affecting the Damping in a Lap Joint." Journal of the Structural Division, ASCE, Vol. 94, No. ST5, May 1968, pp.1197-1217.
- 69. Brown, Elgar, Consulting Engineers, "Bearing test of Eight-Inch Thick Hollow-Core Span-Deck Slab." Permacrete Products Corp., Grove City, Ohio, October, 1976.
- 70. Brown, W. M. and Black, M. S., "Dynamic Strength Study of Small Fixed-Edge, Longitudinally Restrained Two-Way Reinforced Concrete Slabs." Technical Report No. N-73-8, U. S. Army Engineer Waterways Experiment Station, Vicksburg, Miss., 1973.

- 71. Bryan, Eric R. and El-Dakhakhni, M., "Shear Flexibility and Strength of Corrugated Decks." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 94, No. ST11, November 1968, pp. 2549-2619.
- 72. <u>Bulletin of the New Zealand National Society for Earthquake</u> Engineering, June 1980.
- 73. Burrows, C. H. and Johnson, J. W., "Lateral Test on Full-Scale Roof Diaphragm with Diamond-Pattern Sheathing." Oregon Forest Products Laboratory Report T-18, October 1956.
- 74. C. W. Blakeslee Company, "Load Tests on Reinforcing Bars Cast in Keyways and Cores of Span-Deck Units." Nov., 1971.
- 75. Canadian Sheet Steel Building Institute, "Diaphragm Action of Cellular Steel Floor and Roof Deck Construction." Ontario, Canada, December 1972.
- 76. Cardenas, A. E., Hanson, J. M., Corley, W. G. and Hognestad, E., "Design Provisions for Shear Walls." <u>Journal of the</u> <u>American Concrete Institute</u>, Vol. 70, No. 3, March 1973, pp. 221-230.
- 77. Cardenas, A. E. and Magura, D. D., "Strength of High-Rise Shear Walls-Rectangular Cross Section." (RD029.01D), Portland Cement Association, 1975. Reprinted from <u>Response</u> of <u>Multistory Concrete Structures to Lateral Forces</u>, SP-36, American Concrete Institute, Detroit, Michigan, 1973, pp. 119-150.
- 78. Carpenter, J. E., Koar, P. H. and Corley, W. G., "Design of Ductile Flat-Plate Structures to Resist Earthquakes." (RD035.01D), Portland Cement Association, 1975. Reprinted from <u>Proceedings of Fifth World Conference on Earthquake</u> <u>Engineering</u>, Rome, Italy, June 1973, Vol. 2, pp. 2016-2019.
- 79. Carskaddan, Phillip S., "Shear Buckling of Unstiffened Hybrid Beams." Journal of the Structural Division, ASCE, Vol. 94, No. ST8, August 1968, pp. 1965-1990.
- 80. Casillas, Juan G. de Leon and Siess, C. P., "Comparative Studies of Design Procedures for Two-Way Reinforced Concrete Slabs." University of Illinois Civil Engineering Studies, Structural Research Series No. 215, May 1961.
- 81. Cauvin, A. and Zanon, P., "Test Results on Key Joints of Large Panel Prefabricated Buildings Subject to Cyclic Reversed Actions." Eighth World Conference on Earthquake Engineering, Vol. VI, 1984, pp. 765-771.

- 82. Celebi, M., Brady, G., Safak, E., and Converse, A., "Performance of an Earthquake Excited Roof Diaphragm." <u>Dynamic Response of Structures</u>, 1986, pp. 863-870.
- 83. Chajes, Alexander, "Inelastic Deflection of Beams." <u>Journal</u> of the Structural Division, ASCE, Vol. 94, No. ST6, June 1978, pp. 1549-1565.
- 84. Chockalingam, S., Ha, K. and Fazio, P., "Simplified Analysis of Cold-Formed Steel Shear Diaphragms." <u>Canadian Journal of</u> <u>Civil Engineering</u>, Vol. 6, 1979, pp. 232-275.
- 85. Chopra, A. K., <u>Dynamic of Structures</u>. Earthquake Engineering Research Institute, 1980.
- 86. Chopra, A. K. and Gutierrez, J. A., "Earthquake Analysis of Multistory Buildings Including Foundation Interaction." Earthquake Engineering Research Center Report No. EERC 73-13, University of California Berkeley, 1973.
- 87. Chopra, A. K. and Kan, C., "Effects of STiffness Degradation on Ductility Requirements for Multistory Buildings." <u>Earthquake Engineering and Structural Dynamics</u>, John Wiley and Sons, Chichester, Sussex, England, Vol. 2, No. 1, July 1973, pp. 35-45.
- 88. Chow, L., Conway, H. D. and Winter, G., "Stresses in Deep Beams." <u>Transactions of the American Society of Civil</u> <u>Engineers</u>, 118, 1953, p. 686.
- 89. Clough, D. P., "Considerations in the Design and Construction of Precast Concrete Diaphragms for Earthquake loads." Journal of the Prestressed Concrete Institute, V 27:2 March/April, 1982, pp. 78-93.
- 90. Clough, D. P., "Considerations in the Design of Precast Concrete Diaphragms for Earthquake Loads." <u>Proceedings of a</u> <u>Workshop on the Design of Prefabricated Concrete Buildings</u> <u>for Earthquake Loads</u>, Applied Technology Council, Berkeley, CA, Dec., 1981, pp. 617-638.
- 91. Clough, D. P., "Development of Seismic Design Criteria for Connections in Jointed Precast Concrete Structures." <u>Eighth</u> <u>World Conference on Earthquake Engineering</u>, 1984, pp. 645-652.
- 92. Clough, R. W., and Johnston, S. B. "Effect of Stiffness Degradation on Earthquake Ductility Requirements." <u>Proc. at</u> <u>Second Japan Earthquake Engineering Symposium</u>. Tokyo: 1966. 227-232.

- 93. Concrete Technology Associates, "Shear Diaphragm Capacity of Precast Floor System." Technical Bulletin 73-B6, Tacoma, Washington, June 1973.
- 94. Concrete Technology Associates, "Shear Diaphragm Capacity of Untopped Hollow-Core Floor Systems." Tacoma, Washington, August, 1981.
- 95. Conley, C. H., White, R. N. and Gergely, P., "Strength and Stiffness of Reinforced Concrete Panels Subjected to Membrane Shear, Two-Way and Four-Way Reinforcing." (NUREG/CR-2049), U. S. Nuclear Regulatory Commission, Washington, DC, 1981.
- 96. Cook, John P., "Adhesive Bonded Composite Beams." ASCE Convention and Exposition, October 23-25, 1979, Atlanta, Georgia, Preprint 3645.
- 97. Corley, W. G., Sozen, M. A. and Siess, C. P., "The Equivalent Frame Analysis for Reinforced Concrete Slabs." University of Illinois Civil Engineering Studies, Structural Research Series No. 218, June 1961.
- 98. Cosper, S. J., Anderson, A. R. and Jobse, H. J., "Shear Diaphragm Capacity of Untopped Hollow-Core Floor Systems." Concrete Technology Associates, Tacoma, Washington, Tech. Bulletin 80-B3, August, 1981.
- 99. Cotran, F. S. and Hall, W. J., "Effective Width of Floor Systems for Application in Seismic Analysis." Civil Engineering Studies, University of Illinois, Urbana, November 1980.
- 100. Coull, A. and Chee, W. Y., "Design of Floor Slabs Coupling Shear Walls." <u>Journal of Structural Engineering</u>, ASCE, Vol. 109, No. 1, January 1983, pp. 109-125.
- 101. Countryman, D., "Lateral Tests on Plywood Sheathed Diaphragms." American Plywood Association Laboratory Report 55, March 1952.
- 102. Countryman, D. and Colbenson, P., "1954 Horizontal Plywood Diaphragm Tests." American Plywood Association Laboratory Report 63a, January 1955.
- 103. Cox, H. L., "Behavior of Plate-Beam Connection Beyond the Elastic Range." University of Illinois Civil Engineering Studies, Structural Research Series No. 60, August 1953.
- 104. Criswell, M. E., "Strength and Behavior of Reinforced Concrete Slab-Column Connections Subjected to Static and Dynamic Loading." <u>Dissertation Abstracts</u>, Vol. 31: 12(b), p.413.

- 105. CTA, <u>Shear Diaphragm Capacity of Precast Floor Systems</u>. CTA Technical Bulletin, 1973.
- 106. Currier, R. A., "Lateral Tests on a Full-Scale Roof Diaphragm Sheathed with 2-by-3 inch Lumber." Oregon Forest Products Laboratory Report T-10, October 1954.
- 107. Dai, P. K. H. and Siess, C. P., "Analytical Study of Composite Beams with Inelastic Shear Connection." University of Illinois Civil Engineering Studies, Structural Research Series No. 267, June 1963.
- 108. Daniels, J. H. and Du Plessis, D. P., "Strength and Stiffness of Composite Beam-to-Column Connections." Part I of Bulletin No. 28, Plastic Analysis and Tests for Rigid High-Rise Composite Frames, Lehigh University, 1979.
- 109. Daniels, J. H. and Wenk, T., "Composite Assemblage Experiments." Part II of Bulletin No. 28, Plastic Analysis and Tests for Rigid High-Rise Composite Frames, Lehigh University, 1979.
- 110. Davidson, H. L., "Post-Buckling Behavior of Long Rectangular Plates." Fritz Engineering Laboratory Report No. 248.15, 1965.
- 111. Davies, J. M., "The Design of Shear Diaphragms of Corrugated Steel Sheeting." Department of Civil Engineering, The University of Salford, September 1971.
- 112. Davies, J. Michael, "Simplified Diaphragm Analysis." Journal of the Structural Division, ASCE, Vol. 103, No. ST11, November 1977, pp. 2093-2109.
- 113. Davies, J. M. and Bryan, E. R., <u>Manual of Stressed Skin</u> <u>Diaphragm Design</u>. John Wiley and Sons, Inc., NY, 1982.
- 114. Davies, J. M. and Fisher, J., "The Diaphragm Action of Composite Slabs." <u>Inst. Civil Engineers</u>, Proc's Part 2, Research & Theory, V 67, Dec. 1979, pp. 891-906.
- 115. Davies, J. M. and Lawson, R. M., "Light Gage Steel Diaphragms with Openings." <u>IABSE Proceedings</u>, P-16/78, IABSE Periodica, August 1978, pp. 1-20
- 116. Dean, Donald L. and Payne, William W., "Analysis of Reinforced or Cellular Panels and Decks." ASCE Fall Convention and Exhibit, October 17-21, 1977, San Francisco, CA, Preprint 3013.

- 117. DeBuen, Oscar, "Plastic Design of Unbraced Multistory Steel Frames with Composite Beams." <u>AISC Engineering Journal</u>, Fourth Quarter, 1980, pp. 81-89.
- 118. Degenkolb, Henry J., "Basic Design Criteria of the Recommended Lateral Force Requirements and Commentary." Journal of the Structural Division, ASCE, Vol. 98, No. ST9, September 1972, pp. 1913-1922.
- 119. Degenkolb, Henry J., "Earthquake Forces on Tall Structures." Bethlehem Steel Company, Bethlehem, PA.
- 120. DeGroot, A. K., Kusters, G. M. A. and Monnies, T., "Numerical Modelling of Bond-Slip Behavior." Heron, Vol. 26, No. 1B, 1981.
- 121. DeLisle, D. J. and Heidebrecht, A. C., "Comparision of the Behavior of Load Bearing Shear Wall Panels Subjected to Repeated Cyclic Lateral Loading." First Canadian Conference on Earthquake Engineering, 1972, pp. 270-283.
- 122. Derecho, A. T., "Program Description-Analysis of Plane Multistory Frame-Shear Wall Structures Under Lateral and Gravity Loads." PCA, 1971.
- 123. Derecho, A. T., Ghosh, S. K., Iqbal, M. and Fintel, M., "Strength, STiffness, and Ductility Required in Reinforced Concrete Structural Walls for Earthquake Resistance." <u>ACI</u> Journal, August 1979, pp. 875-896.
- 124. Derecho, A. T., Ghosh, S. K., Iqbal, M., Freskakis, G. N. and Fintel, M., "Structural Walls in Earthquake-Resistant Buildings -- Analytical Investigation, Dynamic Analysis of Isolated Structural Walls -- Parametric Studies." Final Report to the National Science Foundation, RANN, under Grant No. ENV74-14766, March 1978.
- 125. Derecho, A. T., Iqbal, M., Ghosh, S. K., et al, "Structural Walls in Earthquake-Resistant Buildings, Dynamic Analysis of Isolated Structural Walls -- Representative Loading History." National Science Foundation Report No. NSF/RA-780255, Washington, DC, August 1978.
- 126. Derecho, A., Schults, D. and Fintel, M., "Analysis and Design of Small Reinforced Concrete Buildings for Earthquake Forces." Portland Cement Association, 1978, pp. 30-32.
- 127. Departments of the Army, the Navy and the Air Force, <u>Seismic</u> <u>Design of Buildings</u>. Army TM 5-809-10, Washington D.C.: U.S. Government Printing Office, 1973.

- 128. Departments of the Army, the Navy and the Air Force, <u>Seismic</u> <u>Design of Buildings</u>. Army TM 5-809-10, Washington D.C.: U.S. Government Printing Office, 1982.
- 129. Disque, Robert O., "Inelastic K-factor for Column Design." <u>AISC Engineering Journal</u>, Second Quarter, 1973, pp. 33-36.
- 130. Dodd, Steven M., "Effect of Edge Fasteners on the Seismic Response of Composite Floor Diaphragms." Master of Science Thesis, Iowa State University, Ames, IA, 1986.
- 131. Douglas, John H., "Earthquake Research (2): Averting Disaster." <u>Science News</u>, Vol. 115, February 10, 1979.
- 132. Dowrick, D. J., Earthquake Resistant Design. 1977.
- 133. Doyle, D. V., "Diaphragm Action of Full-Scale Diagonally Sheathed Roof or Floor Panels." Forest Product Laboratory Report TM-94, April 1955.
- 134. Doyle, D. V., "Diaphragm Action of Wood Roof or Floor Panels Diagonally Sheathed with 2-by-6 Lumber." Forest Product Laboratory Report P&E-123, April 1956.
- 135. Doyle, D. V., "Diaphragm Action of Diagonally Sheathed Wood Panels." Forest Product Laboratory Report 2082, November 1957.
- 136. Earthquake Engineering Research at Berkeley 1976, California University, Berkeley, Earthquake Engineering Research Center, May 1977.
- 137. Earthquake Engineering Research Institute, <u>Proceedings of</u> <u>the U. S. National Conference on Earthquake Engineering -</u> <u>1975</u>, Ann Arbor, Michigan.
- 138. Easley, John T., "Buckling Formulas for Corrugated Metal Shear Diaphragms." <u>Journal of the Structural Division</u>, ASCE, Vol. 101, No. ST7, July 1975, pp. 1403-1417.
- 139. Easley, John T., "Strength and Stiffness of Corrugated metal Shear Diaphragms." <u>Journal of the Structural Division</u>, ASCE, Vol. 103, No. ST1, January 1977, pp. 169-180.
- 140. Easley, John T., Foomani, M. and Dodds, R. H., "Formulas for Wood Shear Walls." <u>Journal of the Structural Division</u>, ASCE, Vol. 108, No. ST11, November 1982, pp. 2460-2478.

- 141. Easterling, W. S., Porter, M. L., "Hysteretic Behavior of Composite Diaphragms." <u>Dynamic Response of Structures</u>, edited by Gary Hart and Richard Nelson, Proceedings of Third Engineering Mechanics Conference, American Society of Civil Engineers, New York, New York, pp. 155-161, April 1, 1986.
- 142. Ebeling, Dick W., "Design Considerations for Diaphragms." ASCE Fall Convention and Exhibit, October 17-21, 1977, San Francisco, CA, Preprint 3002.
- 143. Edinger, John A., "Economics of Using LRFD." Paper presented at the ASCE Spring Convention in Atlanta, May 1984.
- 144. El-Dakhakhni, W. M. and Daniels, J. H., "Frame-Floor-Wall System Interaction in Buildings." Fritz Laboratory Report No. 376.2, Lehigh University, Bethlehem, PA, April 1973.
- 145. El-Ghazzi, M. N., Robinson, H. and Elkholy, I. A. S., "Longitudinal Shear Capacity of the Slabs of Composite Beams." <u>Canadian Journal of Civil Engineering</u>, Vol. 3, 1976, pp. 514-522.
- 146. El-Hakim, Noor El-Din Mohamed, "Diaphragm Action of Asbestos-Cement Decks." Ph.D. Dissertation, Concordia University, Montreal, Quebec, Canada, 1980.
- 147. El-Hakim, Noor El-Din Mohamed, "Diaphragm Action of Asbestos-Cement Decks." <u>Dissertation Abstracts</u> <u>International</u>, Vol. 42:2, August 1981, pp. 705-B.
- 148. Ellifritt, D. S. and Luttrell, L. D., "Strength and Stiffness of Steel Deck Shear Diaphragms." First Specialty Conference on Cold-Formed Steel Structures. August 19-20, 1971, pp. 99-110.
- 149. Elliot, J. R., "Wood Diaphragm Testing -Past, Present, Planned." <u>Proceedings of a Workshop on Design of Horizontal</u> <u>Wood Diaphragms</u>, Applied Technology Council, November 19-20, 1980, pp. 143-172.
- 150. Elsesser, E., "Analytical Modeling Problems of Precast Diaphragms." <u>Proceedings of a Workshop on the Design of</u> <u>Prefabricated Concrete Buildings for Earthquake Loads</u>, Applied Technology Council, Berkeley, CA, Dec. 1981, pp. 463-480.
- 151. Engineers Testing Laboratories, Inc., "Report on Horizontal Shear Tests of Span-Deck Grouted Edge Joints." March, 1973.

- 152. Englekirk, R. E. and Hart, G. C., "Earthquake Design of Concrete Masonry Walls." <u>Proceedings of the Seventh World</u> <u>Conference on Earthquake Engineering</u>, Istanbul, Turkey, Vol. 4, Sept. 1980, pp. 601-604.
- 153. Englekirk, R. E. and Hart, G. C., "Earthquake Design of Concrete Masonry Buildings: Response Spectra Analysis and General Earthquake Modeling Considerations." Prentice-Hall, Inc., 1982.
- 154. Ernst, G. C. and Smith, G. M., "Displacement Versus Force-Seeking Effects for Reinforced Concrete Frames."
- 155. Errera, Samuel J. and Apparao, Tamirisa V. S. R., "Design of I-Shaped Beams with Diaphragm Bracing." <u>Journal of the</u> <u>Structural Division</u>, ASCE, Vol. 102, No. ST4, April 1976, pp. 769-781.
- 156. Ersoy, U., Aktan, E. and Tankut, T., "Earthquake Resistant Prestressed Concrete Structures." <u>Proceedings of the Seventh</u> <u>World Conference on Earthquake Engineering</u>, Istanbul, Turkey, Vol. 4, Sept. 1980, pp. 296-302.
- 157. Esvelt and Saxton, "Report on Shear Strength of Span-Deck Edge Joint." Central Pre-Mix Concrete Co., Spokane, Washington, Feb., 1972.
- 158. Ewing, R. D., Healey, T. J. and Agbabian, M. S., "Seismic Analysis of Wood Diaphragms in Masonry Buildings." <u>Proceedings of a Workshop on Design of Horizontal Wood</u> <u>Diaphragms</u>, November 19-20, 1980, Applied Technology Council, pp. 253-276.
- 159. Ewing, R. D., Kariotis, J. C. and El-Mustapha A., "LPM/1 -A Computer Program for the Nonlinear Analysis of Lumped Parameter Models." National Science Foundation, Technical Report 2.3-1, August, 1987.
- 160. Fardis, Michael N. and Buyukozturk, Oral, "Shear Stiffness of Concrete by Finite Elements." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 106, No. ST6, June 1980, pp. 1311-1327.
- 161. Fardis, Michael N. and Buyukozturk, Oral, "Shear Transfer Model for Reinforced Concrete." <u>Journal of the Engineering</u> <u>Mechanics Division</u>, ASCE, Vol. 105, No. EM2, April 1979, pp. 255-275.
- 162. Farwell, Charles, R., Jr., "Torsion of Wide-Flange Members in the Strain-Hardening Range." Presented at the 1968 Annual Meeting and Technical Session of the Column Research Council, San Francisco, CA, May 1968.

- 163. Fazio, P., Ha, K. and Chockalingam, S., "Strength of Cold-Formed Steel Shear Diaphragms." <u>Canadian Journal of Civil</u> <u>Engineering</u>, Vol. 6, 1979, pp. 5-17.
- 164. Fertis, D. G., "Dynamic Hinge Concept for Beam Vibrations." <u>Journal of the Structural Division</u>, ASCE, Vol. 92, No. ST1, February 1966, pp. 365-380.
- 165. Filippou, F. C. and Popov, E. P., "Hysteretic Behavior of Reinforced Concrete Beams and Joints." National Science Foundation Grant, Paper presented by Faculty members of the University of California, Berkeley, pp. 89-95.
- 166. Fintel, Mark, "A Discussion on an Alternate Procedure for Earthquake-Resistant Design of Multistory Reinforced Concrete Structures Based on Inelastic Dynamic Analysis." 1978.
- 167. Fintel, Mark, "Ductile Shear Walls in Earthquake-Resistant Multistory Building." Portland Cement Association, Reprint from the <u>Journal of the American Concrete Institute</u>, Proceedings, Vol. 71, No. 6, June 1974.
- 168. Fintel, Mark, "Investigation of Earthquake Resistance of Structural (Shear) Wall Building Carried out at Portland Cement Association." <u>Wind and Seismic Effects, Proceedings</u> of the 8th USNR Panel Conference, U. S. Department of Commerce.
- 169. Fintel, M., et al, "Response of Buildings to Lateral Forces." <u>ACI Journal</u>, February 1971.
- 170. Fintel, M. and Ghosh, S. K., "Case Study of Seismic Resistance of a 16-Story Coupled Wall Structure Using Inelastic Dynamic Analysis and a Rational Approach." Portland Cement Association, October 1978.
- 171. Fiorato, A. E., Oesterle, R. G. and Corley, W. G., "Ductility of Structural Walls for Design of Earthquake Resistant Buildings." Sixth World Conference on Earthquake Engineering, 1977, pp. 2797-2802.
- 172. Fisher, John W., "Design of Composite Beams with Formed Metal Deck." AISC National Engineering Conference, May 1970, Pittsburgh, PA.
- 173. Fleischer, Walter H., "Simplified High Rise Drift Analysis and Optimized Adjustment." <u>Engineering Journal</u>, AISC, April 1972.
- 174. Fleischer, Walter H., "Simplified Seismic Drift Analysis of High-Rise Steel Frames." <u>AISC Engineering Journal</u>, Third Quarter, 1974, pp. 53-66.

- 175. Flurh, W. E., Ang, A. and Siess, C. P., "Theoretical Analysis of the Effects of Openings on the Bending Moments in Square Plates with Fixed Edges." University of Illinois Civil Engineering Studies, Structural Research Series No. 203, July 1960.
- 176. Foogiano, Paolo, "Experiences with Fabrication Detailing and Erection of Towers Using Cold-Formed Shapes." ASCE Spring Convention in Atlanta, May 1984.
- 177. Foomani, Medhi, "An Investigation of the Effect of Discrete Fasteners on the Strength and Stiffness of Plywood Shear Diaphragms." <u>Dissertation Abstracts International</u>, Vol. 42:7, January 1982, pp. 2926B-2927B.
- 178. Foschi, Ricardo O., "Analysis of Wood Diaphragms and Trusses, Part I: Diaphragms." <u>Canadian Journal of Civil</u> <u>Engineering</u>, Vol. 4:3, September 1977, pp. 345-362.
- 179. Foschi, Ricardo O. and Barrett, J. D., "Longitudinal Shear in Wood Beams a Design Method." <u>Canadian Journal of Civil</u> <u>Engineering</u>, Vol. 4, 1977.
- 180. Foutch, D. A., Housner, G. W. and Jennings, P. C., "Dynamic Responses of a Six Multistory Buildings During the San Fernando Earthquake." California Institute of Technology, Reprint No. EERL 75-02, Pasadena, CA, 1975.
- 181. Freeman, S. A., "Racking Tests of High-Rise Building Partitions." ASCE National Structural Engineering Convention, April 14-18, 1975, New Orleans, Louisiana, Preprint 2462.
- 182. Friedman, E., "Analysis of Weld Puddle Distortion and its Effect on Penetration." <u>Welding Research Supplement</u>, June 1978.
- 183. Fukumoto, Y. Usami, T. and Yamaguchi, K., "Inelastic Buckling Strength of Stiffened Plates in Compression." <u>International Association for Bridge and Structural</u> <u>Engineering Proceedings</u>, August 1977.
- 184. Gamble, W. L., Sozen, M. A. and Siess, C. P., "An Experimental Study of a Reinforced Concrete Two-Way Floor Slab." University of Illinois Civil Engineering Studies, Structural Research Series No. 211, June 1961.
- 185. Gamble, W. L., Sozen, M. A. and Siess, C. P., "Measured and Theoretical Bending Moments in Reinforced Concrete Floor Slabs." University of Illinois Civil Engineering Studies, Structural Research Series, No. 246, June 1962.

- 186. Gangarao, H. and Luttrell, L. D., "Preliminary Investigations Into the Response of Timber Diaphragms." <u>Proceedings of a Workshop on Design of Horizontal Wood</u> <u>Diaphragms</u>, Applied Technology Council, November 19-20, 1980.
- 187. Gaston, J. R., Siess, C. P. and Newmark, N. M., "An Investigation of the Load-Deformation Characteristics of Reinforced Concrete Beams up to the Point of Failure." University of Illinois Civil Engineering Studies, Structural Research Series No. 40, December 1952.
- 188. Gates, W. E., "Seismic Design Considerations for Untopped Precast Concrete Floor and Roof Diaphragms." <u>Proceedings of</u> <u>a Workshop on Design of Prefabricated Concrete Buildings for</u> <u>Earthquake Loads</u>, Applied Technology Council, Dec. 1980, pp. 517-549.
- 189. Gaus, M. P., "A Numerical Solution for the Transient Strain Distribution in a Rectangular Plate with a Propagating Crack." University of Illinois Engineering Studies, Structural Research Series No. 182, September 1959.
- 190. Gaylord, Edwin H., "Design Recommendations for Cold-Formed Sections." ASCE Spring Convention in Atlanta, May 1984.
- 191. Gebrail, J. A., "Lateral Load Distribution Test on Prestressed Span-Deck Slabs." Con-Force Products Ltd., Vancouver, B. C., 1976.
- 192. Gebrail, J. A., "Punching Shear for Untopped Span-Deck." Vancouver, B. C., 1977.
- 193. Gebremedhin, K. G. and Jorgensen M. C., "Design Height for Modern Post-Frame Buildings Using Diaphragm Theory." Paper No. 85-4502, ASAE Winter Meeting, Dec. 17-20, 1985.
- 194. Ghosh, S. K., et al, "Strength Design of Structural Walls with Particular Reference to Shear and Cyclic Loading." Portland Cement Association, Skokie, Illinois, Engineering Services Department, 1976.
- 195. Ghusn, G. and Saiidi, M., "A Hysteresis Model for Biaxial Bending of Reinforced Concrete Columns." Third U. S. National Conference on Earthquake Engineering, 1986, pp. 1027-1038.
- 196. Goel, Subhash and Berg, Glen V., "Inelastic Earthquake Response of Tall Steel Frames." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 94, No. ST8, August 1968, pp. 1907-1934.

- 197. Goldberg, John E., "Analysis of Multistory Buildings." ASCE Structural Engineering Conference and Annual Meeting, October 19-23, 1964, Preprint 105.
- 198. Goldberg, John E., "Analysis of Multistory Buildings Considering Shear Wall and Floor Deformations."
- 199. Goodman, James R. and Popov, Egor Paul, "Layered Beam Systems with Interlayer Slip." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 94, No. ST11, Novemeber 1968, pp. 2535-2547.
- 200. Grant, John A., Jr., Fisher, John W. and Slutter, Roger G., "Composite Beams with Formed Steel Deck." <u>AISC Engineering</u> <u>Journal</u>, First Quarter, 1977, pp. 24-43.
- 201. Grant, J. A., Jr., Fisher, J. W. and Slutter, R. G., "Shear Connector Behavior in Composite Beams with Metal Decking." ASCE, April 26-30, 1982, Las Vegas, Nevada, Preprint 82-034.
- 202. Green, Norman B., <u>Earthquake Resistant Building Design and</u> <u>Construction</u>, Second Ed., Elsevier, New York, 1981.
- 203. Green, Norman B., <u>Earthquake resistant Building Design and</u> <u>Construction</u>, Third Ed., Elsevier, New York, 1987.
- 204. Grinter, L. E., "Design of the Reinforced Concrete Road Slab." Bulletin of the Agricultural and Mechanical College of Texas, Vol. 1, No. 3, Bulletin No. 39, March 1, 1931.
- 205. Grinter, L. E. and Harding, F., "Tests of Precast-Slab Floor Panel." Technical Bulletin Armour Institute of Technology, Vol. 1, No. 1, Bulletin No. 1, August 1938.
- 206. Guzman, R. A. and Jenning, P. C., "Determination of Design Spectra for Nuclear Power Plants." ASCE National Structural Engineering Convention, April 14-18, 1975, New Orleans, Louisiana, Preprint 2508.
- 207. Ha, H. K., "Corrugated Shear Diaphragms." <u>Journal of the</u> <u>Structural Division</u>, ASCE, Vol. 105, No. ST3, March 1979, pp. 577-587.
- 208. Ha, H. K., El-Hakim, N. and Fazio, P. P., "Simplified Design of Corrugated Shear Diaphragms." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 105, No. ST7, July 1979, pp. 1365-1377.
- 209. Hallam, M. W., "The Behavior of the Stud Shear Connectors Under Repeated Loading." Research Report No. 281, School of Civil Engineering, The University of Sydney, August 1976.

- 210. Hansell, W. C., Galambos, T. V., Ravindra, M. and Viest, I. M., "Composite Beam Criteria in LRFD." <u>Journal of the</u> <u>Structural Division</u>, ASCE, Vol. 104, No. ST9, September 1978, pp. 1409-1426.
- 211. Hanson, N. W. and L. S. Johel, "Design for Vertical Loads on Horizontal Connections in Large Panel Structures." <u>Journal</u> <u>of Prestressed Concrete Institute</u>, Vol. 27, No. 1, Jan/Feb 1982, pp. 62-79.
- 212. Hanson, Robert D., Goel, Subhash C. and Gerg, Glen V., "Seismic Behavior of Staggered Truss Framing System Design Procedure for Earthquake Loading." Department of Civil Engineering, The University of Michigan, January 1972.
- 213. Harper, G. N. and Ang, A., "A Numerical Procedure for the Analysis of Contained Plastic Flow Problems." University of Illinois Civil Engineering Studies, Structural Research Series, No. 266, June 1963.
- 214. Harris, H. G. and Abboud, B. E., "Cyclic Shear Behavior of Horizontal Joints in Precast Concrete Large Panel Buildings." <u>Proceedings of a Workshop on the Design of Prefabricated</u> <u>Concrete Buildings for Earthquake Loads</u>, Applied Technology Council, Berkeley, CA, Dec. 1981, pp. 403-438.
- 215. Harris, H. G. and V. Caccese, "Seismic Behavior of Precast Concrete Large Panel Buildings Using a Small Shaking Table." <u>Eighth World Conference on Earthquake Engineering</u>, Vol. VI, 1984, pp. 757-764.
- 216. Harris, Harry G. and John C. Muskivitch, "Models of Precast Concrete Large Panel Buildings." <u>Journal of the Structural</u> <u>Division</u>, American Society of Civil Engineers, Vol. 106, Feb. 1980, pp. 545-560.
- 217. Hart, G. C. and Nelson, R. B., <u>Dynamic Response of</u> <u>Structures</u>, 1986.
- 218. Hatcher, D. S., Sozen, M. A. and Siess, C. P., "A Study of Tests on a Flate Plate and Flat Slab." University of Illinois Civil Engineering Studies, Structural Research Series No. 217, July 1961.
- 219. Hatcher, D. S., Sozen, M. A. and Siess, C. P., "An Experimental Study of a Quarter-Scale Reinforced Concrete Flat Slab Floor." University of Illinois Civil Engineering Studies, Structural Research Series No. 200, June 1960.

- 220. Hawkins, N. M. and comments by J. Becker, E. Elsesser, R. Englekirk, S. Firnkas, S. Freeman, H. Harris, V. Mujumdar, R. Spencer and the Author, "State-of-the-Art Report on Seismic Resistance of Prestressed and Precast concrete Structures." <u>PCI Journal</u>, Vol. 23:6, Nov/Dec 1978, pp. 83-99.
- 221. Hawkins, N. M. and Mitchell, D., "Seismic Response of Composite Shear Connections." <u>Journal of Structural</u> <u>Engineering</u>, Vol. 110, No. 9, Sept. 1984, pp. 2120-2137.
- 222. Hawkins, N. M. and Roderick, J. W., "The Behavior of Composite Beams." Research Report No. B 275, School of Civil Engineering, The University of Sydney, 1975.
- 223. Hays, C. O. and Santhanam, T. K., "Inelastic Section Response by Tangent Stiffness." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 105, No. ST7, July 1979, pp. 1241-1259.
- 224. Heidebrecht, A. C., "Dynamic Analysis of Asymmetric Wall-Frame Buildings." ASCE National Structural Engineering Convention, April 14-18, 1975, New Orleans, Louisiana, Preprint 2497.
- 225. Hinkley, A. T., "Analysis of a Shear Wall Supported by a Beam." <u>Journal of the Structural Division</u>, ASCE, Vol. 92, No. ST1, February 1966, pp. 121-130.
- 226. Hirosawa, m., Goto, T., Hiraishi, H. and Yoshimura, M., "Full-Scale Experimental Study on a Seismic Performance of Medium- Rise RC Wall Structure." Building Research Institute, Ministory of Construction, Japan, 1981.
- 227. Hirst, M. J. S. and Yeo, M. F., "The Analysis of Composite Beams Using Standard Finite Element Programs." <u>Computers and</u> <u>Structures</u>, Vol. 11, pp. 233-237.
- 228. Hoagland, R. C., "Strength and Stiffness of Screw-Fastened Roof Panels for Pole Buildings." M. S. Thesis, Iowa State University, Ames, Iowa, 1981.
- 229. Holl, D. L., "Analysis of Thin Rectangular Plates Supported on Opposite Edges." Iowa Engineering Experiment Station, Vol. XXXV, No. 30, Bulletin No. 129, December 23, 1936.
- 230. Housner, G. W., "Requirements of Seismic Analysis and Research Needs." ASCE Annual and Environmental Meeting, October 13-17, 1969, Chicago, Illinois, Preprint 1041.
- 231. Houghton, D. L. and D. J. Hall, "Elimination of Grout on Horizontal Construction Joints at Dworshak Dam." <u>ACI Journal</u>, Vol. 69, No. 3, March 1972, pp. 176-178.

- 232. Huang, H. T. and Luttrell, L. D., "Theoretical and Physical Approach to Light Gage Steel Shear Diaphragms." Department of Civil Engineering, West Virginia University, Morgantown, W. Va., 1979.
- 233. Huang, Sampson Chi-Sun, "Measurements and Interpretation of Full Scale Building Responses." Ph.D. Dissertation, University of California, Los Angeles, 1983.
- 234. Huff, W. L., "Collapse Strength of a Two-Way-Reinforced Concrete Slab Contained Within a Steel Frame Structure." Technical Reprint No. N-75-2 U.S. Army Engineer Waterways Experiment Station, Vicksburg, Miss., 1975.
- 235. Hussain, M. I. and Libove, C., "Stiffness Tests of Trapezoidally Corrugated Shear Webs." <u>Journal of the</u> <u>Structural Division</u>, ASCE, Vol. 103, No. ST5, May 1977, pp. 971-987.
- 236. Hussein, R., "Sandwich Plates with Interlayer Slip." Journal of the Engineering Mechanics, Vol. 110, No. 4, April 1984.
- 237. Hwang, T. and Scribner, C. F., "R/C Member Cyclic Response During Various Loadings." <u>Journal of the Structural</u> <u>Engineering</u>, ASCE, Vol. 110, No. 3, March 1984.
- 238. Iliya, R. and Bertero, V. V., "Effects of Amount and Arrangement of Wall-Panel Reinforcement on Hysteretic Behavior of Reinforced Concrete Walls." Report No. UCB/EERC-80/04, Earthquake Engineering Research Center, University of California, Berkeley, Calif., Feb., 1980.
- 239. Inland-Ryerson Construction Products Company, "Composite Slab Test of 3 in. V, 16 Gauge Deck for Sears Office Tower." Chicago, Illinois, 1971.
- 240. Inland-Ryerson Construction Products Company, "Diaphragm Test 3 V-32 in. Deck With 2.5 in. Lightweight Concrete." Test No. D008-71, September 1971.
- 241. Inland-Ryerson Construction Products Company, "Diaphragm Test, 3 in. V-32 in. 16 Gage Deck with 3.25 in. Lightweight Concrete." Test No. 187-73, October 1973.
- 242. Inland-Ryerson Construction Products Company, "Static Diaphragm Test, 3VF-30, 6.25 in. Slab, Lightweight Concrete, .75 in. X 5 in. Nelson Studs." September 1974.
- 243. Irwin, A. W., "Analysis of Tall Wall Buildings Including In-Plane Floor Deformations." <u>Build International</u>, Vol. 8, 1975, pp. 43-55.
- 244. Itani, Rafik Y. and Falk, Robert H., <u>Dynamic Behavior of</u> <u>Full-Scale Wood Diaphragms</u>, Washington State University.
- 245. Itani, Rafik Y. and Hiremath, Girish, "Analysis of Partial Composite Action in Diaphragms."
- 246. Itani, R. Y., Hiremath, G. S. and Hoyle, R. J., Jr., "Parameters Influencing Behavior of Diaphragms." <u>Journal of</u> <u>Structural Engineering</u>, ASCE, Vol. 109, No. 2, February 1983, pp. 369-385.
- 247. Itani, Rafik Y., Morshed, Houssain M. and Hoyle, Robert J., "Experimental Evaluation of Composite Action." Journal of the Structural Division, ASCE, Vol. 107:ST3, March 1981, pp. 551-565.
- 248. Iwan, W. D., "A Distributed-Element Model for Hysteresis and its Steady-State Dynamic Response." <u>Journal of Applied</u> <u>Mechanics</u>, ASCE, Vol. 33, No. 4, December 1966, pp. 893-900.
- 249. Iwan, W. D., "Response of Multi-Degree-of-Freedom Yielding Systems." Jorunal of the Engineering Mechanics Division, ASCE, Vol. 94, No. EM2, April 1968, pp. 421-437.
- 250. Iwan, W. D., "The Response of Simple Stiffness Degrading Structures." <u>Proceedings of the Sixth World Conference on</u> <u>EArthquake Engineering</u>, New Delhi, India, January 1977, Vol. II, pp. 1094-1099.
- 251. Iwankiw, Nestor R., "Thermal Effects on Load Capacity of Concrete Slabs." <u>Journal of the Structural Division</u>, ASCE, Vol. 105, No. ST7, July 1979, pp. 1417-1433.
- 252. Jain, S. K., "Analytical Models for the Dynamics of Buildings." Ph.D. Dissertation, 1983, California Institute of Technology, 191pp.
- 253. Jain, S. K., "Seismic Response of Buildings with Flexible Floors." <u>Journal of Engineering Mechanics</u>, Vol. 110, No. 1, Jan. 1984, pp 125-129.
- 254. Jacobsen, L. S., "Damping in Composite Structures." <u>Proceedings of the Second World Conference on Earthquake</u> <u>Engineering</u>, Tokyo and Kyoto, Japan, July 1960, Vol. II, pp. 1029-1044.
- 255. Jeing-Chang-Jeing, "Substructuring and Rigid Diaphragm System Solution in 3-D Building Analysis." <u>Eighth Conf. on</u> <u>Electronic Computation</u>, ASCE, 1983, pp. 515-525.

- 256. Jennings, Paul C., "Periodic Response of a General Yielding Structure." Journal of the Engineering Mechanics Division, ASCE, Vol. 90, EM2, April 1964, pp. 131-166.
- 257. Jennings, Paul C., "Earthquake Response of a Yielding Structure." <u>Journal of the Engineering Mechanics Division</u>, ASCE, Vol. 91, No. EM4, August 1965, pp. 41-68.
- 258. Jennings, Paul C., "Equivalent Viscous Damping for Yielding Structures." Journal of the Engineering Mechanics Division, ASCE, Vol. 94, No. EM1, February 1968, pp. 103-116.
- 259. Jensen, Vernon P., "Solutions for Certain Rectangular Slabs Continuous Over Flexible Supports." University of Illinois Bulletin, Vol. XXXV, No. 81, Bulletin No. 303, June 7, 1938.
- 260. Jirsa, J. O., Sozen, M. A. and Siess, C. P., "An Experimental Study of a Flat Slab Floor Reinforced with Welded Wire Fabric." University of Illinois Civil Engineering Studies, Structural Research Series No. 249, June 1962.
- 261. Jirsa, J. O., Sozen, M. A. and Siess, C. P., "The Effects of Pattern Loadings on Reinforced Concrete Floor Slabs." University of Illinois Civil Engineering Studies, Structural Research Series No. 269, July 1963.
- 262. Johal, L. S. and Hanson, N. W., "Design for Vertical Load on Horizontal Connections in Large Panel Structures." <u>PCI</u> <u>Journal</u>, Jan/Feb 1982, pp. 63-79.
- 263. Johnson, J. W., "Lateral Tests on Full-Scale Lumber-Sheathed Roof Diaphragms of Various Length-Width Ratios." Oregon Forest Products, Laboratory Report T-9, October 1954.
- 264. Johnson, J. W., "Lateral Test on 12-by-60 Foot Plywood-Sheathed Roof Diaphragm." Oregon Forest Products Laboratory Report T-11, January 1955.
- 265. Johnson, J. W., "Lateral Tests on 12-by-60 Foot and 20-by-80 Foot Lumber Sheathed Roof Diaphragms." Oregon Forest Products Laboratory Report T-12, October 1955.
- 266. Johnson, J. W., "Lateral Test on a 20-by-60 Foot Roof Diaphragm with Stapled Plywood Sheathing." Oregon Forest Products Laboratory Report T-13, December 1955.
- 267. Johnson, J. W., "Lateral Test of a 20-by-60 Foot Section Sheathed with Plywood Overlaid on Decking." Oregon Forest Products Laboratory Report T-29.

- 268. Johnson, J. W. and Burrows, C. H., "Lateral Test on Full-Scale Gable Roofs with Lumber Sheathing." Oregon Forest Products Laboratory Report T-19, October 1956.
- 269. Johnson, T. and Ghadiali, Z., "Load Distribution Test on Precast Hollow Core Slabs with Openings." <u>Journal of the</u> <u>Prestressed Concrete Institute</u>, Vol. 17, No. 5, Sept/Oct 1972, pp. 9-19.
- 270. Johnston, R. A. and J. O. Curtis, "Experimental Verification of Stress Skin Design of Pole-Frame Buildings." <u>Transactions</u> <u>of the American Society of Agricultural Engineers</u>, Vol. 27(1), 1984, pp. 159-164.
- 271. Kaldjian, M. J. and Fan, W. R. S., "Earthquake Response of a Ramberg-Osgood Structure." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 94, No. ST10, October 1968, pp. 2451-2465.
- 272. Kanaan, A. E. and Powell, G. H., "DRAIN-2D, A General Purpose Computer Program for Dynamic Analysis of Inelastic Plane Structures." Report No. UCB/EERC-7316, Earthquake Engineering Research Center, University of California, Berkeley, California, 1973.
- 273. Kar, Nil K., "Seismic Support: Speedy Determination of Frequency." <u>Journal of the Structural Division</u>, ASCE, Vol. 105, No. ST7, July 1979, pp. 1289-1306.
- 274. Karsan, I. D. and Jirsa, J. O., "Behavior of Concrete under Compressive Loads." <u>Journal of the Structural Division</u>, ASCE, Vol. 95, No. ST12, Dec. 1969, pp. 2543-2563.
- 275. Kemp, E. L., Sozen, M. A. and Siess, C. P., "Torsion in Reinforced Concrete." University of Illinois Civil Engineering Studies, Structural Research Series No. 226, September 1961.
- 276. Kennedy, John B. and Bali, S. K., "Rigidities of Concrete Waffle-Type Slab Structures." <u>Canadian Journal of Civil</u> <u>Engineering</u>, Vol. 6, 1979, pp. 65-74.
- 277. Kennedy, J. B., Madugula, M. K. S., Keen, R. G. and Fung, C., "Cold-Formed Steel in Composite Bridge Construction." <u>Canadian Journal of Civil Engineering</u>, Vol. 5, 1978, pp. 164-173.
- 278. Klaas, S. R., "Analysis of One-Way Cellular Steel-Deck-Reinforced Concrete Composite Floor Diaphragms." Master of Science Thesis, Iowa State University, Ames, IA, 1986.

- 279. Klaiber, F. W., Porter, M. L. and Ekberg, C. E., "Comparing the Behavior of Concentrated and Uniform Load Tests of Slab Elements Reinforced with Cold-Formed Steel Decking." Engineering Research Institute, Iowa State University, Progress Report No. 20, March 1974.
- 280. Klaiber, F. W., Porter, M. L. and Ekberg, C. E., "Determining the Effects of Shoring on Ultimate Strength and Checking an Assumption Used in Modifying the Shear-Bond Expression to Include Uniformly Loaded Systems." Engineering Research Institute, Iowa State University, Preliminiary Data Report, August 1973.
- 281. Kokusho, S. and Ogura, K., "Shear Strength and Load-Deflection CHaracteristics of Reinforced Concrete Members." <u>Proceedings of the U.S.-Japan Seminar on Earthquake</u> <u>Engineering with Emphasis on the Safety of School Buildings</u>, Sendai, Japan, September 1970, pp. 364-381.
- 282. Kolston, D. and Buchanan, B. W., "Diaphragms in Seismic Resistant Buildings." <u>Bulletin of the New Zealand National</u> <u>Society for Earthquake Engineering</u>, Vol. 13, No. 2, June 1980, pp. 162-170.
- 283. Kondo, J., "Ultimate Strength of Longitudinally Stiffened Plate Panels Subjected to Combined Axial and Lateral Loading." Fritz Engineering Laboratory Report No. 248.13, 1965.
- 284. Kondo, J. and Ostapenko, A., "Tests on Longitudinally Stiffened Plate Panels with Fixed Ends." Fritz Engineering Laboratory Report No. 248.12, 1964.
- 285. Kriegh, James D., "Arizona Slant Shear Test: A Method to Determine Epoxy Bond Strength." <u>ACI Journal</u>, Vol. 73, July 1976, pp. 372-373.
- 286. Krupicka, G. L., "The Behavior and Analysis of Steel-Deck-Reinforced Concrete Slabs with End-Span Studs." Master's Thesis, Iowa State University, Ames, IA, 1979, 97pp.
- 287. Kubic, C. R. and Daniels, J. H., "Two-Way Flexure of Steel-Deck-Reinforced Slabs." July 1978.
- 288. Kulkarni, G. G. and Ng, S. F., "Dynamic Analysis of Two Dimensional Simply Supported Orthotropic Bridge Decks." <u>Canadian Journal of Civil Engineering</u>, Vol. 5, 1978, pp. 58-69.
- 289. Kusuda, Tadao, "Buckling of Stiffened Panels in Elastic and Strain-Hardening Range." Ph. D. Dissertation, Lehigh University, 1958.

- 290. Kusuda, Tadao and Thurlimann, Burno, "Strength of Wide Flange Beams Under Combined Influence of Moment, Shear, and Axial Force." Fritz Engineering Laboratory Report No. 248.1, 1958.
- 291. Lachance, L. and Hays, C. O., "Accuracy of Composite Section Nonlinear Solutions." <u>Journal of the Structural Division</u>, ASCE, Vol. 106, No. ST11, November 1980, pp. 2203-2219.
- 292. LaGue, D. J., "Load Distribution Tests on Precast Prestressed Hollow-Core Slab Construction." <u>J. Prestr.</u> <u>Concrete Inst.</u>, Vol. 16, No. 6, Nov/Dec 1971, pp. 10-18.
- 293. Lamport, W. B., "Instantaneous Deflections of Simple One-Way Steel-Deck Reinforced Floor Slabs." Master of Science Thesis, Iowa State University, Ames, IA, 1984.
- 294. Lawson, R. M. and Nethercot, D. A., "Lateral Stability of I-Beams Restrained by Profiled Sheeting." <u>Structural Engineer</u>, Vol. 63B, March 1985, pp. 1-7, 13.
- 295. Lee, Seng-Lip, Manuel, Francisco S. and Rossow, Edwin C., "Large Deflections and Stability of Elastic Frames." <u>Journal</u> of the Engineering Mechanics Division, ASCE, Vol. 94, No. EM2, April 1968, pp. 521-547.
- 296. Leffler, R. E., "Stresses in Floors of Staggered-Truss Buildings: Precast Prestressed Concrete Floor Plank." (46.019-400(5)), United States Steel Technical Report, December 1973.
- 297. Lindgren, Soren, "Shear Flexibility." <u>Journal of the</u> <u>Structural Division</u>, ASCE, Vol. 105, No. ST10, October 1979, pp. 2117-2121.
- 298. Lowson, R. Mark and Crisinel, Michel, "Stabilisation des Poutres en Double-Te Les Toles Profilee ICCM 112." Vol. 20, No. 3, 1983.
- 299. Lukens, A. D., "Shear Strengths and Stiffnesses of Steel Clad, Timber-Framed Diaphragm Panels, and a Procedure for Applying Diaphragm Theory to Post-Frame Building Design." M. S. Thesis, Iowa State University, Ames, Iowa, 1988.
- 300. Lukens, A. D. and D. S. Bundy, "Strengths and Stiffnesses of Post-Frame Building Roof Panels." Paper No. 87-4056, 1987 Summer Meeting of ASAE, Baltimore, MD, June 28-July 1, 1987.
- 301. Luttrell, Larry D., "Structural Performance of Light Gage Steel Diaphragms." Research Report 319, Department of Structural Engineering, Cornell University, Ithaca NY, August 1965.

- 302. Luttrell, Larry D., "Light Gage Steel Shear Diaphragms." <u>Civil Engineering Department Publication</u>, West Virginia University, 1965.
- 303. Luttrell, Larry D., "Strength and Behavior of Light-Gage Steel Shear Diaphragms." Cornell Engineering Research Bulletin No. 67-1, Department of Structural Engineering, Ithaca, NY, July 1967.
- 304. Luttrell, Larry D., "Shear Diaphragms with lightweight Concrete Fill." <u>The First Speciality Conference on Cold-</u> <u>Formed Steel Structures</u>, University of Missouri-Rolla, Civil Engineering Dept., 1971, pp. 111-117.
- 305. Luttrell, Larry D., "Tentative Recomendations for the Design of Steel Deck Diaphragms." Steel Deck Institute, Westchester, Illinois, October 1972.
- 306. Luttrell, Larry D., <u>Diaphragm Design Manual</u>. Steel Deck Institute, First Edition, January 1981.
- 307. Luttrell, Larry D., <u>Diaphragm Design Manual</u>, Steel Deck Institute, Second Edition, 1987.
- 308. Luttrell, L. D. and Ellifritt, D. S., "The Strength and Stiffness of Steel Deck Subjected to In-Plane Loading." Report No. 2011, Department of Civil Engineering, West Virginia University, 1970.
- 309. Luttrell, L. D. and Huang, H. T., "Steel Deck Diaphragm Studies." Steel Deck Institute, St. Louis, Missouri, 1981.
- 310. MacLeod, Iain A., "Analysis of Tall Buildings Subject to Torsion." paper presented at Conference on Tall Buildings in Hong Kong, September 1976.
- 311. MacLeod, Iain A., "Shear Wall-Frame Interaction." <u>Portland</u> <u>Cement Association Special Publication</u>, Skokie, Illinois, 1971.
- 312. Magnus, D., <u>Analysis and Design of Pierced Shear-Walls</u>, Concrete Publications Ltd., London, 1968.
- 313. Mahin, S. A., and Bertero, V. V., "An Evaluation of Inelastic Seismic Design Spectra." ASCE Spring Convention and Exhibit, April 24-28, 1978, Pittsburgh, PA, Preprint 3278.
- 314. Martin, L. D. and Korkosz, W. J., "Connections for Precast Prestressed Concrete Buildings Including Earthquake Resistance." Prestressed Concrete Institute, March 1982.

- 315. Masri, S. F. and Safford, F. B., "Earthquake Simulation Testing of Massive Structures by Pulse Techniques." ASCE National Structural Engineering Convention, April 14-18, 1975, New Orleans, Louisiana, Preprint 2483.
- 316. Mattock, A. H., "Cyclic Shear Transfer and Type of Interface." Journal of the Structural Division, ASCE, Vol. 107, No. ST10, October 1981, pp. 1945-1964.
- 317. Mattock, A. H., "Shear Transfer Under Cyclically Reversing Loading Across an Interface Between Concretes." Washington University, Seattle, Department of Civil Engineering, June 1977.
- 318. Mattock, A. H., "Shear Transfer Under Monotonic Loading, Across an Interface Between Concretes Cast." Washington University, Seattle, Department of Civil Engineering, September 1976.
- 319. Mayes, G. T., Sozen, M. A. and Siess, C. P., "Tests on a Quarter-Scale Model of a Multiple-Panel Reinforced Concrete Flat Plate Floor." University of Illinois Civil Engineering Studies, Structural Research Series No. 181, September 1959.
- 320. McFadden, V. J., "Analysis of Diaphragms and Diaphragm Action in a Full-Scale Post-Frame Building." Master's Thesis, Iowa State University, Ames, Iowa, 1988.
- 321. McMackin, P. J., Slutter, R. G., and Fisher, J. W., "Headed Steel Anchors Under Combined Loading." <u>AISC Engineering</u> <u>Journal</u>, AISC, Second Quarter, 1973, pp. 43-52.
- 322. McNatt, J. D. and Galligan, W. L., "Wood Diaphragm Materials." <u>Proceedings of a Workshop on Design of</u> <u>Horizontal Wood Diaphragms</u>, Applied Technology Council, Nov. 19-20, 1980, pp. 17-50.
- 323. McSaveney, L. G., "Load Carrying Capacity of Unfilled Span-Deck Floor Slabs in Bearing Wall Construction." Toronto, Canada, Oct. 1972.
- 324. Mee, A. L. and Jordan, I. J., "Simplified Analysis of Staggered Wall-Beam Frames." <u>Canadian Journal of Civil</u> <u>Engineering</u>, Vol. 2, 1975, pp. 305-313.
- 325. Meyer, R. J., "Effect of Plank Depth Parameter on Seismic Resistance of Precast Hoolow-Core Plank Diaphragms." Master's Thesis, Iowa State University, Ames, Iowa, 1988, 196 pp.

- 326. Miller, C. J., "Analysis of Multistory Frames with Light Gauge Steel Panel Infills." Department of Structural Engineering, Cornell University, Report No. 349, Ithaca, NY, August 1972.
- 327. Minami, K. and Wakabayashi, M., "Seismic Resistance of Reinforced Concrete Beam-and-Column Assemblages with Emphasis on Shear Failure of Column." Sixth World Conference on Earthquake Engineering, 1977, pp. 3101-3106.
- 328. Mohammand, Y. and Hammond, J. K., "Modeling and Response of Bilinear Hysteretic Systems." <u>Journal of Engineering</u> <u>Mechanics</u>, Vol. 113, No. 7, pp. 1000-1013.
- 329. Moustafa, S. E., "Coping with Uncertainties in the Prediction of Diaphragm Loads and Capacities." <u>Proceedings</u> of a Workshop on Design of Prefabricated Concrete Buildings for Earthquake Loads, Applied Technology Council, Dec. 1980, pp. 481-502.
- 330. Moustafa, S. E., "Effectiveness of Shear-Friction Reinforcement in Shear Diaphragm Capacity of Hollow-Core Slabs." <u>Prestressed Concrete Institute Journal</u>, Jan/Feb 1981, pp. 119-133.
- 331. Nakamura, T. and Wakabayashi, M., "Reinforcing Principle and Seismic Resistance of Brick Masonry Walls." Eighth World Conference on Earthquake Engineering, 1984, pp. 661-668.
- 332. Nakashima, M., Huang, T. and Lu, L., "Experimental Study of Beam-Supported Slabs Under In-Plane Loading." <u>ACI Journal</u>, Jan/Feb 1982, pp. 59-65.
- 333. Nakashima, M., Huang, T. and Lu, L., "Seismic Resistance Characteristics of Reinforced Concrete Beam-Supported Floor Slabs in Building Structures." Fritz Engineering Report No. 422.9, Lehigh University, Bethlehem, PA, 1981.
- 334. Nakashima, M., Huang, T. and Lu, L., "Effect of Diaphragm Flexibility on Seismic Response of Building Structures." Eighth World Conference on Earthquake Engineering, 1984, pp. 735-742.
- 335. Neilsen, M. K., "Effects of Gravity Load on Composite Floor Diaphragm Behavior." Master of Science Thesis, Iowa State University, Ames, IA, 1984.
- 336. Netley, R., "Composite Deck Diaphragms." Creative Component Independent study Master's Degree Project C.E. 690 Report (unpublished), Departament of Civil Engineering, Iowa State University, Ames, IA, May 1977, 66pp.

- 337. Newmark, N. M., "A Distribution Procedure for the Analysis of Slabs Continuous Over Flexible Beams." University of Illinois Bulletin, Vol. XXXV, No. 84, Bulletin No. 304, June 17, 1938.
- 338. Newmark, N. M., D'Appolonia, E. and Austin, W. J., "Three Paper on Numerical Methods of Stress Analysis." University of Illinois Civil Engineering Studies, Structural Research Series No. 47.
- 339. Newmark, N. M. and Lepper, H. A., "Tests of Plaster-Model Slabs Subjected to Concentrated Loads." University of Illinois Bulletin, Vol. XXXVI, No. 84, Bulletin Series No. 313, June 16, 1939.
- 340. Newmark, Nathan and Rosenblueth, Emilio, <u>Fundamentals of</u> <u>Earthquake Engineering</u>, 1971, pp. 411-420.
- 341. Nilson, Arthur H., "Diaphragm Action in Light Gage Steel Construction." <u>American Iron and Steel Institute</u>, New York, 1960.
- 342. Nilson, Arthur H., "Shear Diaphragms of Light Gage Steel." <u>Journal of the Structural Division</u>, ASCE, Vol. 86, No. ST11, Nov. 1960, pp. 111-139.
- 343. Nilson, Arthur H. and Ammar, Albert R., "Finite Element Analysis of Metal Deck Shear Diaphragms." <u>Journal of the</u> <u>Structural Division</u>, ASCE, Vol. 100, No. ST4, April 1974, pp. 711-726.
- 344. Nilson, A. H. and Winter, G., <u>Design of Concrete Structures</u>, McGraw-Hill Book Company, N. Y., New York, 10th. edition.
- 345. Nishikawa, T., Hanson, R. D. and Streeter, V. L., "Characteristics Method Applied in Structural Dynamics." ASCE National Structural Engineering Convention, April 14-18, 1975, New Orleans, Louisiana, Preprint 2464.
- 346. Nirmalan, D. R., "The Effects of Concrete Strength of Steel-Deck Reinforced Slabs." Master of Science Thesis, Iowa State University, Ames, Iowa, 1981.
- 347. Nizamuddin, Z. and Bresler, B., "Fire Response of Reinforced Concrete Slabs." ASCE Spring Convention and Exhibit, April 25-29, 1977, Dallas, Texas, Preprint 2868.
- 348. Noland, J. L., "U. S. Research Plan." U.S.-Japan Coordinated Program for Masonry Building Research. Typescript, Atkinson, Noland and Associates, Boulder, Colorado, 1984.

- 349. Oesterle, R. G., et al, "Earthquake Resistant Structural Walls. Tests of Isolated Walls." Portland Cement Association, Skokie, Illinois, November 1976.
- 350. Oesterle. R. G., et al, "Web Crushing of Reinforced Concrete Structural Walls." ACI Journal, May-June 1984.
- 351. Oesterle, R., Fiorato, A., Aristizabal-Ochoa, J. and Corley, W., "Hysteretic Response of Reinforced Concrete Structural Walls." <u>Reinforced Concrete Structures Subjected to Wind and</u> <u>Earthquake Forces</u>, ACI SP-63, 1980, pp. 243-273.
- 352. Office of Civil Defense, "A Computer Program to Analyze the Dynamic Response of High Rise Buildings to Nuclear Blast Loading." Washington, DC, July 1969.
- 353. Ollgaard, J. G., Slutter, R. G., and Fisher, J. W., "Shear Strength of Stud Connectors in Lightweight and Normal- Weight Concrete." <u>AISC Engineering Journal</u>, AISC, Vol. 8, No. 2, April 1971, pp. 55-64.
- 354. Otani, S., "Hysteresis Models of Reinforced Concrete for Earthquake Response Analysis." Eighth World Conference on Earthquake Engineering, Vol. IV, 1984, pp. 551-558.
- 355. Otani, S. "Nonlinear Behavior of Reinforced Concrete Building Structures Especially Under Earthquake Motions." <u>IASS Symposium 1978, Nonlinear Behavior of Reinforced</u> <u>Concrete Structures.</u> Vol. 3. Springer-Verlag Dusseldorf: 1981. pp. 311-351.
- 356. Packard, R. G., "Slab Thickness Design for Industrial Concrete Floors on Grade." Portland Cement Association, 1976.
- 357. Paulay, Thomas, "Earthquake-Resisting Shearwalls-New Zealand Design Trends." <u>ACI Journal</u>, May-June 1980, pp. 144-152.
- 358. Paulay, T., Priestley, M. J. N. and Synge, A. J., "Ductility in Earthquake Resisting Squat Shearwalls." <u>ACI Journal</u>, July-August 1982, pp. 257-269.
- 359. Paulay, T. and Taylor, R. G., "Slabs Coupling of Earthquake-Resisting Shearwalls." <u>ACI Journal</u>, March-April 1981, pp. 130-140.
- 360. PCI Committee on Architectural Precast Concrete Joint Details, "Architectural Precast Concrete Joint Details." <u>PCI Journal</u>, Vol. 18, No. 2, March/April 1972, pp. 10-37.

- 361. <u>PCI Committee on Connection Details</u>, Summary of Basic Information on Precast Concrete Connections, <u>PCI Journal</u>, December 1969, Vol. 14, No. 6, pp. 50-58.
- 362. <u>PCI Committee on Prestress Losses</u>, "Recommendations for Estimating Prestress Losses." <u>PCI Journal</u>, Vol. 20, No. 4, July/August 1975, pp. 44-75.
- 363. PCI, <u>PCI Design Handbook on Precast and Prestressed</u> <u>Concrete</u>. 1985.
- 364. PCI, PCI Manual for the Design of Hollow Core Slabs. 1985.
- 365. PCI, <u>PCI Manual on Design of Connections for Precast</u> <u>Prestressed Concrete</u>, 1973.
- 366. Peacock, J. D. and Michell, W. F., "Precast Concrete Floors." <u>Advances in Concrete Slab Technology</u>, 1979, pp. 462-471.
- 367. Pecknold, David A., "Effective Width of Orthotropic Plate." <u>Journal of the Structural Division</u>, ASCE, Vol. 104, No. ST5, May 1978, pp. 867-872.
- 368. Petrovic, B. and Bouwkamp, J., "Test of the Model of Joint Between Floorslab and Shear Walls of a Precast Multistory Building Made of Prestressed Concrete." Sixth World Conference on Earthquake Engineering, 1977, p. 3253.
- 369. Pfeifer, D. W. and Nelson, T. A., "Tests to Determine the Lateral Distribution of Vertical Loads in a Long-Span Hollow-Core Floor Assembly." <u>PCI Journal</u>, Nov/Dec 1983, pp. 43-57.
- 370. Poland, C. D., "Practical Application of Computer Analysis to the Design of Reinforced Concrete Structures for Earthquake Forces." <u>Reinforced Concrete Structures Subjected</u> to Wind and Earthquake Forces, ACI Publication SP-36, pp. 409-436.
- 371. Popov, Egor P., Le, Dao Q. and Petersson, Hans, "Program SUBWALL: Finite Element Analysis of Structural Walls." <u>ACI</u> <u>Journal</u>, June 1979, pp. 679-696.
- 372. Porter, M. L., "Diaphragm Floor Slabs for TCCMAR Study." <u>Proceedings of the Second Meeting of the Joint Technical</u> <u>Coordinating Committee on Masonry Research with U.S. and</u> <u>Japanese Researchers</u>, Keystone, CO, Sept. 8-10, 1986.
- 373. Porter, M. L., et. al., "A Preliminary Data Summary of Diaphragm Tests 1-3." Report for TCCMAR, Iowa State University, Ames, Iowa, Nov. 1986.

- 374. Porter, M. L., et. al., "Concrete Plank Diaphragm Characteristics." Report for TCCMAR, Iowa State University, Ames, Iowa, Feb. 1987.
- 375. Porter, M. L., et. al., "A Preliminary Data Summary of Diaphragm Tests 1-6." Report for TCCMAR, Iowa State University, Ames, Iowa, March 1987.
- 376. Porter, M. L., et. al., "Plank Diaphragm Characteristics." Final Report for TCCMAR. Engineering Research Institute, Iowa State University, Ames, Iowa, 1990.
- 377. Porter, M. L. and W. S. Easterling, "Results of Full-Scale Tests of Steel-Deck Reinforced Concrete Floor Diaphragms." <u>Proceedings of Third U. S. National Conference on Earthquake</u> <u>Engineering</u>, Charleston, SC, Aug. 25-26, 1986.
- 378. Porter, M. L. and Easterling, W. S., "Behavior, Analysis, and Design of Steel-Deck-Reinforced Concrete Diaphragms." Final Report submitted to the National Science Foundation. Iowa State University, Ames, Iowa, March 1988.
- 379. Porter, M. L. and Ekberg, C. E., "Summary of Full-Scale Laboratory Tests of Concrete Slabs Reinforced with Cold-Formed Steel Decking." Ninth Congress, IABSE, Amsterdam, 1972.
- 380. Porter, M. L. and Ekberg, C. E., "Behavior of Concrete Slabs Reinforced with Three-Inch-Deep Cold-Forming Steel Decking." Engineering Research Institute, Iowa State University, Preprint Project 822-5, May 1973.
- 381. Porter, M. L. and Ekberg, C. E., "Design vs. Test Results for Steel Deck Floor Slabs." Engineering Research Institute, Iowa State University, Preprint Project 101, July 1975.
- 382. Porter, M. L. and Ekberg, C. E., "Design Recommendations for Steel Deck Floor Slabs." <u>Journal of the Structural Division</u> <u>Proceedings</u>, ASCE, Preprint 0101, February 1976.
- 383. Porter, M. L. and Ekberg, C. E., "Behavior of Steel-Deck-Reinforced Slabs." Journal of the Structural Division, ASCE, Vol. 103, No. ST3, March 1977, pp. 663-677.
- 384. Porter, M. L. and Ekberg, C. E., "Compendium of ISU Research Conducted on Cold-Formed Steel-Deck-Reinforced Slab Systems." Draft Copy, June 1977.
- 385. Porter, M. L., Ekberg, C. E., Greimann, L. F. and Elleby, H. A., "Shear-Bond Analysis of Steel-Deck-Reinforced Slabs." Engineering Research Institute, Iowa State University, Preprint Project 0101, April 1976.

- 386. Porter, M. L. and L. F. Greimann, "Earthquake Resistance of Composite Floor Diaphragms." <u>Proceedings Fifth National</u> <u>Meeting Univesities Council for Earthquake Engineering</u> <u>Research</u>, Massachusetts Institute of Technology, June 23, 1978.
- 387. Porter, M. L. and L. F. Greimann, "Composite Floor Diaphragms." <u>Proceedings Sixth National Meeting Universities</u> <u>Council for Earthquake Engineering Research</u>, University of Illinois, Urbana-Champaign, May 1, 1980.
- 388. Porter, M. L. and L. F. Greimann, "Composite Floor Diaphragm Slab Tests." <u>Proceedings of hte Eighth World Conference on</u> <u>Earthquake Engineering</u>, San Francisco, CA, July 21-28, 1984.
- 389. Porter, M. L. and L. F. Greimann, "Pilot Tests of Composite Floor Diaphragms." <u>Proceedings of Third Canadian Conference</u> <u>on Earthquake Engineering</u>, McGill University, Montreal, Canada, June 1979, 24pp.
- 390. Porter, M. L. and L. F. Greimann, "Seismic Resistance of Composite Floor Diaphragm." ISU-ERI-Ames-80133, Iowa State University, Ames, IA, Final Report Submitted to National Science Foundation, May 1980, 174pp.
- 391. Porter, M. L. and L. F. Greimann, "Shear-Bond Strength of Studded Steel Deck Slabs." <u>Proceedings Seventh International</u> <u>Speciality Conference on Cold-Formed Steel Structures</u>, St. Louis, Mo., Nov. 13-14, 1984.
- 392. Porter, M. L., Greimann, L. F., Elleby, H. A. and Ekberg, C. E., "Regression Analysis for Prediction of Shear-Bond Failure of One-Way Slab Elements Reinforced with Cold-Formed Steel Decking." Engineering Research Institute, Iowa State University, Progress Report 23, July 1974.
- 393. Porter, M. L., Klaiber, F. W. and Ekberg, C. E., "An Investigation of the Effects of Various Shoring Conditions on the Ultimate Strength of Steel-Deck Reinforced Slab Elements." Engineering Research Institute, Iowa State University, Progress Report 21, April 1974.
- 394. Porter, M. L., Timm, L. W. and Ekberg, C. E., "Manufacturer's Test Data of One-Way Concrete Slab Elements Reinforced with Cold-Formed Steel Decking." Engineering Research Institute, Iowa State University, Progress Report 24, June 1974.
- 395. Portland Cement Association, "Analysis of Small Reinforced Concrete Buildings for Earthquake Forces." Skokie, Illinois.

- 396. Portland Cement Association, "Design of Combined Frames and Shear Walls." Advanced Engineering Bulletin, Skokie, Illinois, 1965.
- 397. Portland Cement Association, "Frame Constants for Lateral Loads on Multistory Concrete Buildings." Advanced Engineering Bulletin, Skokie, Illinois, 1962.
- 398. Portland Cement Association, "Simplified Design of Concrete Floor Systems." Chicago, Illinois.
- 399. Portland Cement Association, "The Behavior of Reinforced Concrete Buildings Subjected to the Chilean Earthquake of May 1960." Advanced Engineering Bulletin, Skokie, Illinois, 1963.
- 400. Portland Cement Association, "Design and Construction of Large-Panel Concrete Structures." Reports 1-4, Apr. 1975 -Oct. 1977.
- 401. Prescott, W. S., Ang, A. and Siess, C. P., "Analysis of Clamped Square Plates Containing Openings with Stiffened Edges." University of Illinois Civil Engineering Studies, Structural Research Series No. 229, November 1961.
- 402. Prestressed Concrete Manufacturers of California, <u>Plant</u> <u>Cast, Precast, and Prestressed Concrete, A Design Guide</u>. 1980.
- 403. Prestressed Concrete Operations, <u>Prestressed Span-Deck</u> <u>Catalog</u>, Iowa Falls, Iowa, ca. 1985.
- 404. Prins, M. D., "Elemental Tests for the Seismic Resistance of Composite Floor Diaphragms." Unpublished Master's Thesis, Iowa State University, Ames, Iowa, 1985.
- 405. Priestley, M. J. N., "Masonry Structural Systems for Regions of High Seismicity." <u>Proceedings of the Seventh World</u> <u>Conference on Earthquake Engineering</u>, Istanbul, Turkey, Vol. 4, Sept. 1980, pp. 441-448.
- 406. Rabbat, B. G. and N. W. Hanson, "Fatigue Tests of Bolted Connections Designed by Shear Friction." Transportation Research Record 664, <u>Bridge Engineering</u>, Vol. I, pp. 145-152.
- 407. Ramakrishnan, V., Vadivelu, S. K. and Prasad, N. M., "Shear Wall Structures -- A Bibliography of Their Design, Analysis, Construction and Testing." Civil Engineering Department, South Dakota School of Mines and Technology, Rapid City, South Dakota, 1974.

- 408. Ramberg, W. and Osgood, W. R., "Description of Stress-Strain Curves by Three Parameters." <u>Technical Notes for U.S.</u> <u>National Advisory Committee for Aeronautics</u>, National Bureau of Standards, No. 902.
- 409. Raths, Raths, and Johnson, Inc., "Span-Deck Diaphragm Design." 1970.
- 410. Reichard, T. W., Carpenter, E. P. and Leyendecker, E. V., "Design Loads for Inserts Embedded in Concrete." National Bureau of Standards, Report No. NBS-BSS-42, Washington, DC, April 1972.
- 411. Reinhardt, H. W. and Hartjes, H. W., "Invloed van de Voeglente op de Afschuifsterke van Voegen Tussen Geprefabricereerde Kanaalplanten." <u>Cement</u>, 34, No. 3, 1982, pp. 185-187.
- 412. Richart, F. E. and Kluge, R. W., "Tests of Reinforced Concrete Slabs Subjected to Concentrated Loads." University of Illinois Bulletin, Vol. XXXVI, No. 85, Bulletin Series No. 314, June 20, 1939.
- 413. Riddell, R. and N. M. Newmark, "Technical Notes: Force-Deformation Models for Non-Linear Analysis." <u>Journal of the</u> <u>Structural Division</u>, ASCE. Vol. 105, No. ST12, Dec. 1979, pp. 2773-2778.
- 414. Rizzo, P. C., Shaw, D. E. and Snyder, M. D., "Vertical Seismic Response Spectra." ASCE National Structural Engineering Convention, April 14-18, 1975, New Orleans, Louisiana, Preprint 2478.
- 415. Rizzo, S. and Fazio, P., "Sandwich-Panel Assemblies: Analytical Model." Journal of Structural Engineering, ASCE, Vol. 109, No. 11, November 1983, pp. 2715-2732.
- 416. Roark, R. J. and Young, W. C., <u>Formulas for Stress and</u> <u>Strain</u>, N. Y., New York: McGraw-Hill Book Company, 1975.
- 417. Robertson, H. H. and Co., Diaphragm Information Including Load Tables.
- 418. Robinson, H., Fahmy, E. H. and Azmi, M. H., "Composite Open-Web Joists with Formed Metal Floor." <u>Canadian Journal of</u> <u>Civil Engineering</u>, Vol. 5, No. 1, March 1978, pp. 1-10.
- 419. Rockey, K. C. and Skaloud, M., "Influence of Flange Stiffness upon the Load Carrying of Webs in Shear." 8th Congress-New York-Theme Ii C/2.

- 420. Rodriguez, J. J., Bianchini, A. C., Viest, I. M. and Kesler, C. E., "Shear Strength of Two-Span Continuous Reinforced Concrete Beams." <u>Journal of the American Concrete Institute</u>, April 1959, pp. 1089-1130.
- 421. Roeder, Charles W., "Point Loads on Composite Deck-Reinforced." Journal of the Structural Division, ASCE, Vol. 107, No. ST12, Dec. 1981, pp. 2421-2429.
- 422. Roeder, C. W. and Hawkins, N. M., "Connections Between Steel Frames and Concrete Walls." <u>AISC Engineering Journal</u>, First Quarter, 1981, pp. 22-29.
- 423. Rosenblueth, E., <u>Design of Earthquake Resistant Structures</u>. 1980.
- 424. Rosenthal, I., "Full Scale Test of Continuos Prestressed Hollow-Core Slab." <u>Journal of the Prestressed Concrete</u> <u>Institute</u>, Vol. 23:3, May/June 1978, pp. 74-81.
- 425. Rotter, J. M. and Ansourian, P., "Cross-Section Behaviour and Ductility in Composite Beams." Research Report No. R329, School of Engineering, The University of Sydney, September 1978.
- 426. Rutenberg, A., "Laterally Loaded Flexible Diaphragm Buildings." <u>Journal of the Structural Division</u>, Sept. 1980, pp. 1969-1973.
- 427. Saatcioglu, M., Derecho, A. and Corley, W., "Modeling Hysteretic Behavior of Cuopled Walls for Dynamic Analysis." <u>Earthquake Engineering and Structural Dynamics</u>, Vol. 11, 1983, pp. 711-726.
- 428. Sabnis, G. M., Harris, H. G., White, R. N. and Mirza, S. M., <u>Structural Modeling and Experimental Techniques</u>. Prentice-Hall Inc., Inglewood Cliffs, New Jersey, 1983.
- 429. Saiidi, Mehdi, "Simple Nonlinear Modelling of Earthquake Response in Torsionally Coupled R/C Structure -- A Preliminary Study." College of Engineering Report No. 60, University of Nevada, Reno, July 1982.
- 430. Saiidi, Mehdi and Hadson, Kenneth E., Jr., "Analytical Study of Irregular R/C Structure Subjected to In-Plane Earthquake Loads." College of Engineering Report No. 59, University of Nevada, Reno, 1982.

- 431. Saiidi, M. and Sozen, M. A., "Simple and Complex Models for Nonlinear Seismic Response of Reinforced Concrete Structures." University of Illinois, Urbana-Champaign, <u>Structural Research Series</u> #4, Champaign: University of Illinois Press, 1979.
- 432. Sandhu, Balvir S., "Dynamic Analysis of Multistory Buildings." <u>AISC Engineering Journal</u>, Third Quarter, 1974, pp. 67-72.
- 433. Scalzi, John B., "The Staggered Truss System -- Structural Considerations." <u>AISC Engineering Journal</u>, Vol. 8, No. 4, October 1971, pp. 138-143.
- 434. Schaffausen, R. J. and Wegmuller, A. W., "Multistory Rigid Frames with Composite Girders Under Gravity and Lateral Forces." <u>AISC Engineering Journal</u>, Vol. 14, No. 2, Second Quarter, 1977, pp. 68-77.
- 435. Schall, R. E. and Stratta, James L. eds., "Coalinga, California, Erthquake of May 2, 1983, Reconnaissance Report." Earthquake Engineering Research Institute, Report No. 84-03, January 1984.
- 436. Schultz, D. M. and Fintel, M., "Design and Construction of Large-Panel Concrete Structures." Contract No. H-2131R, U.
 S. Department of Housing and Urban Development, 1975.
- 437. Schuster, R. M. and Ekberg, C. E., "Tentative Recommendations for the Design of Cold-Formed Steel Decking as Reinforcement for Concrete Floor Slabs." Engineering Research Institute, Iowa State University, Progress Report 7, January 1970.
- 438. Schuster, R. M. and Ling, W. C., "Mechanical Interlocking Capacity of Composite Slabs." 5th Speciality Conference on Cold-Formed Steel Deck, St. Louis, 1980.
- 439. Scott, N. L., "Performance of Precast Prestressed Hollow Core Slab with Composite Concrete Topping." <u>J. Prestressed</u> <u>Concrete Inst.</u>, Vol. 18, No. 2, March-April, 1973, pp. 64-77.
- 440. Scriber, C. F., et al, "Delaying Shear Strength Decay in Reinforced Concrete Flexural Members under Large Loads Reversals." Michigan University, Ann Arbor, Michigan, Department of Civil Engineering, May 1978.
- 441. Shewmaker, R. E., Yanthakis, M. and Sozen, C. P., "Very Small Scale Reinforced Concrete Multi-Panel Flat Slabs." University of Illinois Civil Engineering Studies, Structural Research Series No. 265, June 1963.

- 442. Shieh, R. C., "Slope-Deflection Method for Elastic-Viscoplastic Frames." Journal of the Engineering Mechanics <u>Division</u>, ASCE, Vol. 105, No. EM6, December 1979, pp. 953-969.
- 443. Shing, P. S. and Mahin, S. A., "Pseudodynamic Test Method for Seismic Performance Evaluation: Theory & Implementation." January 1984.
- 444. Shinozuka, M., Itagaki, H. and Hakuno, M., "Dynamic Safety Analysis of Multistory Buildings." <u>Journal of the Structural</u> <u>Division</u>, ASCE, Vol. 94, No. ST1, January 1968, pp. 309-330.
- 445. Shoub, H. and Ingberg, S. H., "Fire Resistance of Steel Deck Floor Assemblies." <u>Building Science Series II</u>, National Bureau of Standards, Washington, DC, 1967.
- 446. Simmonds, S. H. and Siess, C. P., "Effects of Column Stiffness on the Moments in Two-Way Floor Slabs." University of Illinois Civil Engineering Studies, Structural Research Series No. 253, July 1962.
- 447. Singleton, R. C., "Fasteners for Steel and Concrete Composite Construction."
- 448. Sinno, Raouf and Howard Furr, "Bonded Concrete Overlays." Journal of the Structural Division, American Society of Civil Engineers, Vol. 96, No. ST8, August 1970, pp. 1627-1638.
- 449. Smith, E. F., "An Analysis of the Dynamic Strengths of Reinforced Concrete Slab." Army Engineering Waterways Experiment Station, Vicksburg, Miss., April 1968.
- 450. Smith, K. N. and Vantsiotis, A. S., "Deep Beam Test Results Compared with Present Building Code Models." <u>ACI Journal</u>, July-August 1982, pp. 280-287.
- 451. Smith, K. N. and Vantsiotis, A. S., "Shear Strength of Deep Beams." <u>ACI Journal</u>, May-June 1982, pp. 201-213.
- 452. Sozen, M. A., "Hysteresis in Structural Elements." W. D. Iwan (ed.), <u>Applied Mechanics in EArthquake Engineering, AMD</u>, Vol. 8, ASME, New York, New York, 1974, pp. 63-98.
- 453. Spencer, R., "Serviceability of Precast Diaphragms and Conncections Designed for Earthquake Resistant Structures." <u>Proceedings of a Workshop on Design of Prefabricated Concrete</u> <u>Buildings for Earthquake Loads</u>, Applied Technology Council, Dec. 1980, pp. 551-573.
- 454. Spencer, R. and Neille, D., "Cyclic Tests of Welded Headed Stud Connections." <u>PCI Journal</u>, May/June 1976, pp. 71-83.

- 455. Spencer, R. and Tong, W., "Design of a One-Story Precast Concrete Building for Earthquake Loading." Eight world Conference on Earthquake Engineering, Vol. V, pp. 653-660.
- 456. Srichatrapimuk, T., "Earthquake Response of Coupled Shear Wall Buildings." Earthquake Engineering Research Center Report No. EERC 76-27, University of California Berkeley, November 1976.
- 457. Stanton, J., "Connection in Precast Concrete Structures." <u>Concrete International Design and Construction</u>, 9, No. 11, 1987, pp. 49-53.
- 458. Stanton, J., "Moment Resistant Connections and Simple Connections." Prestressed Concrete Institute, Chicago, Illinois, Research Project No. 1/4, 1986.
- 459. Stillinger, J. R., "Lateral Tests on Full-Scale Lumber-Sheathed Roof Diaphragms." Oregon Forest Products, Laboratory Report T-6, October 1953.
- 460. Stillinger, J. R., "Lateral Tests on Full-Scale Gable Roofs with Lumber Sheathing." Oregon Forest Products Laboratory Report T-19, October 1956.
- 461. Stillinger, J. R. and Countryman, D., "Lateral Tests on Full-Scale Plywood-Sheathed Roof Diaphragm." Oregon Forest Products Laboratory Report T-5, October 1953.
- 462. Stillinger, J. R., Johnson, J. W. and Overholser, J. L., "Lateral Tests on Full-Scale Lumber Sheathed Diaphragms." Oregon Forest Products, Laboratory Report T-3, October 1952.
- 463. Strand, Donald R., "1974 SEAOC Lateral Force Requirements and History." ASCE National Structural Engineering Convention, April 14-18, 1975, New Orleans, Louisiana, Preprint 2459.
- 464. Structural Engineers Association of California, <u>Recommended</u> <u>Lateral Force Requirements and Commentary</u>, Seismology Committee, 1976.
- 465. Structural Engineers Association of Southern California, <u>Seismic Analysis by Computer</u>. Recommended Guidelines for the Performance and Presentation of Computer Generated Seismic Analysis of Structures, prepared by The Electronic Computation Committee, 1977.
- 466. Sucuoglu, H. and Aktan, A. E., "Hysteretic Response of Reinforced Concrete Frames." <u>Reinforced Concrete Structures</u> <u>Subjected to Wind and Earthquake Forces</u>, ACI Publication SP-63, 1980, pp. 153-171.

- 467. Szulczynski, T. and Sozen, M. A., "Load-Deformation Characteristics of Concrete Prisons with Rectilinear Transverse Reinforcement." University of Illinois Civil Engineering Studies, Structural Research Series No. 224, September 1961.
- 468. Tai, J. and Kulka, F., "An Anatomy of Diaphragms." <u>Proceedings of a Workshop on the Design of Prefabricated</u> <u>Concrete Buildings for Earthquake Loads</u>, Applied Technology Council, Berkeley, CA, Dec. 1981, pp. 441-461.
- 469. Takayanagi, T., Scanlon, A. and Corley, W. G., "Earthquake Resistant Structural Walls-Analysis of Coupled Wall Systems." Construction Technology Laboratories, PCA, report submitted to NSF under Grant No. ENV77-15333, 1981.
- 470. Takayanagi, T. and Schnobrich, W. C., "Computed Behavior of Reinforced Concrete Coupled Shear Walls." University of Illinois, Urbana-Champaign, <u>Structural Research Series</u> #4, Champaign: Univ. of Illinois Press, 1976.
- 471. Takeda, T., Sozen, M. and Nielson, N., "Reinforced Concrete Response to Simulated Earthquakes." <u>Journal of the</u> <u>Structural Division</u>, ASCE, Vol. 96, No. ST12, Dec. 1970, pp. 2557-2573.
- 472. Tassios, T. P. and Koroneos, E. G., "Local Bond-Slip Relationships by Means of the Moire Method." <u>Journal of the</u> <u>American Concrete Institute</u>, January-February 1984.
- 473. Tembulkar, J. and Nau, J., "Seismic Energy and Structural Damage." Third U.S. National Conference on Earthquake Engineering, Vol. II, 1986, pp. 979-990.
- 474. Testa, Rene B. and Norris Stubbs, "Bond Failure and Inelastic Response of Concrete." <u>Journal of the Engineering</u> <u>Mechanics Division</u>, April 1977, pp. 295-310.
- 475. Thomas, G. R. and Petalas, N., "Seismic Response of Shear Wall-Frame Systems."
- 476. Thorpe, D. D., "One-Way Push-Out Tests for Composite Steel Deck Reinforced Concrete Slabs." Creative Component Independent Study Master's Degree Project C.E. 690 Report (unpublished), Department of Civil Engineering, Iowa State University, Ames, IA, Dec. 1979, 47pp.
- 477. Tissell, J. R. and Elliot, J. R., "Plywood Diaphragms." American Plywood Association, Research Report 138. Tacoma WA, 1983.
- 478. Tissell, J. R., "1966 Horizontal Plywood Diaphragm Tests." American Plywood Association Laboratory Report 106.

- 479. Toakley, A. Raymond, "Optimum Design Using Available Sections." <u>Journal of the Structural Division</u>, ASCE, Vol. 94, No. ST5, May 1968, pp. 1219-1241.
- 480. Torkamani, M. A. M. and Hart, G. C., "Earthquake Engineering: Parameter Identification." ASCE National Structural Engineering Convention, April 14-18, 1975, New Orleans, Louisiana, Preprint 2499.
- 481. Tremel, P. M., "Boundary Conditions and Orientation Behavioral Characteristics of Hollow-Core Diaphragms." Master's Thesis, Iowa State University, Ames, Iowa, pp. 1988, 234.
- 482. Tsuiji, T., "Strength of Longitudinally Stiffened Plate Panels with Large b/t." Fritz Engineering Report No. 248.14, 1965.
- 483. U. S. Department of Housing and Urban Development, Office of Policy Development and Research, <u>Design and Construction of</u> <u>Large-Panel Concrete Structures -Supplement Report B</u> <u>"Horizontal Joint Tests"</u>, Nov 1978.
- 484. U. S. Department of Housing and Urban Development, Office of Policy Development and Research, <u>Design and Construction of</u> <u>Large Panel Concrete Structures -Supplemental Report C, "</u> <u>Seismic Tests on Horizontal Joints.</u>". Nov. 1978.
- 485. U. S. Japan Research, <u>Earthquake Effects on Reinforced</u> <u>Concrete Structures</u>. ACI SP-84, 1985.
- 486. Umemura, H. and Takizawa, H., "Dynamic Response of Reinforced Concrete Buildings." International Association for Bridge and Structural Engineering, 1982.
- 487. United States Steel, "Technical Report-Staggered Framing Truss Systems for High-Rise Buildings." USS, 1976.
- 488. Vallenas, J. M., Bertero, V. V. and Popov, E. P., "Hysteretic Behavior of Reinforced Concrete Structural Walls." Report No. UCB/EERC-79/20, Earthquake Engineering Research Center, University of California, Berkeley, California, August 1979.
- 489. Vanderbilt, M. D., Sozen, M. A. and Siess, C. P., "An Experimental Study of a Reinforced Concrete Two-Way Floor Slab with Shallow Beams." University of Illinois Civil Engineering Studies, Structural Research Series No. 228, October 1961.

- 490. Vanderbilt, M. D., Sozen, M. A. and Siess, C. P., "Deflections of Reinforced Concrete Floor Slabs." University of Illinois Civil Engineering Studies, Structural Research Series No. 263, April 1963.
- 491. Veletsos A. S. and N. M. Newmark, "Effect of Inelastic Behavior on the Response of Simple Systems to Earthquake Motions." <u>Proceedings of the Second World Conference on</u> <u>Earthquake Engineering</u>, Tokyo and Kyoto, Japan, July 11-18, 1960, Volume II, pp. 895-912.
- 492. Velkov, M., Ivkovich, M. and Perishich, Z., "Experimental and Analytical Investigation of Prefabricated Large Panel Systems to be Constructed in Seismic Regions." Eighth World Conference on Earthquake Engineering, 1984, pp. 773-780.
- 493. Viest, I. M., "Investigation of Stud Shear Connections for Composite Concrete and Steel T-Beams." <u>Journal of the</u> <u>American Concrete Institute</u>, Vol. 27, No. 8, April 1956, pp. 875-891.
- 494. Villanueva, A. S., "Ductility Reduction Factors for Earthquake Design." ASCE Spring Convention and Exhibit, April 24-28, 1978, Pittsburgh, PA, Preprint 3209.
- 495. Wakabayashi, Minoru, et al, "Hysteretic Behavior of Steel Braces Subjected to Horizontal Load due to Earthquake." Sixth World Conference on Earthquake Engineering, 1977, pp. 3188-3194.
- 496. Wang, Pao-Tsan, et al, "Complete Stress-Strain Curve of Concrete and its Effects on Ductility of Reinforced Concrete." Illinois University at Chicago Circle, Department of Materials Engineering, October 1977.
- 497. Wang, T. Y., Bertero, V. V. and Popov, E. P., "Hysteretic Behavior of Reinforced Concrete Framed Walls." Report No. EERC 75-23, Earthquake Engineering Research Center, University of California, Berkeley, Calif., 1975.
- 498. Warnick, W. L. and Walston, W. H., Jr., "Secondary Deflections and Lateral Stability of Beams." <u>Journal of the</u> <u>Engineering Mechanics Division</u>, ASCE, Vol. 106, No. EM6, December 1980, pp. 1307-1325.
- 499. Washaw, C., "Performance of Aluminum Shear Diaphragms." Master's Thesis, Cornell University, Ithaca, New York.
- 500. Watabe, M., Hirosawa, M. and Nakota, S., "Joint Research Program Utilizing the Large Scale Testing Facilities." 7th USNR Panel Conference, US Department of Commerce.

- 501. White, Anthony G., "Earthquakes and Cities: A Selected Bibliography." Council of Planning Librarians, Exchange Bibliography #1109, pp. 1-9.
- 502. White, R. N., "Diaphragm Action of Aluminum-Clad Timber-Frame Buildings." <u>American Society of Agricultural</u> <u>Engineers</u>, ASAE 78-4501, St. Joseph, MI.
- 503. Whitman, R. V., Heger, F. J., Luft, R. W. and Krimgold, F., "Evaluation of Seismic Resistance of Existing Buildings." ASCE Spring Convention and Exhibit, April 24-28, 1978, Pittsburgh, PA, Preprint 3264.
- 504. Whitney, C. S., "Plastic Theory of Reinforced Concrete Design." <u>American Society of Civil Engineers, Transactions</u>, Vol. 107, Paper No. 2133, 1942, pp. 251-326.
- 505. Wiesinger, F. P., "Flexural Analysis of Coplanar Frames." Journal of the Structural Division, ASCE, Vol. 92, No. ST1, February 1966, pp. 131-160.
- 506. Wilhoite, Gene M., "Introduction to Cold-Formed Shapes for Transmission Structures." ASCE Spring Convention in Atlanta, May 1984.
- 507. Winokur, A. and Gluck, J., "Lateral Loads in Asymmetric Multistory Structures." <u>Journal of the Structural Division</u>, ASCE, Vol. 94, No. ST3, March 1968, pp. 645-656.
- 508. Wiss, Janney, Elstner and Associates, "Second Shear Test of Hypar Metal Decking for the Roof of the American Airline Hangers." Report No. 9395, Northbrook, Illinois, 1969.
- 509. Wiss, Janney, Elstner and Associates, "Shear Test of Hypar Metal Decking to be used in the Roof of the American Airline Hangers." Report No. 9139, Northbrook, Illinois, October 1969.
- 510. Woodring, R. E. and Siess, C. P., "An Analytical Study of the Moments in Continuous Slabs Subjected to Concentrated Loads." University of Illinois Civil Engineering Studies, Structural Research Series, No. 264, May 1963.
- 511. Wright, Haskell E. and Ned H. Burns, "Deflection of Double Tees and Hollow-Cored Slabs with High Tension Allowed in Concrete." <u>PCI Journal</u>, Vol. 19, No. 4, pp. 100-118.
- 512. Wyllie, Loring a., Jr., et. al., "Effects on Structures of the Managua Earthquake of December 23, 1972." <u>Bulletin of</u> <u>the Seismological Society of America</u>, Vol. 64, No. 4, August 1974, pp. 1069-1133.

- Yamada, M., et al, "Cyclic Deformation Behavior of Reinforced Concrete Shear Walls." Sixth World Conference on 513. Earthquake Engineering, 1977, pp. 3075-3080.
- "Moment-Rotation 514. Yameshiro, R. and Siess, C. P., Characteristics of Reinforced Concrete Members Subjected to Bending, Shear and Axial Load." University of Illinois Civil Engineering Studies, Structural Research Series No. 260, December 1962.
- Yancey, C. W. C., "The Development of an Improved Test for 515. Evaluating the racking Resistance of Wall Panels." Center for Building Technology, NBS, Washington, DC, 1976.
- Yao, J. T. P. and Tang J., "Active Control of Civil 516. Engineering Structures." Structural Engineering Report No. CE-STR-73-1, Purdue University, 1973.
- 517. Yonezawa, H. and Mikami, I., "Elastic Buckling of Plate Girders from Pure Bending." Journal of the Engineering Mechanics Division, ASCE, Vol. 94, No. EM1, February 1968, pp. 211-224.
- Young, D. H. and Medearis, K., "An Investigation of the 518. Structural Damping Characteristics of Composite Wood Structures Subjected to Cyclic Loading." Department of Civil Engineering, Stanford University, Technical Report No. 11, Stanford, California, April 1962.
- 519. Younis, Christos J. and Tadjbakhsh, I. G., "Response of Sliding Rigid Structure to Base Excitation." Journal of the Engineering Mechanics, ASCE, Vol. 110, No. 2, March 1984.
- 520. Yura, J. A., "Stability of Structures." T. R. Higgins Award Presentation, 1974.
- Yura, J. A., "The Effective Length of Columns in Unbraced 521.
- Frames." <u>AISC Engineering Journal</u>, April 1971, pp. 37-42. Zavelani, Adolfo, "Design Recommendations for Cold-Formed Tipped Angle Sections." ASCE Spring Convention in Atlanta, 522. May 1984.
- Zimmerman, T. J. E. and Bjorhovde, R., "Analysis and Design of Stub Girders." Department of Civil Engineering, 523. University of Alberta, Edwoutors, Alberta, 1981.
- American Concrete Institute, Building Code Requirements for 524. Reinforced Concrete (ACI 318-83). Detroit: ACI, 1983.
- 525. American Institute of Steel Construction, Load and Resistance Factor Design Manual of Steel Construction. Chicago: AISC, 1986.

- 526. American Institute of Steel Construction, <u>Manual of Steel</u> <u>Construction</u>. Chicago: AISC, 1980.
- 527. American Welding Society Structural Welding Committee, <u>Structural Welding Code - Sheet Steel.</u> ANSI/AWS D1.3-81. Miami: AWS.
- 528. American Iron and Steel Institute, <u>Specification for the</u> <u>Design of Cold-Formed Steel Structural Members.</u> Washington, D. C.: AISI, 1986.
- 529. American Plywood Association, <u>Plywood Design Specification.</u> Tacoma Wa, 1986.
- 530. National Forest Products Association, <u>National Design</u> <u>Specifications for wood.</u> 1977.
- 531. International Conference of Building Officials, <u>Uniform</u> <u>Building Code.</u> Whittier CA, 1985.

8. APPENDIX A: DIAPHRAGM DATA

8.1 Cold-Formed Steel Deck

8.1.1 Data obtained from Reference 25, 305, and 306

TABLE A1. WELDED DIAPHRAGM TEST DATA

TEST #	SHEAR	SHEAR	DISPL.	6	FAILURE
	STIFFNESS	STRENGTH	PEAK		MODE
	K/1n.	pli	IN.		
W-1	15.2	624	NR		NR
W-2	18.4	605	NR		NR
W-3	10.6	320	NR		NR
W-4	14.1	650	NR		NR
₩-5	11.1	401	NR		NR
W-6	12.0	461	NR		NR
W-7	16.2	508	NR		NR
W-8	12.1	510	NR		NR
W-9	9.4	467	NR		NR
W-10	40.3	1040	NR		NR
W-11	21.6	640	NR		NR
₩-12	20.7	630	NR		NR
W-13	20.6	580	NR		NR
W-14	15.9	560	NR		NR
W-15	16.9	450	NR		NR
W-16	23.2	505	NR		NR
W-17	14.4	339	NR		NR
W-18	21.3	830	NR		NR
W-19	23.0	807	NR		NR
W-20	24.9	912	NR		NR
W-21	14.2	790	NR		NR
₩-22	67.7	995	NR		NR
W-23	87.2	1090	NR		NR
WB-1	32.3	1260	NR		NR
WB-2	12.6	495	NR		NR
WB-3	11.8	480	NR		NR
WB-4	16.5	775	NR		NR
WB-5	16.0	525	NR		NR
WB-6	16.6	580	NR		NR
WB-7	10.6	492	NR		NR
WB-8	12.7	615	NR		NR
WB-9	11.5	423	NR		NR
WB-10	8.3	311	NR		NR
WB-11	12.8	339	NR		NR
WB-12	32.4	1577	NR		NR
WB-13	19.4	1125	NR		NR
WB-14	14.4	719	NR		NR
WB-15	13.8	694	NR		NR

TABLE A1 CONTINUATION

TEST #	SHEAR STIFFNESS	SHEAR STRENGTH	DISPL. @ PEAK	FAILURE MODE
	K/in.	plf	IN.	
WB-16	8.4	498	NR	NR
WB-17	19.8	- 580	NR	NR
WB-18	22.7	906	NR	NR
WB-19	34.5	1400	NR	NR
WB-20	32.0	1500	NR	NR
WB-21	19.0	1025	NR	NR
WB-22	19.0	950	NR	NR
WB-23	13.6	720	NR	NR
WB-24	13.5	783	NR	NR
WB-25	27.8	1140	NR	NR
WB-26	26.3	1295	NR	NR
WB-27	14.8	750	NR	NR
WB-28	14.9	730	NR	NR
WB-29	10.1	540	NR	NR
WB-30	8.3	530	NR	NR
WB-31	27.4	1250	NR	NR
WB-32	12.3	472	NR	NR
WB-33	29.8	1400	NR	NR
WB-34	23.0	1000	NR	NR
WB-35	31.2	1400	NR	NR
WB-36	15.4	348	NR	NR
WB-37	12.8	409	NR	NR
WB-38	17.6	460	NR	NR
WB-39	31.0	738	NR	NR
WB-40	13.1	696	NR	NR
WB-41	10.2	583	NR	NR
A-1	8.3	339	NR	NR
A-2	10.1	432	NR	NR
A-3	8.2	265	NR	NR
A-4	11.7	240	NR	NR
A-5	11.9	293	NR	NR
A-6	13.3	396	NR	NR
A-7	20.0	479	NR	NR
A-8	7.8	411	NR	NR
A-9	9.1	320	NR	NR
A-10	9.4	316	NR	NR
A-11	15.3	451	NR	NR
A-12	26.1	706	NR	NR
A-13	21.6	591	NR	NR
A-14	6.6	408	NR	NR
A-15	22.4	832	NR	NR
A- 16	12.6	565	NR	NR
A- 17	6.6	245	NR	NR
A-18	20.9	820	NR	NR
A-19	10.9	365	NR	NR
A-20	13.0	548	NR	NR

TEST #	SHEAR STIFFNESS K/in.	SHEAR STRENGTH plf	DISPL. @ PEAK in.	FAILURE MODE
۵ - 21	12 9	565	ND	ND
$\lambda = 22$	14 1	505	ND	ND
A-23	28 8	900	ND	ND
λ-24	20.0	750	ND	ND
Δ-25	6 0	263	ND	
A-25 A-26	0.0 Q 3	425	ND	ND
A-20 λ-27	18 3	575	ND	ND
Δ=28	34 0	365	ND	ND
λ=20	10 7	740	ND	ND
Δ=30	15 7	550	ND	ND
A-31	10 0	345	ND	ND
A-32	20 1	615	NR	ND
Δ-33	11.4	515	NR	ND
T-1	9.1	350	NR	ND
T-2	17.5	550	NR	ND
T-3	21.6	755	NR	NR
T-4	16.1	535	NR	NR
T-5	11.3	315	NR	NR
T-6	8.0	275	NR	NR
T-7	12.7	360	NR	NR
T-8	24.8	520	NR	NR
T-9	17.5	500	NR	NR
T-10	36.6	750	NR	NR
1-B ^a	2.9	283	NR PUDDI	E WELD FATLURE
$2-B^a$	8.3	438	NR	"
3ª	3.1	313	NR	H
4ª	7.6	447	NR	H
5ª	2.1	438	NR	11
6ª	2.4	330	NR	11
7 ^a	4.4	533	NR	11
8 ^a	5.2	675	NR	11
9 ^a	5.0	342	NR	H
10 ^{a,b}		300	NR	11
11 ^a	8.1	475	NR	11
12 ^{a,b}		321	NR	11
13 ^{a,b}		358	NR	11
15 ^{a,b}		358	NR	H
16ª	4.8	367	NR	

TABLE A1 CONTINUATION

NOTES

17^{a,b}

^a data obtained from Reference 25

^b the diaphragm was previously subjected to 5 reversed load cycles at two different load levels (0.4P_u and 0.6P_u) and then loaded from zero to failure under monotonic load

358

Ħ

NR

TEST #	SHEAR	SHEAR	DISPL. @	FAILURE
	STIFFNESS	STRENGTH	PEAK	MODE
	K/in.	pli	IN.	
I-1	22.37	430	NR	NR
I-2	21.78	. 650	NR	NR
I-3	16.00	335	NR	NR
I-4	16.67	495	NR	NR
I-5	15.31	388	NR	NR
I-6	22.96	610	NR	NR
W-7	4.80	285	NR	NR
W-8	5.30	275	NR	NR
W-9	9.10	417	NR	NR
4"-Rib-1	22.40	386	NR	NR
4"-Rib-2	41.00	570	NR	NR
4"-Rib-3	33.20	455	NR	NR
8"-Rib-4	44.70	630	NR	NR
8"-Rib-5	84.20	500	NR	NR
8 "- Rib-6	72.60	453	NR	NR
4"-Rib-7	28.20	532	NR	NR
4"-Rib-8	24.00	337	NR	NR
8"-Rib-9	30.00	375	NR	NR
8"-Rib-10	36.90	470	NR	NR
V-Beam-11	22.24	529	NR	NR
V-Beam-12	22.30	340	NR	NR
V-Beam-13	14.83	461	NR	NR
V-Beam-14	12.90	267	NR	NR
1 ^a	57.50	378	NR	1
2ª	120.50	565	NR	. 1
3 ^{a,b}		550	NR	1
4ª	78.20	550	NR	1
5ª	36.40	370	NR	1
6 ^{a,b}		350	NR	1

TABLE A2. SCREWED DIAPHRAGM TEST DATA

NOTES:

FAILURE MODE 1: First, slip along the seam. Second, loss of shear transfer across the sidelap, and finally, local buckling.

- ^a data obtained from Reference 305
- ^b tests 3 and 6 were previously subjected to several nonreversed cyclic loads at low force level, and then loaded from zero to failure under monotonic load.

NR = not reported.

Discussion of data generation is found in Section 3.1.1 of this document.

8.1.2 Data obtained from Reference 299

(Steel-Deck on Timber Framing)

TEST #	SHEAR STIFFNESS lb/in.	SHEAR STRENGTH lb.	DISPL. @ PEAK IN.	FAILURE MODE
· 1		1600	NR	1
2			NR	-
3	2440		NR	·
4	3490	2750	NR	1
5	4180	2800	NR	1
6	2180	2650	NR	1
7	2040	2150	NR	1
8	2470	2600	NR	1
9	3150	2500	NR	1
10	3340	3200	NR	1
11	4660	3700	NR	1
12	3660	2500	NR	1
13	2060		NR	-
14	4200	3400	NR	1
15	5640	4200	NR	1
16	3630	2300	NR	1
17	2880	2150	NR	1
18	6210	4700	NR	1
19	4910	4600	NR	1
20	7561	7200	NR	2
21	5420	4000	NR	3
22	4420	2900	NR	4

TABLE A3. STEEL-DECK ON TIMBER FRAMING TEST DATA.

NOTES:

Shear Stiffness values are for 40% of the strength of test A Shear Strength values are for test B

NR : not reported.

Discussion of data generation is found in Section 3.1.2 of this document.

FAILURE MODES:

- 1 Splitting of the 2x4 around the screws near the seams, tearing of the sheeting around the screws near the seams and crushing of the wood around the screws near the seams.
- 2 The first row of screws between supports C and D pulled out of the wood purlins.
- 3 The chords split at the supports at corners C and D.
- 4 The purlins 6 in. from the support at corner C broke in tension, this purlin broke at a knot.

8.1.3 Data obtained from Reference 3

TEST #	SHEAR STIFFNESS K/in.	SHEAR STRENGTH Kips	DISPL. @ PEAK in.	FAILURE MODE
Q	19.50	57.40	NR	NR
R	29.50	106.00	NR	NR

TABLE A4. STEEL-DECK TEST DATA

NOTES:

NR = not reported.

Discussion of data generation is found in Section 3.4.1 of this document.

8.1.4 Data obtained from Reference 320.

TEST #	SHEAR STIFFNESS N/mm.	SHEAR STRENGTH N/m	DISPL. @ PEAK in.	FAILURE MODE
2-A	902	2407	NR	1
3 - A	967	2384	NR	1
4-A	931	2311	NR	1
2-B	789	4529	NR	2
3-B	837	4529	NR	3
4-B	732	4529	NR	2
6	852	4767	NR	2
7	958	4529	NR	2
8	967	4767	NR	3

TABLE A5. STEEL-DECK ON TIMBER FRAMING TEST DATA.

NOTES:

NR = not reported.

Panels 2-A to 4-B were tested as a cantilever beam, panels 6-8 were tested as a simple beam.

FAILURE MODES:

1 Purlin 2x4 pulled across 2x8 at the corner C.

- 2 The screws in the first rib between corners C and D pulled out of the purlins.
- 3 The screws in the first rib between corners A and B pulled out of the purlins.

8.2 Wood Diaphragms

TEST ⁻ #	INITIAL STIFFNESS K/in.	SHEAR STRENGTH Kips	DISPL. @ PEAK in.	FAILURE MODE
1	153	44	1.30	NR
2	101	40	1.09	NR
3	88	36	1.14	NR
4	118	40	1.90	NR
5	155	48	1.95	NR
6	60	48	1.74	NR
7	29	26	1.76	NR
8	112	36	1.99	NR

8.2.1 Data obtained from Reference 518

TABLE A6. WOOD DIAPHRAGM TEST DATA

NOTES:

NR = not reported.

The test data was estimated from graphs reported on Reference 518.

Discussion of data generation is found in Section 3.2.1 of this document.

8.2.2 Data obtained from Reference 3.

TABLE A7. WOOD DIAPHRAGM TEST DATA.

TEST #	INITIAL STIFFNESS K/in.	SHEAR STRENGTH Kips	DISPL. @ PEAK in.	FAILURE MODE
E	13.3	6.56	NR	NR
H	23.0	14.20	NR	NR
I	47.2	15.90	NR	NR
K	61.3	40.80	NR	NR
B	14.2	7.91	NR	NR
D	17.4	10.80	NR	NR
С			NR	NR
N	26.2	31.90	NR	NR
P	49.7	57.80	NR	NR

NOTES:

NR = not reported.

Discussion of data generation is found in Section 3.4.1 of this document.

8.3 Steel-Deck-Reinforced Concrete Diaphragms

8.3.1 Data obtained from Reference 378.

TABLE A8. SDRC TEST DATA.

TEST No.	k _i (1) K/in	V _u ⁽²⁾ kips	e ⁽³⁾ in.	m ⁽⁴⁾ K/in	e ⁽⁵⁾ in.	FAILURE MODE
1	1800					-
2	2000	136	0.1	-73.4	1.88	1
3	1600	78	0.1	-27.2	2.75	2
4	1300	70	0.1	-28.5	2.44	2
5	1700	82	0.1	-76.1	1.16	1
6	2600	97	0.4	-39.8	2.84	2
7	1500	113	0.2	-93.5	1.45	2
8	1100	42	0.1	-26.0	1.75	3
9	1900	172	0.2	-164.2	1.16	1
10	1700	131	0.4	-54.1	2.60	1
11	1600	73	0.4	-29.1	2.77	4
12	1800	131	0.2	-88.4	1.87	1
13	1900	205	0.2	-601.0	0.54	1
14	1900	172	0.4	-151.0	1.54	4
15	1300	71	0.4	-64.0	1.51	4
16	1300	75	0.4	-53.5	1.80	1
17	2200	119	0.4	-99.9	1.60	5
18	1700	120	0.2	-44.6	2.86	1
19	1300	118	0.4	-87.7	1.75	1
20	1300	76	0.2	-76.9	1.17	6
21	1200	94	0.4	-103.6	1.31	6
22	2100	139	0.4	-77.1	2.20	1
23	1700	84	0.4	-99.7	1.24	6
24	2100	124	0.4	-69.9	2.18	1
25	1900	133	0.2	-102.0	1.52	1
26	1700	64	0.1	-39.4	1.67	1
27	2000	70	0.1	-121.7	0.68	7
28	2000	93	0.2	-235.5	0.59	7
29	2300	101	0.1	-142.9	0.78	1
30	1900	93	0.1	-63.3	1.41	7
31	1500	47	0.1	-46.5	1.00	8
32	1000	39	0.1	-76.8	0.56	

Initial Stiffness (k_i) based on 0.025 in. of virgin data. Average Peak Stabilized Strength (V_u) . Average Displacement at Peak Strength (e_p) . Degrading Slope after Peak (m).

⁵ Maximum Displacement (e_m)

NOTES:

Maximum displacement and degrading slope after peak were determined by linear regression analysis on stabilized data.

Discussion of data generation is found in Section 3.3.1 of this document.

FAILURE MODES:

- 1 Diagonal tension.
- 2 Interfacial Shear.
- 3 Diagonal tension/shear connector.

4 Shear transfer mechanism-transverse.

5 Shear transfer mechanism-parallel.

- 6 Shear transfer mechanism.
- 7 Concrete shear failure around studs.
- 8 Weld failure.
8.4 Plank Diaphragms

8.4.1 Data obtained from Reference 376.

TEST No.	k _i (1) K/in	V ⁽²⁾ Kips	$e_p^{(3)}$ in.	m ⁽⁴⁾ K/in	e _m ⁽⁵⁾ in.	FAILURE MODE
4	1281	72.1	0.15	-11.66	5.74	1
5	2005	83.4	0.30	-6.69	9.29	1
6	1376	58.1	0.30	-6.23	8.46	2
7	1647	56.7	0.10	-5.12	8.79	2
8	716	25.6	0.30	-3.90	4.81	2
8b	1003	35.7	0.08	-2.86	10.50	2
9	1486	64.8	0.20	-8.24	6.31	1
10	2734	76.4	0.20	-15.69	4.68	2
11	2143	99.8	0.30	-45.24	2.10	3
12	1596	113.1	0.20	-33.21	3.33	3
13	2698	237.6	0.29	-55.85	2.95	3
14	3289	258.4	0.30	-40.87	3.56	3
15	2518	156.6	0.10	-56.76	2.98	1

TABLE A9. PLANK DIAPHRAGM TEST DATA.

(1)

(2)

Initial Stiffness (k_i) based on virgin data. Average Peak Stabilized Strength (V_i) . Average Displacement at Peak Stabilized Strength (e_p) . (3)

(4) ' Degrading Slope after Peak (m).

(5) Maximum Displacement (e_{max}).

NOTE:

Degrading slope after peak and maximum displacement were obtained by linear regression analysis on stabilized data.

FAILURE MODES:

- 1 Shear Bond
- 2 Tension Bond
- Diagonal Tension 3

9. APPENDIX B: DIAPHRAGM SYSTEMS

Masonry buildings are constructed with various types of floor and roof diaphragm systems. Floor systems can be made of composite steel deck, timber, precast concrete, reinforced concrete and coldformed steel.

Composite steel deck floor systems are constructed by fastening sections of cold-formed steel deck to framing members which are typically steel. The fastening may be done with arc spot welds, screws, powder driven pins, positive shear transfer devices such as headed shear studs or some combination of fasteners. Seams between adjacent panels may either be welded, screwed or button punched/crimped. A layer of concrete is placed on the deck with shrinkage and temperature steel added. Also supplemental reinforcing steel may be added in some cases, either in the negative or positive moment regions of the slab. The shear transfer device typically consists of embossments rolled into the deck profile, transverse wires attached to the deck or keystoneshaped profiles. Figure B1 illustrates some steel deck profiles.

Timber panels are constructed by fastening the sheathing material to the framing support. The sheathing material can be lumber (boards), plywood, and particle boards. Lumber sheathing is used in diagonal, double diagonal or straight patterns, and it is generally one-inch nominal thickness. Plywood is generally made up of an odd number of layers of veneer at right angles and is referred to as three-ply, five-ply, etc. Plywood is also constructed with the grain direction of adjacent plies parallel, and is often referred to as four-ply, six-ply, etc. Common thicknesses ranges from 5/16 to 1-1/8 inches. The framing support is generally made of wood members such as 2-inch dimension lumber, glulam (glued laminated timber) beams, built-up plywood beams, and open-web trusses; however, there are some systems that utilize metal framing. The connection between the sheathing material and the framing elements is made by mechanical fasteners such as nails and staples, or adhesives or a combination of both [322]. Figure B2 illustrate some wood panel types.

Precast concrete units are usually produced as standard precast members or precast-prestressed concrete members. Standard precast members utilize ordinary reinforcing bars and they are designed by standard reinforced-concrete theory. Precastprestressed concrete members utilize stretched wires which induce a prestressing force into the element resulting in a fully or partially prestressed member. Precast members may be used as wall panels and as roof and floor elements. Wall panels are used as bearing walls or as curtain walls attached to columns and beams. Roof and floor elements are made in a variety of shapes, and in a wide range of span lengths and depths. The common shapes are flab slab, hollow-core plank, single and double tee. Figure B3 shows



Figure B1. Typical steel-deck profiles [378] 137





Figure B2. Plywood and diagonally sheathed panels [22,149] 138



,



13**9**

the cross section of a typical hollow-core plank.

Reinforced concrete floor and roof systems may be classified as one-way or two-way systems. One-way systems have the main reinforcement running in one direction only, two-way systems have the main reinforcement running in two perpendicular directions. The most common types for one-way systems are the slab-beam-girder systems and the one-way joist floor. The slab-beam-girder systems consists of a slab supported by a series of parallel beams supported at their extremities by girders, which in turn frame into A one-way joist floor consists of a series of small columns. closely spaced joists, framing into girders, which are in turn carried by columns. Common types for two-way systems are the twoway slab on beam, flat slab, and flat plate. Two-way slabs on beams aresolid slabs supported by beams on the column lines on each side of slab panel. Flat slabs are beamless systems with dropped panels or column capitals or both. Flat plates are beamless systems with no dropped panels or column capitals. Figure B4 illustrates typical reinforced concrete floor systems.

Cold-formed steel floor systems are made from steel sheets with thicknesses usually varying from 0.012 to 0.075 in., and with depths usually varying from 1-1/2 to 3 in. The system is constructed by fastening sections of cold-formed steel deck to framing members which are commonly steel. Floor and roof systems are capable of resisting horizontal loads in addition to gravity loads if they are interconnected to each other and adequately fastened to the supporting frame. Figure B5 illustrates typical cross sections for cold-formed steel decks.











Figure B5. Typical cross-sections for Cold-Formed-Steel-Decks [113,299]

.

. . -